Antioch University

AURA - Antioch University Repository and Archive

Antioch University Full-Text Dissertations & Theses

Antioch University Dissertations and Theses

2018

Smallholder Farmers, Environmental Change and Adaptation in a Human-Dominated Landscape in the Northern Highlands of Rwanda

Apollinaire William Antioch University, New England

Follow this and additional works at: https://aura.antioch.edu/etds

Part of the Agriculture Commons, Environmental Studies Commons, Other Forestry and Forest Sciences Commons, and the Water Resource Management Commons

Recommended Citation

William, A. (2018). Smallholder Farmers, Environmental Change and Adaptation in a Human-Dominated Landscape in the Northern Highlands of Rwanda. https://aura.antioch.edu/etds/416

This Dissertation is brought to you for free and open access by the Antioch University Dissertations and Theses at AURA - Antioch University Repository and Archive. It has been accepted for inclusion in Antioch University Full-Text Dissertations & Theses by an authorized administrator of AURA - Antioch University Repository and Archive. For more information, please contact hhale@antioch.edu.



Department of Environmental Studies

DISSERTATION COMMITTEE PAGE

The undersigned have examined the dissertation titled:

SMALLHOLDER FARMERS, ENVIRONMENTAL CHANGE AND ADAPTATION IN A HUMAN-DOMINATED LANDSCAPE IN THE NORTHERN HIGHLANDS OF RWANDA

Presented by Apollinaire William, candidate for the degree of Doctor of Philosophy, and hereby certify it is accepted*.

Beth A. Kaplin, Ph.D., Committee Chair (Core Faculty, Department of Environmental Studies, Antioch University New England)
James S. Gruber, Ph.D., Committee Member (Core Faculty, Department of Environmental Studies, Director Doctoral Program, Antioch University New England)
Joel Hartter, Ph.D., Committee Member (Faculty Director, Masters of the Environment graduate degree program, Environmental Studies Program, University of Colorado)

Defense Date: April 20, 2018 Date Approved by all committee members:

*Signatures are on file with the Registrar's Office at Antioch University New England

SMALLHOLDER FARMERS, ENVIRONMENTAL CHANGE AND ADAPTATION IN A HUMAN-DOMINATED LANDSCAPE IN THE NORTHERN HIGHLANDS OF RWANDA

by

Apollinaire William

A dissertation submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

(Environmental Studies)

at

ANTIOCH UNIVERSITY NEW ENGLAND - KEENE, NH USA

© copyright by Apollinaire William 2018 All Rights Reserved

Dedication

First and foremost, to my parents for their unfailing support throughout my life. To my family and friends for having my back throughout this study.

To each of you from whom I have gleaned academic and professional experience, I dedicate this work.

Acknowledgments

I would like to sincerely thank my dissertation committee chair and adviser, Beth A. Kaplin, for her unfailing support and encouragements all the way through my academic pilgrimage. She, along with my other dissertation committee members James Gruber and Joel Hartter, has supported me along the way and has challenged me to do the best of myself.

I would also like to extend my gratitude to my colleagues and classmates from Antioch University New England for their cheerful smiles that fed my academic and social morale. Let a zephyr carry my whispers of thanks to Antioch University New England faculty, particularly Peter Palmiotto, George Tremblay, Jean Kayira, James Karlan and Joy Ackerman who kept checking on me and encouraging me to keep moving forward towards a successful completion of my dissertation.

In addition, my thanks go to Stephen Jones, Laura Andrews, Jennifer Fritz, Susan Weller, Amy Janvier, Abagail Jones, Rachel Sperling and Dorothy Shuteran, for maintaining a hospitable and nurturing academic environment that enabled me to achieve my academic goals and objectives.

Finally, I am indebted to my friends and colleagues Nicole Gross Camp, Shawn Margles, Stella Jordan, Colin Apse, Cara Lee, Juvenal Lion, Glenn Bush, Marj Adams, Ange Imaninshimwe, Omar Ndamutsa, Robert E. Ford, Jeanne Wright More, Janet and Terry, Erasme Uyizeye and Olivia , Susan H.Wilson and Phillip D. Wilson, Lois Buchan and Wendy, Joseph Mlotha, Tharcisse Ukizintambara, John Bosco Sumani, Dale Fisher, Robin Martino, Donald Strauss, and many other friends for having my back and keeping me going during times of doubt and stress.

Abstract

Climate change and crop intensification are key challenges to the livelihoods and wellbeing of the majority of rural smallholder farmers in developing countries, particularly in humandominated, climate-sensitive landscapes such as the northern highlands of Rwanda where issues of fluvial floods, soil erosion pose serious threats to the livelihoods of smallholder farmers. In this mixed methods study conducted between August and December 2015, I explored smallholder farmers' perceptions by examining what barriers might hinder the process of agroforestry adoption by smallholder farmers, what socio-economic and physical factors and attitudes influence crop choices, motivations for smallholder farmers' willingness to plant trees within riparian buffer zones and opportunities and challenges to the establishment of riparian buffer zones that maintain ecosystem services. Results indicate that challenges to adoption of agroforestry to support climate change adaptation and food security in Rwanda are related to land scarcity, poverty, limited technological and financial capacity among most smallholder farmers, limited engagement of smallholder farmers in agroforestry research and an inclination for short term benefits that could hinder adoption of agroforestry which has a long term investment. Most smallholder farmers believed that the onset of short rains comes earlier in recent years compared to more than ten years ago. In response, most farmers reported that they plant crops earlier in the season. Results from rainfall analysis, although not conclusive, show a shift in rainy day frequency. Respondents who strongly agreed that soil erosion within farms proximal to streams is a serious threat were more likely to support the idea that establishing a riparian buffer would help entrap sediments and mitigate soil erosion within farmlands adjacent to streams. However, farmers reported that establishing a functional riparian buffer requires

iv

engagement with extension services, financial incentives and technological assistance. Perceptions of costs and benefits of riparian zone management was found to be the most important factors influencing farmers' intentions to manage riparian zones. Engagement of smallholder farmers in the agrarian policy development process and their active participation in the implementation of adaptation strategies may be needed in order to provide an opportunity for farmers to learn technologies involved and assess costs and benefits of the practices. This would also insure that smallholder farmers' goals and interests are met.

Dedicationi
Acknowledgmentsii
Abstract iv
List of Figuresix
List of Tablesxi
Chapter 1: General Introduction
References
Chapter 2: The potential and challenges of agroforestry to support climate change adaptation and food security in a human-dominated, climate sensitive landscape
Abstract
Introduction
Colonial and post-colonial agroforestry in Rwanda15
Multifunctional role of agroforestry in Rwanda19
Environmental benefits of Agroforestry in Rwanda
Smallholder farmers and Agroforestry: opportunities and challenges to adoption
New perspectives in agroforestry in Rwanda27
Extension services and agroforestry adoption in Rwanda
Forest and agrarian policy: engagement of farmers
Conclusion
References
Chapter 3: Smallholder farmers perceptions of climate change and adaptation strategies in the northern highlands of Rwanda
Abstract
Introduction
Methods
Study area
Household Climate Perceptions and Risks
Rainfall data analysis
Results
General farmer characteristics and agricultural systems within the study site

Farmers' perceptions of climate.	60
Timing of onset and cessation of rainy seasons	61
Timing of crop planting	63
Rainfall seasonal variability	65
Perception of climate risks.	69
Farmers' perceptions of recent increase in the cultivation of certain crops and rise in concerns about climate change risks.	72
Attitudes towards the adoption of agroforestry.	74
Perceived Challenges to the adoption of agroforestry	75
Discussion	76
Awareness of change of onset of short rainy seasons	76
Timing of planting and onset of short rains	76
Crop plant choice and climate adaptation	77
Reconciliation of climate-sensitive crops and agroforestry.	78
References	84
Chapter 4: Ecosystem services in riparian agricultural landscapes: A study of smallholder	
farmers' perceptions in the northern highlands of Rwanda	95
Abstract	95
Introduction	97
Methods	102
Study area	102
Superpixels and farmer sampling.	103
Farmer survey on perception of the importance of ESs.	105
Perceived role of riparian buffers and maintenance of regulating ESs	106
Results	107
Perception of soil erosion, floods and riparian zones.	110
Smallholder farmers' willingness to manage riparian zones and challenges.	111
Plants and ESs provision: Drivers of smallholder farmer decision making	113
Challenges to the maintenance of ESs	115
Discussion	116
Preferred ESs	116

Determinants of smallholder farmers' choice of crops: environmental concerns va profitability.	s. economic 119
Choice of multifunctional riparian buffer plants.	121
Perception of ESs and experienced environmental degradation	123
Smallholder farmers' willingness to maintain ecosystem services (ES)	126
Determinants of plant choice for riparian zones	128
Theory of Planned Behavior and perception of ESs	
References	134
Chapter 5: General Conclusion	154
Suggestions for Future Research	

List of Figures

Figure 3-1. Upper Mukungwa Watershed in Rwanda with locations (superpixels) where
household interviews were conducted56
<i>Figure 3-2. Perceptions of changes in current climate/weather/seasons change by farmers in the</i>
Upper Mukungwa watershed, Rwanda60
Figure 3-3. Farmers' perceptions of the timing of rainy seasons61
Figure 3-4. Farmers' responses about their perceptions of the timing of crop planting64
Figure 3-5. Average number of days of rain for the period 1971-1998 and 1999-2013 for the
month of August66
Figure 3-6. Average number of days of rain for the period 1971-1998 and 1999-2013 for the
month of September66
Figure 3-7. Cumulative daily rainfall averages for Kigali Airport meteorological station,
<i>Rwanda</i> 67
Figure 3-8. Monthly average rainfall at Kigali Airport meteorological station, Rwanda68
Figure 3-9. Rating of climate risks considered to be of significant concern among farmers in the
<i>study area69</i>
Figure 3-10. Crop susceptibility to damage due to soil erosion, based on farmers' perception -73
Figure 3-11. Rating of factors that drive farmers' choice of crops to plant74
Figure 3-12. Rating of major challenges inherent to potential adoption of agroforestry by
farmers in the Upper Mukungwa Watershed, Rwanda75
Figure 3-13. Agricultural management in the Musanze district79
Figure 4-1. Survey superpixels across the upper inhabited Mukungwa watershed study area,
Rwanda 105
Figure 4-2. Proportion of total farmers who agree on crop choice criteria 107
Figure 4-3. Proportion of all respondents in agreement on crop susceptibility to erosion in the
Upper Mukungwa Watershed, Rwanda 108
Figure 4-4. Increase in farm plots allocated to cultivation of certain crops as reported by survey
participants in the Upper Mukungwa Watershed, Rwanda from 2009 – 2014 109
Figure 4-5. Plants preferred by smallholder farmers for riparian zones in the Upper Mukungwa
Watershed, Rwanda 110
Figure 4-6. Types of ESs expected from fields adjacent to streams if well maintained riparian
zones are present, according to respondents farming in the Upper Mukungwa Watershed,
northern Rwanda 112
Figure 4-7. Choice of plant species for riparian buffer zones by smallholder farmers in the
Upper Mukungwa Watershed, Rwanda 113
Figure 4-8. Recent trends in consolidation of land use areas under cultivation of priority crops
in Season A (from September to February) in Rwanda 124

Figure 4-9. Changes in on-farm yields of major crops in response to the use of distributed inputs (improved seeds/planting materials and/or fertilizers) in Rwanda.------125

List of Tables

Table 2-1 Agroforestry Practices in Rwanda and their Socio-ecological Functions	19
Table 3-1 Population Statistics of Respondents in the Upper Mukungwa Watershed, Rwanda	59
Table 3-2 Perceptions of Local Households about Changes in Onset and Cessation of Rainy	
Season	62
Table 3-3 Relationship between Perceptions of the Timing of the Onset of the Short Rainy Sec	ison
and Timing for Crop Planting	64
Table 3-4 Relationship Between Perceived Climate-related Issues Believed by Farmers to be	of
Concern and Socio-economic Factors	70
Table 4-1 Rwanda Crop Calendar for Major Crops	103
Table 4-2 Chi-square Test Results Showing Relationship between Farmers' Choice of Plant	
Species for the Riparian Buffer and Benefits Farmers Expected from a Functional Riparian	
Buffer	115

Chapter 1: General Introduction

Our planet's climate continues to change at an unprecedented rate. Vulnerability to impacts associated with climate change is adding to ongoing environmental and socio-economic challenges particularly in tropical areas where many societies are dependent on resources that are sensitive to climate change. Climate change has political, socio-economic and environmental effects, in developed countries as well as in developing countries. However, effects are most felt in developing countries, particularly in Africa, due to the magnitude and frequency of extremes weather events such as floods and droughts that overwhelm some poor nations' ability to adapt or mitigate it (IPCC, 2007; Toulmin, 2009; Collier et al., 2008; Omambia & Gu, 2010).

Soil degradation and erosion in Africa are considered to be more rapid and extensive than in other continents of the globe because of the generally high population pressures and land-use intensity resulting in higher nutrient exports through crop removal with significantly lower percentage of arable land under fallow (Mortimore & Harris, 2005). Global river runoff is believed to have increased significantly since the 20th century due essentially to deforestation, and is particularly evident in many African countries. Changes in the distribution of river flows and groundwater recharge over space and time are determined by changes in temperature, evaporation and, crucially, precipitation (Chiew, 2006; Fu et al., 2007). Deforestation of tropical forests, at the rate of about four million hectares every year in Africa (Low, 2005), decreases litter and canopy leaf cover that otherwise intercepts rainfall, facilitates water infiltration, and reduces runoff (Moutinho & Schwartzman, 2005), exacerbating storm events on the continent. Increasing demands for agricultural land, coupled with land scarcity in some countries has led to conversion of natural riparian buffers to crop cultivation, thus compromising the ability of buffer

zones to buffer the effects of floods, control sediments, trap nutrients and maintain stream bank stability. The total rainfall from intense precipitation events is expected to increase over most areas, particularly in tropical and high latitude areas (Solomon et al., 2007), exacerbating this problem. It follows that flood frequency and magnitude is projected to increase in the regions experiencing increase in precipitation intensity (e.g. Meehl et al., 2007). Some drainage basins are projected to experience increases in flood frequency (Kundzewicz et al., 2008). These changes influence the ability of ecosystems to provide services to people living off these lands. This is particularly problematic where so many people rely on rainfed agricultural for a subsistence lifestyle.

Ecosystem services are tangible and intangible benefits derived from the functioning of ecosystems, which include flood and soil erosion control, sediment and nutrient control from runoff, pollination, food and fuel production, carbon sequestration, in agricultural landscapes (Swinton et al., 2007; Zhang et al., 2007; Felipe Lucia et al., 2014; Egoh et al., 2008). Agricultural ecosystems can be managed effectively to optimize provisioning ecosystem services (MEA, 2017), which are reliant on a number of regulating ecosystem services including soil fertility and pollination, as inputs to production (Zang et al., 2007). Despite the merits of regulating ecosystem services, agriculture intensification programs in many African countries have not sufficiently incorporated those benefits in policy development. The lack of integration of regulating ecosystem services has resulted in soil fertility loss, soil erosion and water degradation (Zhang et al., 2007), thus reducing the capacity for smallholder farmers to adapt to climate change in many African countries.

Climate adaptation strategies to restore and maintain ecosystem services within agricultural landscapes include but are not limited to intercropping which consists of mixing two or more crops in one field during same crop season, afforestation, agroforestry and the establishment of functional riparian buffer zones around water bodies. Agroforestry has the potential to improve soil fertility and contribute to alleviating pressure on natural forests by providing fuel wood and timber. Agroforestry can also provide fodder for livestock and sticks to support climbing beans, medicine, handicraft raw material and plant disease control. In agricultural landscapes where slopes are steep or where crop cultivation has seriously encroached on riparian zones, studies show that agroforestry is the most naturally appropriate practice to conserve soil stability against erosion, fertility and control nutrient leaching and sediments from rushing into streams. In agricultural landscapes where agricultural intensification is associated with intensive use of industrial fertilizers, riparian buffers contribute to trapping sediments and nutrients such as phosphorus and denitrify nitrites, thus contributing to water quality in streams.

Owing to their multifunctional roles, trees and woodlots on farms are now considered in the preparation of the Global Forest Assessment 2015 (FAO, 2010b). The idea of sustainable management of natural resources extends beyond forests to apply to trees outside forests as well (Kleinn, 2000). The general objective is to explore farmers' perceptions of climate variability, adaptation and barriers to adaptation.

The adoption of agroforestry by smallholder farmers in developing countries poses challenges, especially for low-income, uneducated farmers. The uptake of agroforestry is hindered by a number of factors including lack of technical and financial capacities to implement the practice effectively. In many places, agroforestry is administered from above in a top-down approach

without gauging farmers' interests and goals and without equipping farmers with necessary tools, skills and financial means required for the success of the practices. Studies show that agroforestry, like any other agricultural practice, can be adopted by farmers if they see its effectiveness in term of cost/benefits ratio by monitoring model farm plots, if they have financial means and if they feel empowered enough to practice agroforestry. The farmer makes decisions to plant a buffer based on net crop prices and costs of land or planting. The choice of buffer type is affected by crop price, farm size, relative incentive payments, relative cost share rates, and amount of damage from various sources (Lynch and Brown, 2015).

Understanding farmers' perceptions of the role of ecosystem services within an agricultural landscape, especially how they support sustainable food production and livelihood improvements is key to formulating effective policies that are not only tailored to farmers' needs and goals but also which accommodate environmental concerns. Socio-psychologists suggest that, based on the theory of planned behavior TPB, intentions for a behavior change, for example decision to adopt certain agricultural practices that accommodate ecosystem services, hinges on a number of factors, including perceived behavioral control, attitudes about sustainable management of farming practices that foster both economic and environmental outcomes, and available labor (McGinty et al., 2008). Overall, perceived behavioral control appears to have the most significant correlation with farmers' intentions to adopt or maintain agroforestry for example (McGinty et al., 2008).

Understanding farmers' perceptions of their self-efficacy in management of agroforestry for ecosystem service restoration and maintenance as well as their attitudes, their socio-economic

characteristics and their source of experience with the practices, whether through participating with agroforestry programs or through the establishment of functional riparian buffer zones, is essential for sustainable agriculture. Unfortunately, most agricultural policies are designed without participation of farmers and do not necessarily foster synergies between regulating and provisioning ecosystem services within agricultural settings. In a human-dominated landscape there is high reliance on scarce farmland, and in Rwanda, for example, about 57% of farms are less than 0.5 ha (Singh, 2000) with an average farm size of 0.76 ha (Booth & Golooba-Mutebi, 2012). In these cases it is important to explore smallholder farmers' perceptions about ecosystem services generated by sustainable agriculture apart from food, their adaptation strategies and their willingness to adopt agroforestry particularly given scarcity of their land coupled with poverty.

My research focused on Rwanda where the issues of reliance on scarce, climate-sensitive landscapes for subsistence agriculture come together with policies for agriculture intensification that may not consider ecosystem services. Given evidence that the Northern and Western Provinces in Rwanda have been exposed to more frequent and extreme climate events while understanding of climate variability and adaptation remains limited, studies of farmers' perceptions of climate variability and barriers to adaptation could shed light on development of effective adaptation strategies. I examined farmers' perceptions of climate variability in the Upper Mukungwa Watershed and their adaptive responses. Socio-economic and psychological barriers to farmers' adaptation are discussed, incorporating perspectives of farmers and from existing literature. Data were obtained from smallholder farmers' interviews. Recorded rainfall data at the Kigali meteorological station based at the Kigali International Airport from 1971 to

2013 are also incorporated to demonstrate the actual climate variability of the region. Hopefully, this study helps enhance understanding of the adaptation process at farm level, particularly in the Musanze District with focus on crop choice, timing of planting, agroforestry and riparian buffers for ecosystem service restoration and maintenance.

In Rwanda, the problems of land degradation coupled with climate change impacts, which translate into landslides, floods, soil erosion, stream channel degradation, and scarcity of arable land come together to create vulnerability among small holder farmers. Riparian zones have been stripped of their natural vegetation in favor of food crop cultivation. While the Rwandan government has introduced a crop intensification program and has set up regulations about protection of riparian zones around streams and lakes, information about farmers' perceptions about government efforts as well as their beliefs about factors that could potentially contribute to maintaining ecosystem services within riparian buffer zones has not been well documented. My research aimed at examining farmers' perceptions regarding climate change, adaptation and barriers to adaptation, and provides state-of-the art information about the status of agroforestry in Rwanda and (3) identifying socio-economic factors that contribute to farmers' intentions to grow particular plant species within riparian zones on their farms.

My research contributes to the theoretical literature and provides recommendations for future research and policy direction with respect to maintaining ecosystem services, whether provisioning or regulating, within agricultural landscapes. To achieve the main goals of this research project, I focused on the following research questions:

1) What are the barriers that might hinder the process of agroforestry adoption by smallholder farmers given historical and present status of agroforestry in Rwanda?

2) What are the socio-economic and physical factors and attitudes that influence the choice of food crops to plant in a given season?

3) What are the motivations and main purposes that drive smallholder farmers' willingness to plant trees within riparian buffer zones on their farms?

4) What are the opportunities and challenges to establishing riparian buffer zones that maintain ecosystem services?

A better understanding of the determinants of household choice of adaptation strategies such as the timing of crop planting, the choice of food crops to plant, and the choice of trees to plant for riparian buffers was analyzed using the theory of planned behavior (TPB) conceptual framework and empirical models to assess motivations for agroforestry adoption and ecosystem service maintenance. The dissertation aims to clarify these dependencies through the combined use of comprehensive farm and household level studies coupled with farm agroforestry assessment, to analyze the relevance of agroforestry in meeting the needs for maintaining ecosystem services for climate change adaptation.

The dissertation consists of five chapters. Chapter 2 presents a review of agroforestry in Rwanda and assesses the potential of agroforestry systems to provide multiple benefits to smallholder farmers. I explored the literature on agroforestry in the face of climate change adaptation and food security in a human-dominated, climate sensitive landscape of the northern highlands of Rwanda. I also reviewed opportunities and challenges pertaining to the adoption of agroforestry in Rwanda. Chapter 3 provides an assessment of smallholder farmers' perceptions of the onset

of rainy seasons and the timing of crop planting as well as the level of vulnerability of currently cultivated crops to climate change. In this chapter, I also assessed smallholder farmers' willingness to adopt agroforestry and examined the relationship between their willingness to adopt agroforestry for climate change adaptation and socio-economic factors. Chapter 4 provides an understanding of farmers' views and experiences about the importance of ecosystem services, and strategies for restoring and maintaining those ecosystem services in riparian buffer zones. It discusses the reasons why farmers choose to plant particular trees in riparian zones on farmlands and identifies the most important aspects that households consider when deciding to plant a variety of different trees species in agricultural lands. Chapter 5 summarizes findings and key discussion points teased out from previous chapters. In this chapter, I formulate recommendations and suggestions for future research as well as the way forward regarding policy directions.

References

- Adger, W. N. (2003). Social capital, collective action, and adaptation to climate change. *Economic geography*, *79*(4), 387–404.
- Booth, D., & Golooba-Mutebi, F. (2012). Policy for agriculture and horticulture in Rwanda: a different political economy? Retrieved from

http://opendocs.ids.ac.uk/opendocs/handle/123456789/2250

- Chiew, F. H. S. (2006). Estimation of rainfall elasticity of streamflow in Australia. *Hydrological Sciences Journal*, *51*(4), 613–625.
- Collier, P., Conway, G., & Venables, T. (2008). Climate change and Africa. Oxford Review of Economic Policy, 24(2), 337–353.
- Fu, G., Charles, S. P., & Chiew, F. H. S. (2007). A two-parameter climate elasticity of streamflow index to assess climate change effects on annual streamflow. *Water Resources Research*, 43(11).

IPCC. (2007). Climate Change 2007. New York, NY.

Kundzewicz, Z. W., Mata, L. J., Arnell, N. W., Doll, P., Jimenez, B., Miller, K., Shiklomanov, I.
 (2008). The implications of projected climate change for freshwater resources and their management. *Hydrological Sciences Journal*, *53*(1), 3–10.

Low, P. (2005). Climate change and Africa. Cambridge: Cambridge University Press.

- Meehl, G. A., Covey, C., Taylor, K. E., Delworth, T., Stouffer, R. J., Latif, M., Mitchell, J. F. B.
 (2007). THE WCRP CMIP3 Multimodel Dataset: A New Era in Climate Change
 Research. *Bulletin of the American Meteorological Society*, 88(9), 1383–1394.
- Mortimore, M., & Harris, F. (2005). Do small farmers' achievements contradict the nutrient depletion scenarios for Africa? *Land Use Policy*, *22*(1), 43–56.

- Moutinho, P., & Schwartzman, S. (2005). *Tropical Deforestation and Climate Change*.
 Washington D.C.: Amazon Institute for Environmental Research. Retrieved from http://www.scribd.com/doc/60507020/4930-Tropical-Deforestation-and-ClimateChange3.
- Omambia, C. S. A. N., & Gu, Y. (2010). The Cost of Climate Change in Tanzania: Impacts and Adaptations. *Journal of American Science*, *6*(3).
- Singh, R. B. (2000). Environmental consequences of agricultural development: a case study from the Green Revolution state of Haryana, India. *Agriculture, Ecosystems & Environment*, 82(1), 97–103.
- Solomon, S., Qin, D., Manning, M., Marquis, M., Averyt, K., Tignor, M. M. B., Chen, Z. (2007).
 Climate Change 2007. The Physical Science basis: Summary for policymakers.
 Contribution of Working Group I to the Fourth Assessment Report of the
 Intergovernmental Panel on Climate Change. *Climate Change 2007: The Physical Science basis: Summary for policymakers. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change.*
- Swinton, S. M., Lupi, F., Robertson, G. P., & Hamilton, S. K. (2007a). Ecosystem services and agriculture: Cultivating agricultural ecosystems for diverse benefits. *Ecological Economics*, 64(2), 245–252.
- Toulmin, C. (2009). *Climate change in Africa*. Zed Books. Retrieved from http://www.cabdirect.org/abstracts/20093271033.html
- Zhang, W., Ricketts, T. H., Kremen, C., Carney, K., & Swinton, S. M. (2007). Ecosystem services and dis-services to agriculture. *Ecological Economics*, *64*(2), 253–260.

Chapter 2: The potential and challenges of agroforestry to support climate change adaptation and food security in a human-dominated, climate sensitive landscape Abstract

Conventional agriculture is the mainstream approach to achieving global food security but has also caused extensive environmental and social harms. Recent consensus shows we need to move away from the current, narrow focus on yield toward an agriculture that accommodates and enhances environmental goals which in return can support agriculture. Such a shift could also enhance climate change adaptation. As the effects of climate change on agricultural activities of small holder farmers in rural developing regions become more evident, attention on the potential of agroforestry for adaptation and food security has heightened. Increasing human populations in many developing countries are placing greater pressures on natural land cover, creating humandominated landscapes with reduced ability to sustain biodiversity and ecosystem services such as water retention, carbon sequestration, and soil fertility. In the face of increasing land scarcity for agricultural purposes in many African countries, agroforestry has been proposed as an alternative for restoring some of the emergent ecosystem services originally provided by natural forests. In this review, I explore the potential for and constraints to adoption of agroforestry practices to support climate change adaptation and food security. I focus this review on the case of smallholder farmers in Rwanda, a country that has undergone significant recent and rapid transformations in agriculture, economic development and land degradation. Although agroforestry practices were imposed on Rwandan farmers during the colonial period, implementation of agroforestry has slowed in Rwanda since its independence in 1962. Factors that contributed to the slow adoption of tree planting on farms include socio-economic constraints, land scarcity, lack of access to input capital and seedlings, and low income. In the

face of land scarcity, farmers prefer particular species based on local socio-economic needs, agro-ecological suitability, growth rates to avoid competition with other food crops. However, agroforestry practices are not necessarily driven by environment concerns. This gap could be bridged by proposing agroforestry practices that contribute to environmental sustainability and meet farmer's goals without being risky. Two-way knowledge transfer between farmers, agronomists and ecologists in a participatory approach helps to encourage an agroforestry management regime that balances economic and ecological needs and provides a diversified food and cash crop livelihood strategy.

Key-Words: agricultural intensification, ecological-economic trade-offs, fuel wood, multifunctional trees, perceptions, smallholder farmer, soil fertility, water quality

Introduction

Agroforestry is defined as land-use practices that deliberately combine woody perennials with animals and/or crops on the same managed land (ICRAF, 2010). Increasingly agroforestry is viewed as providing ecosystem services, environmental benefits, and economic commodities, including net primary production, pest control (Schroth et al., 2000; Altieri & Nicholls, 2003), pollination and seed dispersal (Varah et al., 2013), soil enrichment (Ilany et al., 2010; Konig, 2007), soil stabilization and erosion control (Atangana et al., 2014a; Roose & Ndayizigiye, 1997; Narain et al., 1997), clean water (Jose, 2009; Udawatta et al., 2010), flood mitigation (Matocha et al., 2012), carbon sequestration (Montagnini & Nair, 2004), maintenance of biodiversity (McNeely, 2004) and aesthetic and cultural values. Published work provides evidence of the role of agroforestry in supporting water quality, controlling erosion, enriching soil and diversifying farming systems by producing timber, fruits and non-timber forest products (NTFPs) or other commodities (e.g. Rice, 2008). The diversity of products provided by agroforestry can in many cases help meet the needs of smallholder farmers for fuel wood, food, livestock fodder, and other household and farm needs. These products can provide additional income to supplement commodity crops and can offer flexibility in production and income to buffer the effects of changes in commodity prices and markets, crop failure, or other sources of financial or economic difficulty (Idol et al., 2011).

In most African countries, the merits of agroforestry for climate change adaptation and maintenance of ecosystem services among smallholder farmers are increasingly being recognized (Roose & Ndayizigiye, 1997; Johansson et al., 2013; Nyasimi et al., 2014; Pattanayak et al., 2003; Konig, 2007). However, in some countries government policies can

work against practices that promote maintenance of ecosystem services such as agroforestry. In Rwanda, for example, a government-mandated crop intensification program based on monocropping systems and conversion of marshlands to rice cultivation has been extensively adopted by smallholder farmers since 2007. Although the program has resulted in crop production increase since its implementation (MINAGRI, 2011), recent studies have revealed that such practices are detrimental for the environment in the long run, with adverse impacts on soil conservation and food crop production (Gomiero et al., 2011; Vaast & Somarriba, 2014; Cole et al., 2015). Existing literature on agroforestry in Rwanda has only emerged near the end of the 1900s (e.g. Yamoah & Burleigh, 1990; Yamoah & Grosz, 1988; Drechsel et al., 1996), and has focused essentially on food security and ecosystem services that support food production; little attention has been paid to the climate change adaptation element (Bucagu et al., 2013). In landscapes that are climate-sensitive, that is sensitive to the variability of climate and seasons, it is crucial to review some of the strategies geared toward climate adaptation.

In this chapter, I review the role of agroforestry in balancing human needs with environmental benefits, and discuss the potential ecological and economic benefits of appropriate management. I explore the historic and current state of agroforestry and its adoption by smallholder farmers facing climate change and government-mandated agricultural intensification policies in Rwanda, a human dominated climate sensitive landscape. I assess current agroforestry practices adopted by Rwandan farmers, and the constraints and incentives associated with such practices. Finally, I conclude the chapter with suggestions on steps towards sustainable adoption and management of agroforestry in Rwanda and other developing countries that is financially profitable and environmentally sound, including future research horizons.

Colonial and post-colonial agroforestry in Rwanda

The clearing of land for farming and felling of forests for charcoal production opened up the landscape in Rwanda for pastoralism that started around 1400 (Page & Daves, 2005). As the Rwandan population increased, more forested land was cleared for agro-pastoral activities. By the early 1900s, dense forests had decreased to about three percent of the Rwanda-Urundi territory, which included the current countries of Rwanda and Burundi, or 6.5 percent if including savanna areas (Everaerts, 1947). Reforestation consisted of planting trees in woodlots rather than incorporating them into farming systems. Colonial agronomists often considered trees more of an obstacle than an aid for agricultural development and, other than fruit trees and a few economically important woody species such as coffee, tea, and oil palm, they made no mention of trees in their descriptions of farming systems in the region and trees were merely planted in woodlands (e.g. Mortehan, 1918 & Everaerts, 1947, in den Biggelaar & Hart, 1996).

The Rwandan population density grew from 88/km² in year 1950 to 313 persons/km² in 2000, 414 in year 2012, and is predicted to reach 688/km² in year 2050 (MINECOFIN & NISR, 2012). Given the reliance of a vast majority of the Rwandan population on agriculture, and the state of land scarcity for the growing population, forest land and wetlands have been cleared for farming activities, increasing risks of erosion and flooding. For example, between 1960 and 1999, Nyungwe forest lost about 21.8% of its cover, Mukura forest lost 46.7%, Birunga forest lost 62.5% and Akagera savanna woodlands lost 66.3% (MINITERE, 2005).

In order to protect soil against erosion as a result of intensive agriculture and overgrazing, the Belgium administration agricultural service implemented an erosion control program in

1926. The agriculture program developed by the Belgian administration was conceived not only to control erosion, but also to prevent famines like those which occurred in Gashogoro in 1904, Kimwaramwara in 1906, Kazuba in 1910, Rumanura in 1917-1918, and Gakwege in 1924-1925. Recommendations included anti-erosion hedges planted in levelled curves on at-risk slopes. A program of re-forestation was thus established. It was not until the 1930s that active planting of trees began when Belgian colonists introduced non-native species such as *Eucalyptus spp., Grevillea robusta,* and *Cupressus lusitanica*, mainly in woodlots. The main purpose of the endeavor was to reforest areas that had been degraded following heavy deforestation. Everaerts (1947) estimated the total area reforested with non-native tree species at the end of 1938 at 22,000 ha, with 9,723 ha in Rwanda. Communities were obliged to establish one hectare woodlots per 300 people each year in order to combat shortages of fuel wood and timber (den Biggelaar & Hart, 1996). Therefore, these colonial mandated agroforestry approaches focused on woodlots rather than including trees in agricultural fields.

After the 1962 independence of Rwanda, tree planting slowed because farmers considered slopes to be stabilized enough by planting *Pennisetum sp.*, used to counteract erosion. The new Rwandan government of independence encouraged intercropping using *Pennisetum sp.* which was used as fodder, support for beans and as fuel material for cooking (Rossi, 1998). Intercropping here refers to the systematic planting of two or more crops on the same field. Also during this time, various agriculture development projects emphasized the establishment of terraces, the planting of species that contribute to erosion control around farmlands and the establishment of *Eucalyptus* plantations, but farmers often abandoned the practices despite awareness campaigns about the usefulness of modern farming practices because it was

considered a measure imposed by the colonial administration (Sibomana and Moeyersons, 1996). A 2010 survey of farmers' impressions of changes in tree cover during the previous 10 years in Rwanda showed little difference in the proportion of households that reported an increase in tree cover (44%) and a decline in tree cover (46%) for the whole study area, suggesting that the uptake of agroforestry has not been as extensive as is often suggested (Ndayambaje et al., 2012). Some research shows that slow adoption of some agroforestry techniques by farmers in Rwanda is explained by socio-economic factors (Ndayambaje et al., 2012; Salam, Noguchi, & Koike, 2000; Arbuckle et al., 2013; Dang et al., 2014) and limited capacity for government extension services to engage farmers effectively with respect to the uptake of the agroforestry technology.

The development of agroforestry in Rwanda is among the guiding principles of the current forest policy (MINIFOM, 2010). Since 2011, the Rwanda government has paid particular attention to agroforestry by promoting tree planting on farmlands in order to curb depletion of forest resources, declining soil fertility and environmental degradation, and to contribute to solving the rural energy crisis, land scarcity, and soil erosion problems (Ndayambaje et al., 2012). However, the limited capacity of extension agents to provide adequate agroforestry information to farmers to realize both environmental and economic objectives (MINAGRI, 2011; Stainback et al., 2011; Ndayambaje et al., 2012), combined with limited access to capital investment and microcredits among smallholder farmers (Ansoms, 2010; Huggins, 2009), scarcity of farm land, limited access to extension services and lack of seedlings (Cantore, 2011; MINAGRI, 2011; Isaacs et al., 2016) has hindered uptake of agroforestry practices. Those farmers who have planted trees favor certain tree species that are suitable to local soil and climatic conditions and which address their various needs most of which are economic in nature (e.g. Ndayambaje & Mohren, 2011),

suggesting that if these constraints can be overcome, adoption of agroforestry practices could be more successful and sustainable by accommodating environmental concerns related to soil and water conservation.

Efforts to make agroforestry viable should capitalize on the fact that most farmers are now taking steps towards incorporating tree planting into their farms (Ndayambaje et al., 2013). Trees commonly planted today in Rwanda in agricultural settings include *Leucaena sp.* and *Calliandra sp.* which grow at all elevation gradients, *Alnus acuminata* and *Sesbania sesban* which are found mainly in high elevations, and *Cassia spectabilis* which is predominant in lower elevations in Rwanda; *Eucalyptus* and *Grevilea* are found mostly in woodlots on marginal lands (MINAGRI, 1991). *Leucaena sp.* and *Calliandra sp.* are preferred by farmers in various parts of the country (Dusengemungu & Zaongo, 2006) and are believed to contribute to soil fertility restoration (Konig, 2007). Many *Eucalyptus* species have a high coppicing ability (the capacity for a tree to put out new shoots from the stump or roots when main stem is cut), which increases stock density and profitability per unit area (Bagchi & Mittal, 1996; Babitha et al., 2000; Turnbull et al., 2000; Little & Gardner, 2003; Albaugh et al., 2013).

Studies on the role of agroforestry in soil conservation in Rwanda are still scarce (e.g. Roose and Ndayizigiye, 1997; Konig, 2007), despite the concern of widespread and significant soil erosion and degradation (Roose and Ndayizigiye, 1997). The Rwandan government is deploying efforts to restore soil fertility through erosion control measures such as terracing, tree planting, increased use of mineral fertilizers but those efforts generally do not take into consideration poor smallholder farmers' financial and technical capacity needs (Ndayambaje et al., 2012) Bryan et

al., 2013; Salam et al., 2000; Hassan et al., 2008) to maximize the output within small farmland plots (Ansoms et al., 2008). This is exemplified by farmers' preferences for certain tree species that provide multiple benefits, examined in the following section.

Multifunctional role of agroforestry in Rwanda

Agroforestry in Rwanda currently consists of seven major categories: boundary planting, contour hedgerows, homegardens, silvopastoralism, woodlots, intercropping/alley cropping and scattered trees on farm (Ndayambaje & Mohren, 2011; Dusengemungu & Zaongo, 2006) (Table 2-1). In some farms a mix of agroforestry practices can be found. Literature about the adoption and distribution of each type of agroforestry practice in Rwanda is scarce (e.g. MINIRENA, 2014), and the valuation of socio-economic benefits at all geographic scales in the country is not available yet. In addition, an understanding of the relationship between various agroforestry practices and provision of environmental benefits for climate change adaptation among smallholder farmers is still lacking. Nevertheless, it is clear that agroforestry offers valuable benefits to smallholder farmers in Rwanda (Nduwamungu, 2011; Konig, 2007; Ndayambaje & Mohren, 2011).

Table 2-1

Agroforestry type	Description	Geographic distribution where research was done	Function
Boundary planting	Trees planted for delimitation between farms; used as live	Buheruka Highlands,	Erosion control (Konig, 2007) (Roose

Agroforestry Practices in Rwanda and their Socio-ecological Functions
	fencing or buffer between roads and farms	Eastern Savannah, Nile-Congo Crest, Volcanic Highlands	& Ndayizigiye, 1997), fuel wood, stakes for climbing beans, fodder (Ndayambaje et al., 2012)
Contour hedgerows	Hedges of trees planted along contour lines, on ditche edges and on cropped bench terraces for stabilization	Buberuka Highlands, Volcanic Highlands	Soil erosion control (Mutegi et al., 2008; Xu et al., 2001; König, 1992), provision of stakes for climbing beans and fodder for animals (Ndayambaje et al., 2012).
Homegardens	Mixes of non-native and indigenous trees with annual or perennial crops (e.g., maize, beans, vegetables, coffee and banana) and livestock to produce products and services in the proximity of the homesteads	Buberuka Highlands, Central Plateau, Eastern Plateau, Eastern Savannah, Nile-Congo Crest, Volcanic Highlands	Soil erosion control, stakes for climbing beans, building material, fuel wood, fodder for animals (Ekise et al., 2013; Ndayambaje et al., 2013)
Silvopastoralism	Complementary relationship between trees and pasture in a forest products and livestock production system	Eastern Savannah, Volcanic Highlands	Soil conservation, fodder for livestock (Barret, 1990; Roose & Ndayizigiye, 1997)
Woodlots	Lots of trees planted primarily for the production of fuel wood, charcoal, building poles and timber. Species typically used include <i>Eucalyptus</i> spp., <i>Cupressus</i> sp., <i>Pinus</i> sp., <i>Acacia</i> mearnsii, <i>Ficus</i> sp.,	Central Plateau, Eastern Plateau, Easter Savannah, Nile-Congo Crest,	Provision of fuel wood, other timber products, poles, stakes for climbing beans and erosion control. (Ndayambaje et al., 2013)

	<i>Mitragyna rubrostipulosa</i> , <i>Markhamia</i> sp., <i>Grevillea</i> sp., and <i>Ricinus officinalis</i>	Volcanic Highlands	
Intercropping/Alley cropping	Trees planted within agricultural fields such that the food crops are planted between hedges of trees, which are coppiced regularly to reduce competition for light	Eastern Plateau, Eastern Savannah	Erosion control, stakes for climbing beans and reduction of the use of synthetic fertilizers (Yamoah & Burleigh, 1990)
Scattered trees on farm	Trees placed randomly in fields in relatively wide spacing, with the surrounding space used for crops or pasture	Central Plateau, Eastern Plateau, Easter Savannah, Nile-Congo Crest, Volcanic Highlands	Soil fertility restoration, erosion control, provision of fuel wood and other timber products, stakes for climbing beans and fodder for livestock (Ekise et al., 2013)

Agroforestry in Rwanda has the potential to provide multiple benefits to smallholder farmers. For example, an alley cropping experiment in Rwanda with *Sesbania sesban* found that trees seemed to reduce the incidence of maize rust (*Puccinia sorhii*). Disease incidences were lower in areas with hedgerows compared with those without (Lasco, Delfino, & Espaldon, 2014). *Sesbania sesban* is found mainly in high altitudes of Rwanda, and is known for its medicinal properties (Kamel et al., 2011) in addition to providing fodder for livestock (Oosting et al., 2011; Klapwijk et al., 2014), fuel wood and stakes for climbing beans (Roose & Ndayizigiye, 1997). There are other agroforestry species found in Rwanda which perform multiple functions to meet the needs of farmers. These include *Eucalyptus* spp., *Grevilea robusta, Alnus acuminata, Calliandra calothryrsus* and *Cassia spectabilis*. Agroforestry supports pollinators that visit

agricultural crops. Given the potential of agroforestry for climate adaptation, the following paragraphs discuss challenges and opportunities pertaining to the adoption of agroforestry in the face of climate change and a recent government-mandated crop intensification program.

Environmental benefits of Agroforestry in Rwanda

Despite claims from previous studies on agroforestry and climate change in Rwanda which suggest that agroforestry is effective in strengthening smallholder farmers' ability to adapt to climate change (Stainback et al., 2012; Ndayambaje et al., 2013; Nsabimana et al., 2008; Rockwood et al., 2004), agroforestry has mostly been practiced to provide fuelwood (Ndayambaje & Mohren, 2011) and other products that support food production such as poles to support beans, and fodder for animals (Nahayo et al., 2016). Soils in these agricultural systems continue to degrade and lose fertility through erosion, and the quality of water in streams continues to decline (e.g., Wali et al., 2011; Nhapi, 2011). However, studies to accurately quantify the impact of agroforestry intervention on soil fertility, erosion and pest control and water quality management are still limited (e.g. (Roose & Ndayizigiye, 1997).

Recent studies on agroforestry in Rwanda emphasize the need to implement agroforestry in farmlands to improve soil productivity (Habiyaremye et al., 2015), and to improve the microclimate within agricultural landscapes for the improvement of crop performance as trees can buffer climatic extremes that affect crop growth (Atangana et al., 2014a; Mbow et al., 2014; Chaudhury et al., 2011; Bishaw et al., 2013; De Souza et al., 2012; Negra, 2013). However, research on agroforestry in Rwanda has not fully covered the whole spectrum of potential environmental benefits such as pollination, pest control, and non-point nutrient-associated

pollution control, and field research involving farmers to demonstrate how agroforestry can practically provide such environmental benefits is still limited. At the same time, it is clear that integration of environmental goals into agroforestry policy opens up opportunities for sustainable intensification of agriculture.

Smallholder farmers and Agroforestry: opportunities and challenges to adoption

The government-mandated crop intensification (CI) program initiated in 2007 in Rwanda is aimed at transforming subsistence agriculture into a productive, high value, market oriented sector (Kaberuka et al., 2000; Pritchard, 2013). CI is characterized by a monocropping system associated with intensive use of synthetic fertilizers and conversion of marshlands into rice plantations. The program has resulted in an increase in crop production ever since; corn, beans, Irish potatoes and rice have particularly increased in production (MINAGRI, 2011).

While the CI program seems to perform satisfactorily in terms of increased crop production, scientists have expressed concern over long-term environmentally adverse impacts of such short-term benefit agricultural practices and these impacts are likely to be exacerbated by the effects of climate change (Lasco et al., 2014; Midega et al., 2015; Hassan et al., 2008). A number of studies predict an increase of rainfall and temperature in Rwanda (e.g., Asumadu-Sarkodie et al., 2015; Shongwe et al., 2011; Stocker, 2014; REMA, 2011; McSweeney et al., 2011). Rainfall projections predict an increase of 20% by 2050 and 30% by 2080 (Asumadu-Sarkodie et al., 2015). Mean precipitation for the short rainy season of October-December is projected to increase by more than 10% (Shongwe et al. (2011). There is still no consensus about the

magnitude of rainfall increase and some analyses even suggest a slight decline in rainfall (Henninger, 2013).

As a result of climate change, increased vulnerability of agricultural lands to soil erosion and floods is predicted and has started to be experienced, particularly in the highlands of the country (Nahayo, Ekise, & Niyigena, 2013; Zimmerman et al., 2012; REMA, 2011; Musanze District, 2013; SEI, 2009; Wali et al., 2011; Kagabo et al., 2013; Pritchard, 2013). Although studies on climate change impacts on crops are still limited (e.g. Jarvis et al. , 2012), the increase in temperature and precipitation is likely to favor the incidence of pests among crops (Chakraborty et al., 2000; Ghini et al., 2008) which, in a CI system, would require the use of more pesticides. Potatoes, maize and beans, which are among the staple foods for most Rwandans, are predicted to be affected by pests as a result of temperature increases (Jarvis et al., 2012). In addition, increases in precipitation coupled with intensive use of synthetic fertilizers is likely to increase nutrient leaching which impacts water quality in streams and lakes (e.g. Wali et al., 2011; Kimwaga et al., 2012) and land degradation.

Agroforestry has been proposed as a strategy in climate change adaptation within agricultural landscapes (Hassan et al., 2008). In climate-sensitive areas such as Rwanda , it has been recommended that soil loss, which poses a severe threat to agriculture (Konig, 2007) may be dramatically reduced through hedgerow/agroforestry contour planting and improved mulching practices (Verchot et al., 2007). However, agroforestry is more effective in agricultural landscapes characterized by intercropping systems. Results from a study that analyzed determinants of farm-level climate adaptation measures from 11 African countries showed that

specialized crop cultivation (mono-cropping) is the agricultural practice most vulnerable to climate change in Africa (Hassan et al., 2008). A recent assessment of the value of diverse cropping systems under the new agricultural policy environment in Rwanda (Isaacs et al., 2016) demonstrated that an improved intercropping system tends to outperform the government-mandated system of alternating single-cropped bean and maize season-by-season.

Despite the potential for agroforestry to provide environmental benefits that can enhance food productivity, its adoption by smallholder farmers is still challenged by a number of factors including scarcity of arable land per household and concerns about high competition for land between forestry and agriculture (Ndayambaje et al., 2013; MINIFOM, 2010), and rampant poverty (Malyon, 2014). In addition, support from extension services which are the main source of agricultural information has so far remained ineffective (Ndayambaje et al., 2012; Stainback et al., 2011) as will be discussed below. Furthermore, while micro-credit schemes in Rwanda, for example, are widespread and are believed to contribute to the efficiency of agricultural production, credit is virtually inaccessible to smallholder poor and uneducated farmers (Nabahungu & Visser, 2013; Ali et al., 2014; Musanze, 2013; Mutandwa & Kanyarukiga, 2016). A study of farmers' perceptions of agroforestry and the restoration of ecological and economic productivity of agricultural land revealed that agroforestry is considered by farmers as too risky (Verdone & Seidl, 2016). The priority of smallholder farmers is to maximize crop production on their land, and they find it risky to invest in woodlots or agroforestry (Atangana et al., 2014; Dyer, 2014; Ali & Deininger, 2015; Chand et al., 2011). Efforts to improve farming systems, including agroforestry, require innovative and inclusive approaches that enable adaptation to the socio-ecological realities of farmers (Isaacs et al., 2016; Larochelle et al., 2016) which can vary

from place to place. Agroforestry is a very "knowledge intensive" endeavor (Place et al., 2012). Farmers must learn new skills which takes time, effort and money (Pye-Smith, 2010). Therefore, since profitability is a major driver in adoption of climate adaptation strategies among farmers particularly with respect to the choice of farming systems and crop choice (Chouinard et al., 2008; Musshoff & Hirschauer, 2008; Cary & Wilkinson, 1997), the success of the adoption of agroforestry hinges on its profitability. Financial incentives coupled with support from extension services should lead to profitability.

To ensure the success of agroforestry adoption among poor smallholder farmers, it is critically important to investigate the economic returns from the integration of trees with agriculture in addition to providing financial incentives and technological support. Choosing between agriculture and tree crops can be facilitated by a forest policy that supports strategies that might include mainstreaming agroforestry in agricultural policies, promoting best management practices of trees and woodlots on farms, building capacity of farmers, providing incentives for tree planting, and developing and disseminating farm forestry information and technology (Ndayambaje et al., 2013).

The forest policy could capitalize on benefits that agroforestry could provide to farmers in addition to environmental outcomes. Adoption of agroforestry in Rwanda is possible as long as the ways that it contributes to resolving the increasing demand for fuelwood and other wood products (Kalinganire, 1996; Bucagu et al., 2013) among rural communities, the majority of which are farmers (Ndayambaje & Mohren, 2011) can be demonstrated without compromising food crop production (Ndayambaje et al, 2013). Given the limited capacity among most

smallholder farmers to acquire pesticides and synthetic fertilizers, extension services could, through field-based experiments on farmers' land and through farmer-to-farmer extension efforts, collaboratively demonstrate how agroforestry, in association with efficient cropping systems, can significantly cut the use of those chemicals. To bring a positive, substantial impact to farmers, and bring them into durable adoption of agroforestry in the case of this review, requires that experiments involve farmers directly, taking into account smallholder farmers' objectives and constraints (Isaacs et al., 2016).

New perspectives in agroforestry in Rwanda

The economic development and poverty reduction policy documents in Rwanda set clear targets to increase national forest cover to 30% of total land area equivalent to 790,140 ha of forest and to achieve agroforestry cover of 85% of arable lands by 2020. However, due to high population density (507 inhabitants/km2 in 2017), farming land per household is shrinking resulting in high competition for land between forestry and agriculture (MINIFOM, 2010). One of the objectives of the forestry policy is to "contribute to sustainable land use through soil, water and biodiversity conservation, and tree planting through the sustainable management of forests and trees" (MINIFOM, 2010). As farm land per household decreases and soils become exhausted, crop cultivation is being pushed into marginal areas, particularly on steep slopes and riparian zones, leading to widespread landslides, soil erosion and siltation of water bodies (MINIFOM, 2010). Restoring soil conservation on marginal lands and riparian zones through agroforestry are important steps for sustaining food production while fostering other environmental and economic benefits.

Past and most current research on agroforestry in Rwanda has been conducted mostly in experimental fields owned by research institutions (e.g. Dusengemungu & Zaongo, 2006), and capacity building in agroforestry has been provided mainly to government entities without including smallholder farmers (Nduwamungu, 2011). Rwandan policy makers have recognized a need to integrate agroforestry into agrarian policies and strategic plans, and efforts have been made to disseminate best agroforestry practices among farmers in Rwanda by developing farmer and private sector capacity to implement agroforestry effectively, to intensify research on agroforestry technologies and plant species suitable for various agro-ecological zones (Nduwamungu, 2011). While these efforts are valuable, scientists have highlighted the necessity to incorporate farmers' input into the adoption of agroforestry technology. Not only do farmers have knowledge about suitability of local tree species to local geophysical and socio-economic conditions of their area, but also their motivation to plant trees will largely depend on their specific goals and needs (e.g. Ndayambaje et al., 2012; Nahayo et al., 2013).

Extension services and agroforestry adoption in Rwanda

A number of studies in sub-Saharan African have shown that the effectiveness of extension services, in addition to credit access and land, are key to building farmers' capacity and catalyzing their decision to adapt to climate change (Bryan et al., 2009; Hassan et al., 2008; Franz et al., 2010) through the adoption of new agricultural practices that include agroforestry. Extension services often do not reach poor marginalized farmers including women, minorities, and people in very remote areas (Anandajayasekeram et al., 2007). Challenges pertaining to access to good extension services can also be seen in Rwanda.

The important role of extension services in supporting smallholder farmer adoption of agroforestry in Rwanda has been noted in recent publications (Muhongayire et al., 2013; Andrew & Masozera, 2010). However, while the number of extension services has increased in Rwanda, their efficiency has been questioned (Stainback et al., 2011). Given their substantial contribution to wood products in Rwanda, Rwandan smallholder farmers should be well equipped with knowledge and technical skills needed to put in place sustainable agroforestry that accommodates both environmental benefits and economic benefits. Extension services efforts coupled with formal training are some of key channels for the acquisition of knowledge about environmental benefits of farming technologies (e.g., Hussain et al., 1994; Feder & Slade, 1984). Successful agroforestry requires, among other things, that extension services accommodate the variability farmers' specific needs in different regions and social groups across the country (Stainback & Masozera, 2010).

Forest and agrarian policy: engagement of farmers

Farmers should be actively involved in the agrarian reform policy making process so that their goals, objectives and socio-ecological environment are taken into account (Harrison, 2016; Isaacs et al., 2016; Larochelle et al., 2016), including being made aware of agroforestry options (Nsengiyumva et al., 2016; Mutandwa & Kanyarukiga, 2016; Nahayo et al., 2016). Farmers can begin to understand how important environmental benefits can be secured through agroforestry without compromising economic profits normally obtained from their farmlands. Procuring environmental benefits from agroforestry such as soil conservation, nutrient recycling, nonpoint pollution control, pest control, biodiversity conservation, water quality and many others requires long-term planning and goals (Nduwamungu, 2011). Given the fact that the majority of

Rwandan farmer households own less than 0.6 ha of farm land, it is logical to suggest that farmer priorities are oriented towards the immediate goals of food crop production and other strategies that support food production.

Forest policy can support adoption of agroforestry practices through strategies that integrate farm forestry in agricultural policies, promoting best management practices of trees and woodlots on farms, building capacity of farmers (Konig, 2007; Rushemuka et al., 2014), providing incentives for tree planting, and developing and disseminating farm forestry techniques in a unified extension system for all agroecological zones of the country (Ndayambaje et al., 2013). Of the numerous published agroforestry research activities conducted in Rwanda over the last 20 years, few have engaged with farmers through the use of participatory research methods (Bucagu et al., 2013).

Conclusion

Agroforestry is increasingly being recognized as an important adaptation and mitigation strategy against the negative effects of climate change. In particular, the use of multi-purpose trees has been promoted as an agroforestry practice that contributes to the improvement of soil fertility and food crop yields. While the importance of agroforestry is widely discussed, research on how agroforestry can be adopted successfully by reconciling socio-economic and environmental outcomes is still limited. In this study, I reviewed existing literature to examine opportunities for reconciling environmental and economic goals of agroforestry adoption. Historically, agroforestry has emphasized the plantation of woodlots on agricultural lands more than the association of trees and food crops. Furthermore, development and dissemination of agroforestry

practices has typically not engaged the farmers themselves in a participatory process. Research on the contributions of various agroforestry practices to environmental integrity and crop yield improvement is lacking. Agrarian policies could make agroforestry adoption more effective by accommodating farmers' goals, objectives and socio-ecological environment and by involving farmers through demonstration plots to assess how agroforestry can successfully integrate economic and environmental outcomes.

There is a need for research to accurately assess the performance of different agroforestry practices in different contexts in improving soil fertility, supporting pollination processes, controlling soil erosion, abating pest incidence, controlling water quality and flow, while providing other socio-economic benefits such as fuel wood, fodder for livestock, building materials, and stakes for climbing beans. This research should be participatory by involving farmers themselves in order to subsequently facilitate the adoption and the dissemination of the technology.

To ensure that the full potential of agroforestry to provide environmental benefits, ecosystem services and food security is realized, there is a need for research to ensure that variation in farmer social, economic and environmental context is considered to improve uptake and adaptation. In this respect, studies to assess the relationship between tree species that farmers prefer and their environmental and economic function are needed to develop agroforestry practices tailored to farmers' economic needs and which also satisfy environmental parameters and anticipated outcomes. Agroforestry may be successful if factors that contributed to the slow adoption of tree planting on farms since the colonial era are considered seriously and if farmers'

input in the choice of agroforestry practices and trees threes of their choice incorporated.

Engagement of farmers requires active support from agricultural extension services.

References

- Albaugh, J. M., Dye, P. J., & King, J. S. (2013). Eucalyptus and Water Use in South Africa [Research article].
- Ali, D. A., Deininger, K., & Duponchel, M. (2014). Credit Constraints and Agricultural Productivity: Evidence from rural Rwanda. *Journal of Development Studies*, 50(5), 649– 665.
- Ali, D. A., & Deininger, K. W., --. (2015). Is There a Farm Size–Productivity Relationship in African Agriculture? Evidence from Rwanda. *Land Economics*, *91*(2), 317–343.
- Altieri, M. A., & Nicholls, C. I. (2003). Soil fertility management and insect pests: harmonizing soil and plant health in agroecosystems. *Soil and Tillage Research*, *72*(2), 203–211.
- Anandajayasekeram, P., Davis, K. E., & Workneh, S. (2007). Farmer field schools: an alternative to existing extension systems? Experience from Eastern and Southern Africa. *Journal of International Agricultural and Extension Education*, 14(1), 81–93.
- Andrew, G., & Masozera, M. (2010). Payment for ecosystem services and poverty reduction in Rwanda. *J Sustain Develop Afr*, *12*, 122–139.
- Ansoms, A. (2010). Views from Below on the Pro-poor Growth Challenge: The Case of Rural Rwanda. *African Studies Review*, *53*(2), 97–123.
- Arbuckle, J. G., Morton, L. W., & Hobbs, J. (2013). Farmer beliefs and concerns about climate change and attitudes toward adaptation and mitigation: Evidence from Iowa. *Climatic Change*, 118(3–4), 551–563.
- Asumadu-Sarkodie, S., Rufangura, P., Jayaweera, M. P. C., & Owusu, P. A. (2015). Situational analysis of flood and drought in Rwanda. *International Journal of Scientific and Engineering Research*, 6(8), 960.

- Atangana, A., Khasa, D., Chang, S., & Degrande, A. (2014a). Agroforestry for Soil Conservation. In *Tropical Agroforestry* (pp. 203–216). Springer Netherlands.
- Atangana, A., Khasa, D., Chang, S., & Degrande, A. (2014b). Major Agroforestry Systems of the Humid Tropics. In *Tropical Agroforestry* (pp. 49–93). Springer Netherlands.
- Babitha, M., Sreenivasulu, A., Meru, E., Rao, P. S., & others. (2000). Regenerated shoot number and coppicing ability of two year old Eucalyptus tereticornis stumps of different girth classes. *Indian Forester*, *126*(7), 721–726.
- Bagchi, S. K., & Mittal, M. C. (1996). Regenerated shoot number (coppicing ability) after pruning at different height levels in one year old Eucalyptus (Mysore-gum). *Indian Forester*, 122(8), 731–733.
- Bishaw, B., Neufeldt, H., Mowo, J., Abdelkadir, A., Muriuki, J., Dalle, G., ... others. (2013).
 Farmers' strategies for adapting to and mitigating climate variability and change through agroforestry in Ethiopia and Kenya. *Forestry Communications Group, Oregon State University, Corvallis, Oregon*. Retrieved from

https://www.researchgate.net/profile/Badege_Bishaw/publication/271767422_Farmers'_s trategies_for_Adapting_to_Mitigating_Climate_Variability_and_Change_through_Agrof orestry_in_Ethiopia_and_Kenya/links/550885780cf26ff55f82e2b1.pdf

- Bryan, E., Deressa, T. T., Gbetibouo, G. A., & Ringler, C. (2009). Adaptation to climate change in Ethiopia and South Africa: options and constraints. *Environmental Science & Policy*, 12(4), 413–426.
- Bryan, E., Ringler, C., Okoba, B., Roncoli, C., Silvestri, S., & Herrero, M. (2013). Adapting agriculture to climate change in Kenya: Household strategies and determinants. *Journal* of Environmental Management, 114, 26–35.

- Bucagu, C., Vanlauwe, B., Wijk, M. T., & Giller, K. E. (2013). Assessing farmers' interest in agroforestry in two contrasting agro-ecological zones of Rwanda. *Agroforestry Systems*, 87(1), 141–158.
- Cantore, N. (2011). The crop intensification program in Rwanda: a sustainability analysis.
 Overseas Development Institute.[Available Online]: Http://Dspace. Cigilibrary.
 Org/Jspui/Bitstream/123456789/31634/1/ODI-% 20The% 20Crop% 20Intensification%
 20Program% 20in% 20Rwanda. Pdf, 1. Retrieved from
 http://dspace.africaportal.org/jspui/bitstream/123456789/31634/1/ODI-%20The%20Crop%20Intensification%20Program%20in%20Rwanda.pdf
- Cary, J. W., & Wilkinson, R. L. (1997). Perceived profitability and farmers 'conservation behaviour. *Journal of Agricultural Economics*, 48(1–3), 13–21.
- Chakraborty, S., Tiedemann, A. V., & Teng, P. S. (2000). Climate change: potential impact on plant diseases. *Environmental Pollution*, *108*(3), 317–326.
- CHAND, R., PRASANNA, P. A. L., & SINGH, A. (2011). Farm Size and Productivity:
 Understanding the Strengths of Smallholders and Improving Their Livelihoods.
 Economic and Political Weekly, 46(26/27), 5–11.
- Change, I. P. on C. (2014). Climate Change 2013 The Physical Science Basis: Working Group I Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Chaudhury, M., Ajayi, O. C., Hellin, J., & Neufeldt, H. (2011). Climate change adaptation and social protection in agroforestry systems: enhancing adaptive capacity and minimizing risk of drought in Zambia and Honduras. *Documento de Trabajo*, (137). Retrieved from http://www.worldagroforestry.org/downloads/Publications/PDFS/WP11269.pdf

- Chouinard, H. H., Paterson, T., Wandschneider, P. R., & Ohler, A. M. (2008). Will Farmers Trade Profits for Stewardship?: Heterogeneous Motivations for Farm Practice Selection. *Land Economics*, 4(1), 66–82.
- Cole, L. J., Brocklehurst, S., Robertson, D., Harrison, W., & McCracken, D. I. (2015). Riparian buffer strips: their role in the conservation of insect pollinators in intensive grassland systems. *Agriculture, Ecosystems & Environment, 211*, 207–220.
- Dang, H., Li, E., Bruwer, J., & Nuberg, I. (2014). Farmers' perceptions of climate variability and barriers to adaptation: lessons learned from an exploratory study in Vietnam. *Mitigation & Adaptation Strategies for Global Change*, 19(5), 531–548.
- de Dieu Habiyaremye, J., Muthuri, C., Matiru, V., Nyaga, J., Mukuralinda, A., Ruganzu, V., ... Sinclair, F. (2015). Occurrence and abundance of arbuscular mycorrhizal fungi (AMF) in agroforestry systems of Rubavu and Bugesera Districts in Rwanda. *African Journal of Microbiology Research*, 9(12), 838–846.
- De Souza, H. N., de Goede, R. G., Brussaard, L., Cardoso, I. M., Duarte, E. M., Fernandes, R.
 B., Pulleman, M. M. (2012). Protective shade, tree diversity and soil properties in coffee agroforestry systems in the Atlantic Rainforest biome. *Agriculture, Ecosystems & Environment*, 146(1), 179–196.
- den Biggelaar, C., & Hart, N. (1996). Farmer experimentation and innovation. A case study of knowledge generation processes in agroforestry systems in Rwanda. *Community Forestry Case Study Series (FAO)*.
- Drechsel, P., Steiner, K. G., & Hagedorn, F. (1996). A review on the potential of improved fallows and green manure in Rwanda. *Agroforestry Systems*, *33*(2), 109–136.

- Dusengemungu, L., & Zaongo, C. (2006). Etat de la recherche agroforestière au Rwanda, etude bibliograhpique, période ; 1987 – 2003. ICRAF Working Paper – no. 30. Nairobi : World Agroforestry Centre.
- Dyer, G. (2014). *Class, state and agricultural productivity in Egypt: study of the inverse relationship between farm size and land productivity*. Routledge. Retrieved from https://books.google.com/books?hl=en&lr=&id=73UBAwAAQBAJ&oi=fnd&pg=PR9& dq=+Class,+state+and+agricultural+productivity+in+Egypt:+A+study+of+the+inverse+r elationship+between+farm+size+and+land+productivity&ots=BXLvT3_fo5&sig=ti5WC C5D5k9GYMq2W7Rek_TZcE0
- Everaerts, E. (1947). *Monographie agricole du Ruanda-Urundi*. Ministère des Colonies, Direction générale de l'Agriculture.
- Feder, G., & Slade, R. (1984). The Acquisition of Information and the Adoption of New Technology. *American Journal of Agricultural Economics*, 66(3), 312–320.
- Franz, N. K., Piercy, F., Donaldson, J., Westbrook, J., & Richard, R. (2010). Farmer, agent, and specialist perspectives on preferences for learning among today's farmers. *Journal of Extension*, 48(3), 3RIB1.
- Ghini, R., Hamada, E., & Bettiol, W. (2008). Climate change and plant diseases. *Scientia Agricola*, 65(SPE), 98–107.
- Gomiero, T., Pimentel, D., & Paoletti, M. G. (2011). Environmental impact of different agricultural management practices: conventional vs. organic agriculture. *Critical Reviews in Plant Sciences*, *30*(1–2), 95–124.
- Harrison, G. (2016). Rwanda: an agrarian developmental state? *Third World Quarterly*, *37*(2), 354–370.

- Hassan, R., Nhemachena, C., & others. (2008). Determinants of African farmers' strategies for adapting to climate change: Multinomial choice analysis. *African Journal of Agricultural* and Resource Economics, 2(1), 83–104.
- Henninger, S. M. (2013). Does the global warming modify the local Rwandan climate? Retrieved from http://file.scirp.org/Html/4-8301848_27614.htm
- Huggins, C. (2009). Agricultural Policies and Local Grievances in Rural Rwanda. *Peace Review*, 21(3), 296–303.
- Hussain, S. S., Byerlee, D., & Heisey, P. W. (1994). Impacts of the training and visit extension system on farmers' knowledge and adoption of technology: Evidence from Pakistan. *Agricultural Economics*, 10(1), 39–47.
- ICRAF (International Center for Research in Agroforestry). (2010). Agroforestry defined. Retrieved from http://www.ciesin.columbia.edu/IC/icraf/agrodef.html
- Idol, T., Haggar, J., & Cox, L. (2011). Ecosystem services from smallholder forestry and agroforestry in the tropics. In *Integrating agriculture, conservation and ecotourism: examples from the field* (pp. 209–270). Springer.
- Ilany, T., Ashton, M. S., Montagnini, F., & Martinez, C. (2010). Using agroforestry to improve soil fertility: effects of intercropping on *Ilex paraguariensis* (yerba mate) plantations with *Araucaria angustifolia*. *Agroforestry Systems*, 80(3), 399–409.
- Isaacs, K. B., Snapp, S. S., Chung, K., & Waldman, K. B. (2016). Assessing the value of diverse cropping systems under a new agricultural policy environment in Rwanda. *Food Security*, 8(3), 491–506.
- Jarvis, A., Ramirez-Villegas, J., Campo, B. V. H., & Navarro-Racines, C. (2012). Is Cassava the Answer to African Climate Change Adaptation? *Tropical Plant Biology*, 5(1), 9–29.

- Johansson, K.-E., Axelsson, R., Kimanzu, N., Sassi, S. O., Bwana, E., & Otsyina, R. (2013). The pattern and process of adoption and scaling up: Variation in project outcome reveals the importance of multilevel collaboration in agroforestry development. *Sustainability*, 5(12), 5195–5224.
- Jose, S. (2009). Agroforestry for ecosystem services and environmental benefits: an overview. *Agroforestry Systems*, 76(1), 1–10.
- KABERUKA, D., & others. (2000). Rwanda Vision 2020. Retrieved from https://repositories.lib.utexas.edu/bitstream/handle/2152/5071/4164.pdf?sequence=1
- Kagabo, D. M., Stroosnijder, L., Visser, S. M., & Moore, D. (2013). Soil erosion, soil fertility and crop yield on slow-forming terraces in the highlands of Buberuka, Rwanda. *Soil and Tillage Research*, 128, 23–29.
- Kalinganire, A. (1996). Performance of Grevillea robusta in plantations and on farms under varying environmental conditions in Rwanda. *Forest Ecology and Management*, 80(1), 279–285.
- Kamel, E. G., El-Emam, M. A., Mahmoud, S. S. M., Fouda, F. M., & Bayaumy, F. E. (2011). Parasitological and biochemical parameters in Schistosoma mansoni-infected mice treated with methanol extract from the plants Chenopodium ambrosioides, Conyza dioscorides and Sesbania sesban. *Parasitology International*, 60(4), 388–392.
- Kimwaga, R. J., Mashauri, D. A., Bukirwa, F., Banadda, N., Wali, U. G., & Nhapi, I. (2012).
 Development of best management practices for controlling the non-point sources of pollution around Lake Victoria using SWAT Model: A Case of Simiyu catchment Tanzania. *Eng. J*, *5*, 77–83.

- Klapwijk, C. J., Bucagu, C., van Wijk, M. T., Udo, H. M. J., Vanlauwe, B., Munyanziza, E., & Giller, K. E. (2014). The 'One cow per poor family'programme: Current and potential fodder availability within smallholder farming systems in southwest Rwanda. *Agricultural Systems*, 131, 11–22.
- König, D. (1992). The potential of agroforestry methods for erosion control in Rwanda. *Soil Technology*, 5(2), 167–176.
- Konig, D. (2007). Contribution de l'agroforesterie à la conservation de la fertilité des sols et à la lutte contre le réchauffement climatique au Rwanda. Pré-Actes des JSIRAUF, Hanoi. Retrieved from http://www.infotheque.info/fichiers/JSIR-AUF-

Hanoi07/articles/AJSIR_3-5_Koenig.pdf

- Larochelle, C., Dorene, A.-M., Ekin, B., & Jeffrey, A. (2016). *Assessing the adoption of improved bean varieties in Rwanda and the role of varietal attributes in adoption decisions*. Intl Food Policy Res Inst.
- Lasco, R. D., Delfino, R. J. P., Catacutan, D. C., Simelton, E. S., & Wilson, D. M. (2014).
 Climate risk adaptation by smallholder farmers: the roles of trees and agroforestry.
 Current Opinion in Environmental Sustainability, 6(Supplement C), 83–88.
- Lasco, R. D., Delfino, R. J. P., & Espaldon, M. L. O. (2014). Agroforestry systems: helping smallholders adapt to climate risks while mitigating climate change. *Wiley Interdisciplinary Reviews: Climate Change*, 5(6), 825–833.
- Little, K. M., & Gardner, R. A. (2003). Coppicing ability of 20 Eucalyptus species grown at two high-altitude sites in South Africa. *Canadian Journal of Forest Research*, 33(2), 181–189.

Malyon, S. (2014). Feeding Rwanda's livestock revolution. Appropriate Technology, 41(4), 48.

- Matocha, J., Schroth, G., Hills, T., & Hole, D. (2012). Integrating climate change adaptation and mitigation through agroforestry and ecosystem conservation. In *Agroforestry-the future of global land use* (pp. 105–126). Springer. Retrieved from http://link.springer.com/10.1007/978-94-007-4676-3_9
- Mbow, C., Smith, P., Skole, D., Duguma, L., & Bustamante, M. (2014). Achieving mitigation and adaptation to climate change through sustainable agroforestry practices in Africa. *Current Opinion in Environmental Sustainability*, 6, 8–14.
- Mbow, C., Van Noordwijk, M., Luedeling, E., Neufeldt, H., Minang, P. A., & Kowero, G.
 (2014). Agroforestry solutions to address food security and climate change challenges in Africa. *Current Opinion in Environmental Sustainability*, *6*, 61–67.
- McNeely, J. A. (2004). Nature vs. nurture: managing relationships between forests, agroforestry and wild biodiversity. *SpringerLink*, 155–165.
- McSweeney, R., Cole, M., Semafara, J. N., & Washington, R. (2011). Rwanda's Climate: Observations and Projections, Appendix E. Smith School of Enterprise (SSEE) and the Environment, University of Oxford. Oxford.
- Midega, C. A. O., Bruce, T. J. A., Pickett, J. A., Pittchar, J. O., Murage, A., & Khan, Z. R. (2015). Climate-adapted companion cropping increases agricultural productivity in East Africa. *Field Crops Research*, 180(Supplement C), 118–125.
- MINAGRI. (2011). Sustainable Crop Intensification in Rwanda. Republic of Rwanda, Ministry of Agriculture and Animal Resources. Retrieved from http://www.minagri.gov.rw/fileadmin/user_upload/documents/CIP/CIP_Strategies_2011. pdf

- MINIFOM. (2010, May). National Forestry Policy. Ministry of Forestry and Mines. Retrieved from http://rnra.rw/uploads/media/final_national_forestry_policy_2011f.pdf
- Montagnini, F., & Nair, P. K. R. (2004). Carbon sequestration: an underexploited environmental benefit of agroforestry systems. *Agroforestry Systems*, *61*(1–3), 281.
- Mortehan. (1918). L'Agriculture au Ruanda-Urundi: notes techniques. Ministère des Colonies, Direction générale de l'Agriculture.
- Muhongayire, W., Hitayezu, P., Mbatia, O. L., & Mukoya-Wangia, S. M. (2013). Determinants of Farmers' Participation in Formal Credit Markets in Rural Rwanda. *Journal of Agricultural Sciences*, 4(2), 87–94.
- Musanze District. (2013). Musanze development plan (2013-2018). Kigali: Government of Rwanda.
- Musanze, R. (2013). Determinants of Farmers' Participation in Formal Credit Markets in Rural Rwanda. Retrieved from http://www.krepublishers.com/02-Journals/JAS/JAS-04-0-000-13-Web/JAS-04-2-000-13-Abst-PDF/JAS-04-2-087-13-081-Hitayezu-P/JAS-04-2-087-13-081-Hitayezu-P-05-Tt.pdf
- Musshoff, O., & Hirschauer, N. (2008). Adoption of organic farming in Germany and Austria: an integrative dynamic investment perspective. *Agricultural Economics*, *39*(1), 135–145.
- Mutandwa, E., & Kanyarukiga, R. (2016). Understanding the role of forests in rural household economies: experiences from the Northern and Western provinces of Rwanda. *Southern Forests: A Journal of Forest Science*, 0(0), 1–8.
- Mutegi, J. K., Mugendi, D. N., Verchot, L. V., & Kung'u, J. B. (2008). Combining napier grass with leguminous shrubs in contour hedgerows controls soil erosion without competing with crops. *Agroforestry Systems*, 74(1), 37–49.

- Nabahungu, N. L., & Visser, S. M. (2013). Farmers' Knowledge and Perception of Agricultural Wetland Management in Rwanda. *Land Degradation & Development*, *24*(4), 363–374.
- Nahayo, A., Ekise, I. E., & Niyigena, D. (2013a). Assessment of the contribution of Non Timber
 Forest Products to the improvement of local people's livelihood in Kinigi sector,
 Musanze District, Rwanda. *Ethiopian Journal of Environmental Studies and Management*, 6(6), 698–706.
- Nahayo, A., Ekise, I. E., & Niyigena, D. (2013b). Assessment of the contribution of Non Timber
 Forest Products to the improvement of local people's livelihood in Kinigi sector,
 Musanze District, Rwanda. *Ethiopian Journal of Environmental Studies and Management*, 6(6), 698–706.
- Nahayo, Alphonse, Pan, G., & Joseph, S. (2016). Factors influencing the adoption of soil conservation techniques in Northern Rwanda. *Journal of Plant Nutrition and Soil Science*, 179(3), 367–375.
- Narain, P., Singh, R. K., Sindhwal, N. S., & Joshie, P. (1997). Agroforestry for soil and water conservation in the western Himalayan Valley Region of India 1. Runoff, soil and nutrient losses. *Agroforestry Systems*, 39(2), 175–189.
- Ndayambaje, J. D., Heijman, W. J. M., & Mohren, G. M. J. (2012). Household Determinants of Tree Planting on Farms in Rural Rwanda. *Small-Scale Forestry*, *11*(4), 477–508.
- Ndayambaje, J. D., & Mohren, G. M. J. (2011). Fuelwood demand and supply in Rwanda and the role of agroforestry. *Agroforestry Systems*, *83*(3), 303–320.
- Ndayambaje, Jean Damascene, Heijman, W. J. M., & Mohren, G. M. J. (2013). Farm woodlots in rural Rwanda: purposes and determinants. *Agroforestry Systems*, 87(4), 797–814.

- Nduwamungu, J. (2011). Plantations forestières et îlots boisés au Rwanda. In *African Forest Forum*. Retrieved from http://afforum.org/sites/default/files/French/French_5.pdf
- Negra, C. (2013). *Agriculture and climate change in national green growth strategies* (Working Paper). Retrieved from https://cgspace.cgiar.org/handle/10568/33278
- Nhapi, I. (2011). Assessment of Water Pollution Levels in the Nyabugogo Catchment, Rwanda. *The Open Environmental Engineering Journal*, 4(1), 40–53.
- Nsabimana, D., Klemedtson, L., Kaplin, B. A., & Wallin, G. (2008). Soil carbon and nutrient accumulation under forest plantations in southern Rwanda. *African Journal of Environmental Science and Technology*, *2*(6), 142–149.
- Nsengiyumva, G., Kagabo, D. M., Birachi, E. A., & Hansen, J. (2016). Planning workshop for Rwanda Climate Services for Agriculture project. Retrieved from https://cgspace.cgiar.org/handle/10568/76248
- Nyasimi, M., Amwata, D., Hove, L., Kinyangi, J., & Wamukoya, G. (2014). Evidence of impact: Climate-smart agriculture in Africa. Retrieved from https://cgspace.cgiar.org/handle/10568/51374
- Oosting, S. J., Mekoya, A., Fernandez-Rivera, S., & Van der Zijpp, A. J. (2011). Sesbania sesban as a fodder tree in Ethiopian livestock farming systems: Feeding practices and farmers' perception of feeding effects on sheep performance. *Livestock Science*, *139*(1), 135–141.
- Page, W. F., & Daves, J. / T. L. S. R. H. (2005). Encyclopedia of African History and Culture, 5 Vol. Set (Revised edition). New York NY.
- Pattanayak, S. K., Mercer, D. E., Sills, E., & Yang, J.-C. (2003). Taking stock of agroforestry adoption studies. *Agroforestry Systems*, 57(3), 173–186.

- Place, F., Ajayi, O. C., Torquebiau, E., Detlefsen, G., Gauthier, M., & Buttoud, G. (2012). Improved policies for facilitating the adoption of agroforestry. In *Agroforestry for Biodiversity and Ecosystem Services-Science and Practice*. InTech. Retrieved from https://www.intechopen.com/download/pdf/34871
- Pritchard, M. F. (2013). Land, power and peace: Tenure formalization, agricultural reform, and livelihood insecurity in rural Rwanda. *Land Use Policy*, 30(1), 186–196.
- REMA. (2011). Atlas of Rwanda's changing environment: implications for climate change resilience. Kigali, Rwanda: REMA.
- Rice, R. A. (2008). Agricultural intensification within agroforestry: the case of coffee and wood products. *Agriculture, Ecosystems & Environment, 128*(4), 212–218.
- Rockwood, D. L., Naidu, C. V., Carter, D. R., Rahmani, M., Spriggs, T. A., Lin, C., Segrest, S. A. (2004). Short-rotation woody crops and phytoremediation: Opportunities for agroforestry? In *New Vistas in Agroforestry* (pp. 51–63). Springer. Retrieved from http://link.springer.com/chapter/10.1007/978-94-017-2424-1
- Roose, E., & Ndayizigiye, F. (1997). Agroforestry, water and soil fertility management to fight erosion in tropical mountains of Rwanda. *Soil Technology*, *11*(1), 109–119.
- ROSSI, G. (1998). Une relecture de l'érosion en milieu tropical. *Annales de Géographie*, *107*(601), 318–329.
- Rushemuka, P. N., Bock, L., & Mowo, J. G. (2014). Science du sol et développement agricole au Rwanda: état de la question (synthèse bibliographique). *Biotechnologie, Agronomie, Société et Environnement, 18*(1), 142–154.
- RWANDA, G. O., DU RWANDA, G., & others. (1991). Stratégie nationale de conservation des sols: Evaluation des système d'exploitation agricole pour une régionalisation des

techniques de conservation et d'amélioration de fertilité des sols au Rwanda. Retrieved from https://repositories.lib.utexas.edu/bitstream/handle/2152/5212/2515.pdf?sequence=1

- Salam, M. A., Noguchi, T., & Koike, M. (2000). Understanding why farmers plant trees in the homestead agroforestry in Bangladesh. *Agroforestry Systems*, *50*(1), 77–93.
- Schroth, G., Krauss, U., Gasparotto, L., Aguilar, J. A. D., & Vohland, K. (2000). Pests and diseases in agroforestry systems of the humid tropics. *Agroforestry Systems*, 50(3), 199–241.
- Shongwe, M. E., van Oldenborgh, G. J., van den Hurk, B., & van Aalst, M. (2011). Projected changes in mean and extreme precipitation in Africa under global warming. Part II: East Africa. *Journal of Climate*, 24(14), 3718–3733.
- Stainback, G. A., Masozera, M., Mukuralinda, A., & Dwivedi, P. (2011). Smallholder Agroforestry in Rwanda: A SWOT-AHP Analysis. *Small-Scale Forestry*, 11(3), 285– 300.
- Stainback, G. A., Masozera, M., Mukuralinda, A., & Dwivedi, P. (2012). Smallholder Agroforestry in Rwanda: A SWOT-AHP Analysis. *Small-Scale Forestry*, 11(3), 285– 300.
- Turnbull, J. W., & others. (2000). *Economic and social importance of eucalypts*. CSIRO Publishing, Australia. Retrieved from https://books.google.com/books?hl=en&lr=&id=PENpGhQ1qmgC&oi=fnd&pg=PA1&d q=+Economic+and+social+importance+of+eucalypts.+In:+Keane+&ots=0hrPcGCEqW &sig=l0dxhcaqL3OJIsYW15WAF_s9NaE
- Udawatta, R. P., Garrett, H. E., & Kallenbach, R. L. (2010). Agroforestry and grass buffer effects on water quality in grazed pastures. *Agroforestry Systems*, *79*(1), 81–87.

- Vaast, P., & Somarriba, E. (2014). Trade-offs between crop intensification and ecosystem services: the role of agroforestry in cocoa cultivation. *Agroforestry Systems*, 88(6), 947– 956.
- Varah, A., Jones, H., Smith, J., & Potts, S. G. (2013). Enhanced biodiversity and pollination in UK agroforestry systems. *Journal of the Science of Food and Agriculture*, 93(9), 2073– 2075.
- Verchot, L. V., Noordwijk, M. V., Kandji, S., Tomich, T., Ong, C., Albrecht, A., ... Palm, C. (2007). Climate change: linking adaptation and mitigation through agroforestry. *Mitigation and Adaptation Strategies for Global Change*, *12*(5), 901–918.
- Verdone, M., & Seidl, A. (2016). How do the risk preferences of smallholder farmers affect the attractiveness of restoration activities? Evidence from forest and agricultural land in Rwanda. *Journal of Development and Agricultural Economics*, 8(9), 200–214.
- Wali, U. G., Nhapi, I., Ngombwa, A., Banadda, N., Nsengimana, H., Kimwaga, R. J., &
 Nansubuga, I. (2011). Modelling of Nonpoint Source Pollution in Akagera
 Transboundary River in Rwanda. *Open Environmental Engineering Journal*, 4, 124–132.
- Yamoah, C. F., & Grosz, R. (1988). Linking on-station research with on-farm testing: The case of agroforestry and organic matter-based cropping systems for the Rwanda farming systems improvement project. *Agroforestry Systems*, 6(1–3), 271–281.
- Zimmerman, L. S., Byizigiro, & Vaillant. (2012). Rapid risk and capacities assessment and livelihoods prolifering in Nyabihu, Musanze and Burera Districts affected by floods and landslides. Kigali: MIDIMAR.

Chapter 3: Smallholder farmers perceptions of climate change and adaptation strategies in the northern highlands of Rwanda

Abstract

Smallholder farmers in poor countries generally rely heavily on agricultural production for their livelihoods, generally have the least capacity to adapt to climate change, and are expected to suffer the most from climate change and climate variability. Timing of crop planting, crop choice and agroforestry practices are adaptation strategies to reduce the impacts of climate change on smallholder farmers. Understanding local farmers' perceptions of climate change, willingness to adopt adaptation strategies and challenges to adaptation are fundamental to addressing issues of soil conservation and poverty reduction under environmental change. To develop appropriate policies and strategies for climate change adaptation in the agricultural sector, it will be crucial not only to understand changes in climate, but also to understand how they are perceived and adapted to by local residents, and how any adaptation strategies proposed are perceived by local farmers. The northern highlands region of Rwanda is one of the most climate-sensitive areas in the country due to dense smallholder agriculture landuse with limited agroforestry. This study explored farmers' perceptions of climate change and current adaptation strategies, and challenges and opportunities pertaining to the adoption of agroforestry as one of the adaptation option. Results of a survey conducted with 430 farmers within the northern highlands (Musanze District) of Rwanda between August and December 2015 indicate that most farmers perceived a shift in the onset of short rains and they have adapted planting timing accordingly. Analysis of rainfall data shows a recent increase in average monthly rainfall for the short rainy season as well. The results indicate that farmers perceived risks imposed by climate change on certain crops,

particularly maize and potatoes. However, farmers did not respond accordingly; their goal of increasing short-term financial profitability overwrote their willingness to adopt agricultural practices that sustain soil conservation. Regarding farmers' willingness to adopt agroforestry as a strategy for climate adaptation within their farmlands, a significant positive relationship was found between recognition of the importance of planting trees in association with crops, and age, level of education and income. In promoting awareness among famers about climate change adaptation strategies including agroforestry, results highlight the need to have farmers realize a greater longer term economic benefit from adopting the strategies over short term economic gain, to explore the potentials of farmer-to-farmer extension, using participatory approaches to generate solutions that are based on their experiences of climate change, and to improve the effectiveness of government-led extension to farmers.

Key words: Agroforestry, extension services, financial incentives, food security, short-term profitability, soil conservation, soil degradation

Introduction

Many scientists agree that developing countries, where agriculture is the main source of livelihood and where there is the least capacity to adapt and reduce risk, are expected to suffer the most from climate change and climate variability (IPCC, 2014; McBean & Ajibade, 2009; de-Graft Acquah & Onumah, 2011; Gbetibouo, 2009). Because agricultural production remains the main source of income for most rural communities, farmers' adaptation to climate change is critical (Adger et al., 2007). The term adaptation refers to "adjustment in natural or human systems in response to actual or expected climate stimuli or their effects, which moderates harm or exploits beneficial opportunities' (IPCC 2001). Adaptation at the farm-level can involve two stages: perceiving the change in climate, and deciding whether to adapt or not; or which adaptation strategy to choose (Maddison, 2007). Farmers are believed to perceive recent climate changes based on their personal experience (Tessema et al., 2013) and some local farmers often have a very clear memory of years dominated by extreme weather events (e.g. Mertz et al., 2009; Manandhar et al., 2011; Tambo & Abdoulaye, 2013; Yaro, 2013). Such perceptions of strong recent climate stimuli can trigger short-term responses by farmers (Gbetibouo, 2009; Smith et al., 1996). However, it may not always be an easy exercise to put together an accurate mental picture of past climate trends since farmers' memories are usually focused on extreme weather events rather than general trends (Tambo & Abdoulaye, 2013). Thus, adaptation may be more likely based on responses to extreme events rather than slow, long term shifts in climate.

Farmers' responses to climate stimuli hinge not only on recently-experienced changes and impacts of climate but also on a complex interplay between social, economic and institutional factors (Bryan et al., 2009a; Maharjan et al., 2011; Maddison, 2007), including access to

information (Hassan, Nhemachena, & others, 2008; Temesgen Tadesse Deressa, Hassan, Ringler, Alemu, & Yesuf, 2008) which can be acquired through formal education, the media and extension services. Access to agricultural extension services has been shown to have a significant impact on adaptation to climate change (Fosu-Mensah et al., 2012; Schuck, Nganje, & Yantio, 2002; Hassan et al., 2008; Maddison, 2007; Franz et al., 2010). Farmer-to-farmer extension services, coupled with government-led extension services, has been effective in adaptation given its contagious nature and trust among neighboring farmers (Kiptot et al., 2006; Davis et al., 2012). Inaccessibility of extension services by poor marginalized farmers including women, minorities, and people in very remote areas can even undermine effective adaptation to climate change (Davis et al., 2012; Anandajayasekeram et al., 2007).

Besides access to information and training, it has been argued that farmer financial capacity contributes substantially to climate change adaptation. Farmers with higher household income have been shown to be more likely to adopt new agricultural practices, given greater access to financial resources such as credits and subsidies (Franzel, 1999; Knowler and Bradshaw, 2007; Croppenstedt et al., 2003; Nhemachena and Hassan, 2007; Gandure et al., 2013; Deressa et al., 2011). However, in some countries such as Rwanda, where micro credit schemes are widespread and are believed to contribute to the efficiency of agricultural production, micro credit is virtually inaccessible to smallholder poor and uneducated farmers, and could threaten farmers' capacity to adapt their agricultural practices to climate change (Nabahungu & Visser, 2013; Ali et al., 2014; Musanze, 2013; Mutandwa & Kanyarukiga, 2016). Access to information and financial capacity may be an important enabling and empowering means to ensure economic

profitability and sustainability of agricultural practices under climate change, especially when available to poor farmers with minimal formal education.

Another strategy proposed as an effective means to adapt to climate change involves agroforestry practices, which may not only contribute to soil fertility but also provide other benefits such as soil erosion control, stream bank stability, control of sedimentation and nutrient loading into streams (Atangana et al., 2014). The adoption of agroforestry among smallholder farmers is believed to depend largely on farmers' understanding of the importance of the practice, through extension services and farmer-to-farmer extension (Kiptot et al., 2006). In addition, adoption of agroforestry, like any other agricultural practice, hinges on a clear understanding of its economic profitability relative to the costs of implementation (McGinty et al., 2008).

Economic profitability is a major driver of climate adaptation among farmers particularly with respect to the choice of farming systems and crop choice (Chouinard et al., 2008; Musshoff & Hirschauer, 2008; Cary & Wilkinson, 1997). Economic profitability includes maximization of gross margin, and minimization of working capital, hired labor, management difficulties, and risk (Sheeder & Lynne, 2011). From a socio-psychological perspective, following the Theory of Planned Behavior (TPB), access to information and financial incentive are imbedded in what Läpple & Kelley (2013) is termed *belief-based measures* which influence the intentions to perform a behavior, innovative, adaptive agricultural practices in this case. Scholars of the Theory of Planned Behavior (TPB) such as Ajzen (1991, 1985, 1991) and Gellman & Turner (2013) suggest that perceptions drive willingness or reluctance to adopt any given behavior; that is, any given attitude or behavior will be influenced by the perception of expected benefit of the

behavior or past experiences that justify a need to adopt a new behavior. TPB includes the possibility of heterogeneous decision structures among farmers who have different levels of environmental concerns, a component that falls under social norms (Läpple & Kelley, 2013).

While research on farmers' perceptions of climate change and adaptation abound in Africa, focus has been virtually restricted to dry regions and adaptation has focused on farmers' adaptation practices and challenges pertaining to adaptation (Bryan et al., 2013). Little has been done in humid regions such as the northern highlands of Rwanda, Uganda, and Congo (e,g. Diem et al., 2016; Osbahr et al., 2011). In addition, research on climate change perception has not looked at farmers' attitudes towards innovative agricultural practices for adaptation, factors influencing willingness to adopt certain behaviors, nor the opportunities and challenges that could be considered to facilitate the adoption of the practice.

In this study, I used survey data collected from smallholder farmers in the northern highlands of Rwanda to understand farmers' perceptions of climate change and decisions about adopting adaptation practices. I also analyzed rainfall data from the Kigali airport climate station to analyze seasonal variability in order to validate local perceptions of changes in seasonal rainfall variability. I hypothesized that (1) farmers' perceptions of seasonal variability are associated with residence time, household income, and level of education; (2) planting season is associated with the perception of variability of the onset of rainfall; (3) farmers choice of crops to plant are primarily driven by short-term profitability rather than sustainability of their farming activities; and (4) farmers' willingness to adopt agroforestry is associated with their residence time and education level rather than their level of income.

Methods

The research used qualitative and quantitative methods to gather primary data in 2015 from 430 smallholder farmers within the Upper Mukungwa Watershed. The goal of the research was to understand farmers' responses to climate stimuli especially with regard to the onset of rains and their attitudes about the adoption of agroforestry as a climate adaptive strategy. Understanding the relationship between farmers' perceptions of climate change and variability, and their decisions about planting timing and crop choices can help policy and decision makers design solutions that are tailored to smallholder farmers' needs for effective climate adaptation.

Study area.

The upper Mukungwa watershed (Figure 3-1) covers about 60% of the Musanze District land surface area, which is located in the northern highlands of Rwanda. The watershed lies in a region characterized by a bimodal rainfall pattern where rainfall seasons are significantly influenced by the Intertropical Conversion Zone which oscillates annually between the northern and southern tropics (McSweeney et al., 2010). The major rainy season is between February and June with a peak in April, and the minor rainy season is between September and December with a peak in November. The region experiences severe soil erosion and flooding due to steep slopes and episodes of high rainfall (Nahayo, Ekise, & Niyigena, 2013; Zimmerman et al., 2012; REMA, 2011; Musanze District, 2013; SEI, 2009).

Musanze has a landscape divided in two main areas: volcanic plains and the mountain range. The volcanic plains cover the central and north part of the district, including the Musanze, Muhoza, Muko, Kimonyi and Cyuve sectors; average altitude is 1,860 m. The mountain range is located in the southeast of the district, covering over a third of the total surface of the district,

with an altitude range from 1,900 m to 2,000 m, covering the Muhoza, Cyuve, Gacaca, Rwaza, Gashaki, Remera and Nkotsi sectors (Rugazura & Murugesan, 2015).

The Musanze District is 76% agricultural lands. The study site lies on volcanic soil in an area characterized by a moderate and humid climate due to the high elevation and abundant rainfall, with an annual temperature and rainfall average of 16° C and 1400mm, respectively (Musanze District, 2014). Over 90% of the population relies on subsistence agriculture for its economy. Potato is the major food and cash crop in the highland regions of Rwanda, followed by maize and beans. Wheat, peas, banana, and other vegetables and fruits are also cultivated (Muhinyuza et al., 2012). According to the National Institute of Statistics (NISR, 2012), females account for 54.1% of the Musanze population. The average household size is 4.8. Nearly 80% of the population is identified as non-poor, 14% as poor and 6% as extremely poor. The Ministry of Local Government uses the following definitions for each income category: extremely poor do not have a house or cannot pay rent, have a poor diet, and cannot get basic household tools and clothing; the poor include those who have their own houses or can afford to rent a house, can generally obtain food and can earn a wage from working; the non-poor include those who have at least one person in the family working in the government or the private sector; high income people are considered those who earn high incomes, own houses and can afford a luxurious lifestyle (MINALOC, 2015).

Agriculture is the main industry for 67% of the population aged 16 and above. The mean size of land cultivated per household is 0.5 ha; 87% of households cultivate under 0.9 ha of land (Musanze District, 2014), which is below the minimum household farm size of 0.9 ha required to
conduct sustainable agriculture. The majority of females in Musanze district are small scale-farm workers (73%); small-scale farm workers among males account for 37%. Only 46% of agricultural households have access to chemical fertilizer. Figures about the rate of illiteracy were not available but 85% of individuals aged six and above have attended primary school.



Figure 3-1. Upper Mukungwa Watershed in Rwanda with locations (superpixels) where household interviews were conducted

Household Climate Perceptions and Risks.

Household surveys were used to examine local farmers' perceptions of environmental change within the Upper Mukungwa watershed. The watershed (244 km²) is predominantly agricultural besides a few grouped settlements known as "imidugudu" and the Musanze town. A set of 77 random geographic coordinates within this area was selected, and those points became the centers of 9-hectare areas (circles with radii of 170 m) termed "superpixels" (Hartter, 2009) (Fig. 1). Interview respondents were selected from within each superpixel. The number of respondents selected per superpixel was based on availability during the period of interviews for a given superpixel and at least one interview was conducted in each superpixel. Houses were selected for the first interview, the next closest for the second interview, and so on. A full description of the geographic selection methodology can be found in Hartter, (2009). The survey was conducted between August and December 2015.

Interviews were conducted in person by myself with two field assistants, using Kinyarwanda, the local language. I spent one week training the field assistants before allowing them to collect data on their own, and after I trained them, each field assistant collected data interviewing a farmer and we would meet at the each of the interview to comment on how the interview went and discuss how the interviews could be improved. In the next step, field assistants started to collect data and I would check quality of the data every day for a couple weeks. After two weeks of data collection, field assistants continued to collect data and send in records every end of the week. I would then check data for errors, typos and missing data. Interview questions were structured and responses were then coded into categories during data analysis. Relationships between

categorical responses and independent variables were examined with chi-square tests for independence (gender, location, wealth, newcomer status), while continuous variables were examined using binomial logistic regression (residence time, distance to park, stream and road, and respondent age).

Rainfall data analysis.

Long-term 24-hour rainfall data have been collected from the Kigali Airport Meteorological station since 1971. In order to compare recent and past trends in terms of onset of rains particularly for the short rainy season, I compared rainfall data between two time periods, the period of 1971-1998 and the period of 1999-2013. The later period was determined based on questions I asked to farmers about recent changes they have seen with respect to the onset of rains compared to more than ten years ago. With this information I found that this 15 year span of time can reasonably reflect the recent rainfall changes experienced by farmers. I focused only on the months of August and September for this analysis because September has been the traditional date that farmers use to start planting. I assigned an index to each month (August and September) depending on the number of rainy days from the first day of rain in that month to the last (thus for August there are 31 possible days for rain and a possible index of 31, and September has 30 possible days for rain); a day without rain was scored '0' and a day with rain was scored '1' and these scores were summed in each month to obtain the Index. This was done for both months across each year of the two periods, yielding an index of rain based on number of rain days for August and September in each year of the time period. I then averaged the indices assigned to days of rain for each month across all years of each of both periods.

A t-test was performed to compare means of indices between the two periods of 1971-1998 and 1999-2013 for the months of August and September respectively. Results were also plotted in a graph to show changes in rain index peaks between the two periods.

Results

General farmer characteristics and agricultural systems within the study site.

A total of 70 superpixels were sampled and 430 farmers were interviewed within those superpixels. More females than males were interviewed. A large proportion of interviewed farmers were illiterate, owned small farmlands and were poor (Table 3-1).

Table 3-1

Population Statistics of Respondents in the Upper Mukungwa Watershed, Rwanda

Population characteristics	Total
Sample size	430
Average Household size	4.6
Male (%)	43
Female (%)	57
Illiterate	50
Completed primary education (%)	39
Completed secondary education (%)	7
Higher education (%)	4
Mean experience	29
Average age	44
Farm size between 0 and 0.5 hectares (%)	51

Farm size between 0.5 and 1 hectare (%)	40
Farm size between 1 and 5 hectares (%)	9
Very low income (%)	27
Low income (%)	39
Medium income (%)	33
High income (%)	1

Farmers' perceptions of climate.

Farmers were asked to describe what they could say about current climate. The most common response from the respondents was that they were experiencing a change in weather and in onset and duration of rains in rainy seasons, and the second most frequent response was that they did not know anything to say about the climate (Figure 3-2).



Figure 3-2. Perceptions of changes in current climate/weather/seasons change by farmers in the Upper Mukungwa watershed, Rwanda

Timing of onset and cessation of rainy seasons.

To further investigate respondents' perceptions of climate change/climate variability, I asked questions about the timing of onset and cessation of rainy seasons. Answers to these questions helped understand whether farmers' perceptions about perceived changes in the onset and cessation of seasons drive their crop planting timing.

When respondents were asked if they felt that the short rainy season (typically from September-December) has been starting: earlier, about the same time, later, or different each year in recent years compared to 10 years ago or more, the majority of participants reported that the onset of rains comes earlier and another large number of respondents reported that it is variable from year to year (Figure 3-3). Only a small number or respondents said the onset has not changed (2%) or comes later (8%). Respondents were also asked about the end of the long rainy season and most of respondents reported that the season ends earlier.



Figure 3-3. Farmers' perceptions of the timing of rainy seasons in northern Rwanda, including perceptions of the cessation of the long rainy season and the onset of the short rainy season.

Logistic regression test results (Table 3-2) suggest that farmers with relatively higher income are more likely to feel that the short rainy season (September-November) begins earlier or is now more variable from year to year. Likewise, they are more likely to feel that the cessation of the long rainy season comes late or varies from year to year. These perceptions were not associated with gender or age. However, there was a significant association of those perceptions and the length of residency of farmers in the area. Finally, there was a positive relationship between farmers' perceptions that the onset of the short rainy season comes earlier or later and the status of being literate. Likewise, perceptions that the cessation of the long rainy season comes later or varies from year to year are also associated with the state of being literate.

Table 3-2

Perceptions of Local Households about Changes in Onset and Cessation of Rainy Season (Bolded Numbers Indicate Significance of Dependency)

		Onset rainy	of short season	Cessation of long rainy season								
	N	Earlier	Variable	Early	Late	Same	Variable					
Total	430	51%	37%									
Gender												
Male (code = 1)	184	46%	40%	11%	36%	12%	35%					
Female (code = 0)	246	55%	35%	15%	41%	13%	34%					
p-value		0.064	0.269	0.250	0.424	0.883	0.759					
Wealth												

Very poor	115	67%	19%	11%	61%	13%	16%
poor	166	60%	25%	14%	46%	14%	23%
Medium income	144	29%	65%	13%	13%	11%	62%
Wealthy	5	20%	60%	20%	40%	0%	40%
P-value		< 0.001	< 0.001	0.753	< 0.001	0.635	< 0.001
Education							
Illiterate	216	57%	25%	13%	47%	14%	43%
literate	214	44%	49%	14%	30%	11%	46%
p-value		0.007	< 0.001	0.803	< 0.001	0.322	< 0.001
Age							
p-value		0.076	0.240	0.452	0.387	0.405	0.758
Residency							
p-value		< 0.001	< 0.001	0.496	0.10	0.001	< 0.001

Timing of crop planting.

Farmers were asked if in recent times compared to 10 years ago, they plant their crops earlier, later, same time each year, or if the timing for crop planting varies from year to year. The majority of respondents reported that they plant crops earlier than 10 years ago. There was also a substantial number of respondents who reported that the timing for crop planting varies from year to year (Figure 3-4).



Figure 3-4. Farmers' responses about their perceptions of the timing of crop planting

Results from the survey data also show that farmers who perceive that the onset of the short rainy season comes earlier are significantly more likely to plant their crops later (Table 3-3). Those respondents who believe that the onset of short rains is variable from year to year are significantly more likely to plant their crops at variable times (Table 3-3).

Table 3-3

	Onset of planting											
	Earlier	Later	Same	Varies each year								
Onset of short rain season												
Earlier	18%	56%	15%	8%								
p-value	0.003	< 0.001	0.040	< 0.001								

Relationship between Perceptions of the Timing of the Onset of the Short Rainy Season and Timing for Crop Planting

Variable	5%	11%	4%	79%
p-value	< 0.001	< 0.001	0.001	< 0.001

Rainfall seasonal variability

I analyzed average number of rainy days between the periods 1971-1998 and the period 1999-2013 to see if there was a significant difference in the variability of number of rainy days and if the short rains comes earlier in recent years compared to more than fifteen years ago. First I assigned number 1 to each rainy day, then I averaged the numbers for each day across each of the two periods (1971-1998 and 1999-2013 for the months of August and September respectively.

Results show graphically higher peaks over average rainy days in August (Figure 3-5) and relatively lower peaks in September in recent years compared to more than 15 years ago (Figure 6). However, a statistically significant difference is for the month of September ((t (56) = 2.003, p = 0.018).



Figure 3-5. Average number of days of rain for the period 1971-1998 and 1999-2013 for the month of August



Figure 3-6. Average number of days of rain for the period 1971-1998 and 1999-2013 for the month of September

I analyzed 24-hour rainfall data from 1971-2013 using a t-test to determine if there was a significant difference in the variability of average daily rainfall for the months of August right

after the dry season ends and September when the short rains typically start. I compared recent years (1999-2013) with previous years (1971-1998) by looking at cumulative daily rainfall starting from the first day of August (considered as day 1) to the last day of September (considered day 61) (Figure 3-7). The crop planting in the study area normally occurs during this period. The results were compared to what respondents reported about the onset of short rains. Results from a t-test revealed that there was no significant difference in the cumulative rainfall from beginning of August to end of September, between the periods 1999-2013 and 1971-1998 (t (119) = 1.98, p = .393).



Figure 3-7. Cumulative daily rainfall averages for Kigali Airport meteorological station, Rwanda

I further examined seasonal distribution of monthly average rainfall for the periods 1999-2013 and 1971-1998. For both periods, April and November correspond to peak rainfall months. A ttest shows a significant difference in monthly average rainfall between the two rainfall period in April (t (29) = 2.011, p = .05), but there was no significant difference for the month of November between both periods (t (29) = -0.183, p = 0.42). This suggests a significant decline in the rainfall peak of April during the long rain season in the period 1999-2013 compared to 1971-1998 period (Figure 8). There was no significant difference between October average rainfall between both periods (t (30) = 1.69, p = 0.11). The month of September does not show any statistical difference either (t (29) = 0.81, p = 0.21). There was no significant difference in rainfall variability between the two periods (t(22) = 2.074, p = .79), indicating there is not sufficient evidence to warrant the claim that rainfall for the period 1971-1998 was higher on average than rainfall of the period 1999-2013.



Figure 3-8. Monthly average rainfall at Kigali Airport meteorological station, Rwanda. Numbers 1 to 12 correspond to the months of a year.

Perception of climate risks.

Farmers were asked to rate current climate-related issues based on the level of adverse impacts of their households. Farmers in the study area reported climate-related issues that they experience and they ranked them based on the severity in terms of the extent to which their household is affected. Flooding, sediment accumulation, stream bank erosion, soil erosion and water pollution were identified and ranked (Figure 3-9). Soil erosion was reported to be the most serious risk, followed by sediment accumulation and stream bank erosion.



Figure 3-9. Rating of climate risks considered to be of significant concern among farmers in the study area

A multinomial logistic regression was used to ascertain a relationship between climate change related issues (independent variables) that farmers believe affect their households and respondents characteristics (dependent variables) including gender, education, farm size, income, residency length and proximity of respondents' household to streams and the Volcanoes National Park (VNP), and distance of farmland to streams (Table 3-4).

Table 3-4

Relationship Between Perceived Climate-related Issues Believed by Farmers to be of Concern and Socio-economic Factors (Significant relationships are highlighted in bold).

	Total		Flooding			Sediments			Stream. Bank erosion			Soil erosion			Water pollution		
		0	1	2	0	1	2	0	1	2	0	1	2	0	1	2	
Gender																	
Total	430	23	127	63	24	148	70	22	140	72	23	188	66	54	117	60	
Female	246	56.5%	49.6%	42,9%	54,2%	49.3 %	58.6%	54.5%	52.1%	56.9%	56.5%	52.7%	52.6%	57.4%	53%	60%	
Male	184	43.5%	50.4%	57.1%	45.8%	50.7 %	41.4%	45.5%	47.9%	43.1%	43.5%	47.3%	47.4%	42.6%	47%	40%	
p-value		.372	.003	.001	.387	.010	.487	.544	.097	.526	.567	.062	.472	.945	.315	.868	
Education																	
Total	430	23	127	63	24	148	70	22	140	72	23	188	66	54	117	60	
Illiterate	216	43,5%	41.7%	44.4%	41.7%	37.8 %	51.4%	45.5%	34.3%	55.6%	47.8%	40,4%	56.1%	51.9%	37.6 %	46.7 %	
Literate	214	56.5%	58.3%	55.6%	58.3%	62.2 %	48.6%	54.5%	65.7%	44.4%	52.2%	59.6%	43.9%	48.1%	62.4 %	53.3 %	
p-value		.199	.005	.066	.081	.000	.183	.187	.000	.493	.267	.000	.574	.397	.000	.114	
Wealth																	
Total		23	127	63	24	148	70	22	140	72	23	188	66	54	117	60	

Very poor	115	30.4%	18.9	20.6	29.2%	18.9 %	27.1%	27.3%	14.3%	47.2%	30.4%	18.1%	34.8%	31.5%	17.1 %	21.7 %
poor	166	43.5%	30.7%	38.1%	45.8%	20.9 %	40.1%	45.5%	23.6%	34.7%	47.8%	27.7%	40.9%	46.3%	26.5 %	25.0 %
Medium income	149	26.1%	50.4%	41.3%	25%	60.1 %	32.9%	27.3%	62.1%	18.1%	21.7%	54.3%	24.2%	22.2%	56.4 %	53.3 %
p-value	Incom e 1	.813	.000	.011	.332	.000	.023	.688	.000	.063	.594	.000	.443	.704	.000	.000
	Incom e 2	.925	.000	.053	.339	.000	.007	.568	.000	.661	.627	.000	.158	.695	.000	.000
Residency																
p-value		.074	.224	.156	.294	.000	.960	.477	.005	.391	.958	.001	.713	.198	.201	.513
Age																
p-value		.083	.024	.985	.068	.061	.0.46	.083	.288	.023	.460	.144	.619	.090	.186	.018
Farm size																
p-value		.037	.000	.143	.001	.000	.015	.006	.000	.672	.009	.000	.779	.043	.007	.006
Proximity to s	tream															
p-value		.012	.021	.517	.023	.000	.021	.018	.001	.022	.040	.000	.204	.421	.002	.006
Proximity to V	/NP															
p-value		.007	.555	.001	.024	.164	.019	.161	.215	.123	.037	.699	.145	.000	.179	.955
Closeness of fa	armland to	o streams														
p-value		.395	.000	.072	.058	.000	.062	.069	.000	.060	.060	.000	.078	.548	.000	.552

Results show a significant association between farmers' beliefs that flooding, sediment accumulation, stream bank erosion, soil erosion and water pollution represent serious risks for households and variables including education, income, farm size, closeness of a household from a nearby stream, closeness of any of the respondent's farm plot to a stream (Table 3-4). Results in the table suggest that relatively high-income level and educated farmers are more likely to believe that flooding, sedimentation, stream bank erosion, and soil erosion are challenges to agricultural production than poor, illiterate farmers. However, farmers whose farms are in proximity to streams and the Volcanoes National Park are more likely to believe that those environmental issues are of concern than farmers who do not own farms in this proximity.

Farmers' perceptions of recent increase in the cultivation of certain crops and rise in concerns about climate change risks.

First, I asked farmers about which crop types have increased or decreased in cultivation over recent years compared to ten years ago. A great majority of farmers (82% and 75% of respondents respectively) strongly agreed that corn and bean cultivation has increased substantially. Besides corn and beans, Irish potatoes was reported to have increased in cultivation over recent years (49% of respondents). Secondly, farmers were asked about which type of crops are the most susceptible to the effect of soil erosion. This is a very relevant question because the Musanze District administration recognizes soil erosion as one of major challenges to agricultural production (Musanze, 2013). Results show that the majority of respondents believe that beans, corn and Irish potatoes are the most vulnerable to soil erosion (Figure 3-10).



Figure 3-10. Crop susceptibility to damage due to soil erosion, based on farmers' perception

To understand what motivates farmers to plant crops most vulnerable to soil erosion, I asked farmers to indicate their level of agreement (strongly agree, somewhat agree, disagree) on factors that may underpin their choice. Results reveal a vast majority of farmers reported that their choice is dictated either by crop rotation cycles (the growing of different crops in succession on a piece of land), economic profitability, family nutritional needs or recommendations from the government (Figure 3-11).



Figure 3-11. Rating of factors that drive farmers' choice of crops to plant

Attitudes towards the adoption of agroforestry.

Recent literature suggests agroforestry can serve as an effective strategy to limit soil erosion from happening and to contribute to soil fertility recovery while providing other ecosystem services that support farming and farmers' livelihoods (Atangana et al., 2014a; Mbow et al., 2014). I gauged farmers' interest and willingness to adopt agroforestry after I presented to them the benefits of interplanting crops with trees. Results show that a majority of respondents (64% of farmers) expressed willingness to adopt agroforestry practices. A logistic regression model was used to assess the relationship between willingness to adopt agroforestry and sociodemographic factors as well as the perception of soil erosion as a major problem. Positive attitudes towards agroforestry were significantly negatively associated with the length of respondents' residence ($\chi 2 = 4.115$, df = 1, p = 0.043); that is, recently established farmers compared to farmers who have been farming the land for a long time are more likely to adopt agroforestry. Similarly, there was also a positive relationship with literacy category; literate farmers had significantly positive attitudes about the adoption of agroforestry in comparison to illiterate farmers ($\chi 2 = 7.946$, df = 1, p = 0.005). The positive attitude towards agroforestry was also positively associated with the income level of respondents ($\chi 2 = 8.086$, df = 1, p < 0.004). However, no significant association was found with soil erosion as a problem ($\chi 2 = .207$, df = 1, p = .649).

Perceived Challenges to the adoption of agroforestry.

Six challenges to the adoption of agroforestry were identified by the majority of respondents and can be summarized into two major categories: financial and technological challenges (Figure 3-12). Virtually all respondents indicated that the fact that agroforestry does not generate immediate benefits could be an impediment to its adoption.



Figure 3-12. Rating of major challenges inherent to potential adoption of agroforestry by farmers in the Upper Mukungwa Watershed, Rwanda

Discussion

Awareness of change of onset of short rainy seasons.

The vast majority of farmers in this study believe that the onset of the short rainy season comes earlier in recent years compared to more than ten years ago. Analysis of average rainy days between the periods 1971-1998 and 1999-2013 shows a significant shift in the average of rainy days for the month of September with relatively lower peaks and higher peaks in August compared to the period 1971-1998. A recent study of onset, cessation and length of growing season in western Kenya by Mugalavai et al. (2008) showed no shift in seasons for the past 15-34 years. Other studies on East Africa, including Rwanda, have predicted an increase of precipitations during the short rain season. Mean precipitation in Rwanda is projected to increase by more than 10% by year 2100 (Shongwe et al., 2008; Black, 2005). During the season from mid-September to December, humidity in Rwanda comes from air masses humidified by the Indian Ocean and Lake Victoria (NAPA-RWANDA, 2006). This increase could be explained by the Indian Ocean Dipole (IOD) (Behera et al., 2005) but it does not seem to affect average monthly rainfall in Rwanda. Studies on climate change in Rwanda are needed to elucidate the effect of an increase in humid air masses from the Indian Ocean on rainfall intensity, distribution and onset of seasons in order to support climate adaptation strategies.

Timing of planting and onset of short rains.

Regarding the question of whether the planting season is associated with the perception of variability of the onset of rainfall, most of farmers who claimed that the onset of short rains comes earlier reported that they tend to plant crops relatively later than they did 10 years ago. However, most farmers, including those who did not report that short rains come earlier, claimed

to plant crops earlier. This may suggest that recent increase in short rains provides the soil with enough moisture and farmers can grow their crops earlier, However, given rainfall variability when considering each year individually, it is possible that some farmers wait until other farmers have planted their crops so that they can plant their own, or there might be other factors such as lack of input such as fertilizers, pesticides and labor.

Crop plant choice and climate adaptation.

The main goal of farmers and what generally drives their practice, according to the findings of this study, is to increase short-term financial profitability of their agricultural practices. Financial profitability has been suggested to be a major driver of climate adaptation among farmers particularly with respect to the choice of farming systems and crop choice (Chouinard et al., 2008; Musshoff & Hirschauer, 2008; Cary & Wilkinson, 1997). Economic profitability includes maximization of net profit with minimum risks (Sheeder & Lynne, 2011). Results of this study corroborate previous studies that farmers tend to subscribe to short term financial benefits and are reluctant to buy into long-term agricultural schemes that represent environmental benefits and effective soil conservation (Dale & Polasky, 2007; Karali et al., 2013). Interestingly, these short term financial benefits can have additional long term costs: as the results suggest, increased use of synthetic fertilizers is associated with increased cultivation of Irish potatoes and corn, which have become the main goal for farmers as part of their adaptation strategy. Soil loss following soil erosion is generally more pronounced in corn and potato crops planted in pure stands than in other crops commonly found cultivated in Rwanda (e.g. Clay and Lewis, 1990).

Reconciliation of climate-sensitive crops and agroforestry.

Most of farmers in this study reported that they are willing to adopt agroforestry. However, they also reported that successful adoption depends on financial and technological capacity. Along the same lines, it appears that the adoption of agroforestry, which is a long-term commitment, requires not only the understanding of its long-term benefits but also and most importantly the "know how" or technological capacity required for effective implementation of agroforestry that generates longer economic benefits that outweigh some short term reduction of current benefits (Nhemachena & Hassan, 2007). The results from this study also suggest that farmers with relatively higher income are more likely to adopt agroforestry. This seems to make sense in a country like Rwanda where virtually all farmers rely on subsistence agriculture with an average farm holding of 0.76 ha (Booth & Golooba-Mutebi, 2012) and an area of about 0.45 ha in the Musanze District, generally split up into fragments (Figure 10). The mean size of household in Musanze is 4.8 (National Institute of statistics of Rwanda, 2012). In this situation, smallholder farmers attempt to maximize the short term output from the plots. They plant potatoes which grow fast in all seasons and provide income for families. Beans also provide very good income and are the main source of protein for most smallholder farmers. Recent agricultural reforms in Rwanda (Pritchard, 2013) have yielded satisfactory results economically but may contribute to soil degradation and pollution through nutrients loads into streams (Wali et al., 2011; Huggins, 2009). From the farmers' perspective, food crop production is a priority. Farmers understand that agroforestry can contribute to protecting crops against the impact of climate change while providing other benefits they also need either for food production or their livelihoods such as fodder for animals and spikes for climbing beans. For this reason, a reconciliation of food crop

cultivation and agroforestry is possible. This reconciliation would also help farmers cultivate on unfavorable hill slopes and hill tops as shown for example in Figure 3-13.



Figure 3-13. Agricultural management in the Musanze district, northern Rwanda

Farmers' willingness to adopt agroforestry was associated with the level of education, literate farmers being more willing to adopt the practice than illiterate ones. Studies from Ethiopia and South Africa (Bryan et al., 2009b) and in many other sub-Saharan countries (Hassan et al., 2008) show that farmers are more likely to adapt to climate change if they have access to extension, credit, and land. Extension services and information on climate change were found to facilitate adaptation among the poorest farmers (Hassan et al., 2008; Bryan et al., 2009b).

In Rwanda however, the problem does not seem to be related to the presence of extension services but rather to the inadequacy of those services as noted by Stainback, Masozera,

Mukuralinda, & Dwivedi (2011) and Stainback et al. (2011). This gap might also be supported by the finding that about 63% respondents in this study claimed that lack of training and expertise is one of the major challenges pertaining to the adoption of agroforestry. Extension services can play a key role in disseminating skills and technology needed by smallholder farmers for effective adoption of agroforestry and by capitalizing on farmers' knowledge. Unfortunately, as pointed out by Ansoms (2010), extension services often serve merely as the implementers of the government's agricultural agenda without considering farmers' views. In anecdotal discussions during this study, I learned that extension services often focus on implementation of governmental agricultural policy and have less time to spend with farmers. Furthermore, as argued by Twomlow et al. (2008), among African extension services, the understanding of climate processes, driving forces, and meaningful coping and adaptive strategies remains insufficient. Therefore, while extension officers may tell farmers when it is time for planting specific crops, extension service support to farmers in terms of helping them understand climate change and how to cope in the long term can be limited.

Previous studies have demonstrated that farmers adopt innovations, agroforestry in this particular case, if they not only have required skills and knowledge pertaining to the designing, planning and implementation of such innovations among other things but also if they are involved in the process of planning, implementing and evaluating innovations (Mccann et al., 1997;Wennink et al., 2006). Furthermore, smallholder farmers are an important group of actors that need to be considered in strategies concerning sustainable land use and the restoration of degraded lands (Montagnini, Francesconi, & Rossi, 2011) because of their knowledge of the landscape in which they live (McCall & Minang, 2005). Therefore, extension services should

work hand-in-hand with farmers. Innovations are more likely to be adopted when they have a high 'relative advantage' (perceived superiority to the idea or practice that it supersedes). When they are readily easy to test and learn about before adoption (Pannell & Vanclay, 2011) and economically profitable as previous studies have indicated (Drechsel, Steiner, & Hagedorn, 1996), farmers may be reluctant to endorse the idea of agroforestry as long as they have never seen successful cases of agroforestry in their agro-ecological zone.

Farmers' income level has been highlighted by a number of scientists as one of the major drivers of farmers' adoption of any given adaptation measure. Findings from this study suggest that about 91% of interviewed farmers own less than 1 hectare of total farmland per household and farming was indicated to be their main source of income. I found that the farm size per household seems to be an indicator of the household income. This finding corroborates other scientists' claims that farmers with higher household income are more likely to adopt new agricultural practices, given greater access to information and financial resources (Franzel, 1999; Knowler and Bradshaw, 2007; Croppenstedt et al., 2003; Nhemachena and Hassan, 2007; Deressa et al., 2009; Asfaw et al., 2011; Bryan et al., 2009a). A study conducted in South Africa to gauge farmers' perception of and adaptation to climate change revealed that although a majority of farmers were aware of climate change, a relatively small proportion adapted to climate change with lack of access to credit as the major inhibitor of adaptation (Gbetibouo, 2009).

Maddison (2007) suggested that some farmers perceive climate change but fail to adapt due lack of incentives or assistance to do what is in their own best longer term interests. These findings

corroborate what I found in this study and are similar to the findings of other studies where financial and technological factors have been highlighted as the major drivers of effective adaptation of agricultural practices that provide long-term benefits (Hassan et al., 2008; Deressa et al., 2008) Nhemachena & Hassan, 2007; Hisali et al., 2011; Sanchez, 2002). I argue that incentives would be more effective if they could maintain farmers' current agricultural income level until agroforestry can start generating significant returns. This is a model used in the Green Belt Movement in which farmers received incentives during the initial tree planting phase so that returns from their farmlands could be maintained until trees could begin to provide benefits (Maathai, 2004). Financial incentives should also be supplemented by effective support from extension services through experimental plots where farmers could work together with extension services to test the performance of agroforestry within their own plots. While micro-credits schemes in Rwanda are widespread and are believed to contribute to the efficiency of agricultural production, credit is virtually inaccessible to smallholder poor and uneducated farmers (Nabahungu & Visser, 2013; Ali et al., 2014; Musanze, 2013; Mutandwa & Kanyarukiga, 2016). Lack of subsidies in terms of seeds, seedlings and fertilizers can thus deter farmers' willingness to adopt agroforestry.

In Rwanda, farmers' awareness and adaptation to climate change appears to be strongly influenced by policies such as the introduction of new corn, bean, and Irish potato varieties associated with increased use of synthetic fertilizers and pesticides, and the deployment of extensions services to facilitate the adoption of the new land policy. While these schemes have resulted in an increase in crop production and are well accepted by the majority of farmers, other crops such as banana, sweet potatoes and sorghum have not been promoted and have actually

been reduced. In this regard, the policy should make it possible for farmers to experience how the adoption of agroforestry can generate benefits that outweigh the assignment of fields to monocropping systems. Furthermore, information related to weather forecasting and season change should be made available to farmers so that the timing of the planting season is not challenged by lack of adequate climate information.

References

- Africa Research Bureau. (2004). Ajzen, I. (1991). The theory of planned behaviour. Organizational Behaviour and Human Decision Processes, 50, 179-211. *De Young*, 509–526.
- Akinnagbe, O. M., Ajayi, A. R., & others. (2010). Challenges of farmer-led extension approaches in Nigeria. *World Journal of Agricultural Sciences*, 6(4), 353–359.
- Ali, D. A., Deininger, K., & Duponchel, M. (2014). Credit Constraints and Agricultural Productivity: Evidence from rural Rwanda. *Journal of Development Studies*, 50(5), 649– 665.
- Amdu, B., Ayehu, A., & Deressa, A. (2013). Farmers' perception and adaptive capacity to climate change and variability in the upper catchment of Blue Nile, Ethiopia. Retrieved from http://www.africabib.org/rec.php?RID=383156459
- Anandajayasekeram, P., Davis, K. E., & Workneh, S. (2007). Farmer field schools: an alternative to existing extension systems? Experience from Eastern and Southern Africa. *Journal of International Agricultural and Extension Education*, 14(1), 81–93.
- Ansoms, A. (2010). Views from Below on the Pro-poor Growth Challenge: The Case of Rural Rwanda. *African Studies Review*, *53*(2), 97–123.
- Asfaw, S., Shiferaw, B., Simtowe, F., & Haile, M. G. (2011). Agricultural technology adoption, seed access constraints and commercialization in Ethiopia. *Journal of Development and Agricultural Economics*, 3(9), 436–447.
- Atangana, A., Khasa, D., Chang, S., & Degrande, A. (2014a). Agroforestry for Soil Conservation. In *Tropical Agroforestry* (pp. 203–216). Springer Netherlands.

- Behera, S. K., Luo, J.-J., Masson, S., Delecluse, P., Gualdi, S., Navarra, A., & Yamagata, T.
 (2005). Paramount Impact of the Indian Ocean Dipole on the East African Short Rains: A
 CGCM Study. *Journal of Climate*, *18*(21), 4514–4530.
- Booth, D., & Golooba-Mutebi, F. (2012). Policy for agriculture and horticulture in Rwanda: a different political economy? Retrieved from http://opendocs.ids.ac.uk/opendocs/handle/123456789/2250
- Bryan, E., Deressa, T. T., Gbetibouo, G. A., & Ringler, C. (2009a). Adaptation to climate change in Ethiopia and South Africa: options and constraints. *Environmental Science & Policy*, *12*(4), 413–426.
- Bryan, E., Deressa, T. T., Gbetibouo, G. A., & Ringler, C. (2009b). Adaptation to climate change in Ethiopia and South Africa: options and constraints. *Environmental Science & Policy*, 12(4), 413–426.
- Bryan, E., Ringler, C., Okoba, B., Roncoli, C., Silvestri, S., & Herrero, M. (2013). Adapting agriculture to climate change in Kenya: Household strategies and determinants. *Journal* of Environmental Management, 114, 26–35.
- Cary, J. W., & Wilkinson, R. L. (1997). Perceived profitability and farmers 'conservation behaviour. *Journal of Agricultural Economics*, *48*(1–3), 13–21.
- Charles, N., & Rashid, H. (2007). *Micro-level analysis of farmers' adaptation to climate change in Southern Africa*. IFPRI Discussion Paper 00714.
- Chouinard, H. H., Paterson, T., Wandschneider, P. R., & Ohler, A. M. (2008). Will FarmersTrade Profits for Stewardship?: Heterogeneous Motivations for Farm Practice Selection.*Land Economics*, 4(1), 66–82.

- Clay, D. C., & Lewis, L. A. (1990). Land use, soil loss, and sustainable agriculture in Rwanda. *Human Ecology*, 18(2), 147–161.
- Dale, V. H., & Polasky, S. (2007). Measures of the effects of agricultural practices on ecosystem services. *Ecological Economics*, 64(2), 286–296.
- Davis, K., Nkonya, E., Kato, E., Mekonnen, D. A., Odendo, M., Miiro, R., & Nkuba, J. (2012).
 Impact of Farmer Field Schools on Agricultural Productivity and Poverty in East Africa.
 World Development, 40(2), 402–413.
- de-Graft Acquah, H., & Onumah, E. E. (2011). Farmers Perception and Adaptation to Climate Change: An Estimation of Willingness to Pay. *Agris On-Line Papers in Economics & Informatics*, 3(4), 31–39.
- Deressa et al,. (2009). Determinants of farmers' choice of adaptation methods to climate change in the.pdf. (n.d.). Retrieved from http://www.dspace.up.ac.za/bitstream/handle/2263/9270/Deressa_Determinants(2009).pd f?sequence=1
- Deressa, T. T., Hassan, R. M., & Ringler, C. (2011). Perception of and adaptation to climate change by farmers in the Nile basin of Ethiopia. *The Journal of Agricultural Science*, 149(01), 23–31.
- Deressa, T. T., Hassan, R. M., Ringler, C., Alemu, T., & Yesuf, M. (2008). Analysis of the determinants of farmers' choice of adaptation methods and perceptions of climate change in the Nile Basin of Ethiopia [in Amharic]. International Food Policy Research Institute (IFPRI). Retrieved from https://ideas.repec.org/p/fpr/resbrf/15(9)amh.html

- Diem, J. E., Hartter, J., Salerno, J., McIntyre, E., & Grandy, A. S. (n.d.). Comparison of measured multi-decadal rainfall variability with farmers' perceptions of and responses to seasonal changes in western Uganda. *Regional Environmental Change*, 1–14.
- Drechsel, P., Steiner, K. G., & Hagedorn, F. (1996). A review on the potential of improved fallows and green manure in Rwanda. *Agroforestry Systems*, *33*(2), 109–136.
- Fosu-Mensah, B. Y., Vlek, P. L. G., & MacCarthy, D. S. (2012). Farmers' perception and adaptation to climate change: a case study of Sekyedumase district in Ghana. *Environment, Development and Sustainability*, 14(4), 495–505.
- Franz, N. K., Piercy, F., Donaldson, J., Westbrook, J., & Richard, R. (2010). Farmer, agent, and specialist perspectives on preferences for learning among today's farmers. *Journal of Extension*, 48(3), 3RIB1.
- Gandure, S., Walker, S., & Botha, J. J. (2013). Farmers' perceptions of adaptation to climate change and water stress in a South African rural community. *Environmental Development*, 5, 39–53.
- Gbetibouo, G. A. (2009). Understanding Farmers' Perceptions and Adaptations to Climate Change and Variability: The Case of the Limpopo Basin, South Africa. Intl Food Policy Res Inst.
- Gellman, M. D., & Turner, J. R. (Eds.). (2013). Theory of Planned Behavior. In *Encyclopedia of Behavioral Medicine* (pp. 1964–1964). Springer New York. Retrieved from http://link.springer.com.antioch.idm.oclc.org/referenceworkentry/10.1007/978-1-4419-1005-9 1699.
- Grossman, J. M. (2003). Exploring farmer knowledge of soil processes in organic coffee systems of Chiapas, Mexico. *Geoderma*, 111(3–4), 267–287.

- Hartter, J. (2009). Attitudes of rural communities toward wetlands and forest fragments around Kibale National Park, Uganda. *Human Dimensions of Wildlife*, *14*(6), 433–447.
- Hassan, R., Nhemachena, C., & others. (2008). Determinants of African farmers' strategies for adapting to climate change: Multinomial choice analysis. *African Journal of Agricultural* and Resource Economics, 2(1), 83–104.
- Hisali, E., Birungi, P., & Buyinza, F. (2011). Adaptation to climate change in Uganda: Evidence from micro level data. *Global Environmental Change*, *21*(4), 1245–1261.
- Huggins, C. (2009). Agricultural Policies and Local Grievances in Rural Rwanda. *Peace Review*, *21*(3), 296–303.
- Ingram, J. (2008). Agronomist–farmer knowledge encounters: an analysis of knowledge exchange in the context of best management practices in England. *Agriculture and Human Values*, *25*(3), 405–418.
- Islam, M. M., Gray, D., Reid, J., & Kemp, P. (2011). Developing sustainable farmer-led extension groups: lessons from a Bangladeshi case study. *The Journal of Agricultural Education and Extension*, 17(5), 425–443.
- Karali, E., Brunner, B., Doherty, R., Hersperger, A. M., & Rounsevell, M. D. A. (2013). The Effect of Farmer Attitudes and Objectives on the Heterogeneity of Farm Attributes and Management in Switzerland. *Human Ecology: An Interdisciplinary Journal*, 41(6), 915–926.
- Kiptot, E., & Franzel, S. (2013). Voluntarism as an investment in human, social and financial capital: evidence from a farmer-to-farmer extension program in Kenya. *Agriculture and Human Values*, 31(2), 231–243.

- Kiptot, E., Franzel, S., Hebinck, P., & Richards, P. (2006). Sharing seed and knowledge: farmer to farmer dissemination of agroforestry technologies in western Kenya. *Agroforestry Systems*, 68(3), 167–179.
- Läpple, D., & Kelley, H. (2013). Understanding the uptake of organic farming: accounting for heterogeneities among Irish farmers.
- Maathai, W. (2004). *The Green Belt Movement: Sharing the approach and the experience*. Lantern Books. Retrieved from https://books.google.com/books?hl=en&lr=&id=4bDghrBcXzUC&oi=fnd&pg=PR7&dq =Wangari+Mathai&ots=_gBEytGEdr&sig=3BDpIA6HXMjjvtNOtktmV821eTc
- Maddison, D. J. (2007). The perception of and adaptation to climate change in Africa. *World Bank Policy Research Working Paper*, (4308).
- Maharjan, S. K., Sigdel, E. R., Sthapit, B. R., & Regmi, B. R. (2011). Tharu community's perception on climate changes and their adaptive initiations to withstand its impacts in Western Terai of Nepal. *International NGO Journal*, 6(2), 35–42.
- Manandhar, S., Vogt, D., Perret, S., & Kazama, F. (2011). Adapting cropping systems to climate change in Nepal: a cross-regional study of farmers' perception and practices. *Regional Environmental Change*, 11(2), 335–348.
- Mati, B. M. (2000). The influence of climate change on maize production in the semi-humid– semi-arid areas of Kenya. *Journal of Arid Environments*, *46*(4), 333–344.
- Mbow, C., Smith, P., Skole, D., Duguma, L., & Bustamante, M. (2014). Achieving mitigation and adaptation to climate change through sustainable agroforestry practices in Africa. *Current Opinion in Environmental Sustainability*, 6, 8–14.

- McBean, G., & Ajibade, I. (2009). Climate change, related hazards and human settlements. *Current Opinion in Environmental Sustainability*, 1(2), 179–186.
- McCALL, M. K., & Minang, P. A. (2005). Assessing participatory GIS for community-based natural resource management: claiming community forests in Cameroon. *Geographical Journal*, 171(4), 340–356.
- Mccann, E., Sullivan, S., Erickson, D., & Young, R. D. (1997). Environmental Awareness, Economic Orientation, and Farming Practices: A Comparison of Organic and Conventional Farmers. *Environmental Management*, 21(5), 747–758.
- McGinty, M. M., Swisher, M. E., & Alavalapati, J. (2008). Agroforestry adoption and maintenance: self-efficacy, attitudes and socio-economic factors. *Agroforestry Systems*, 73(2), 99–108.
- Mertz, O., Mbow, C., Reenberg, A., & Diouf, A. (2009). Farmers' perceptions of climate change and agricultural adaptation strategies in rural Sahel. *Environmental Management*, 43(5), 804–816.
- Montagnini, F., Francesconi, W., & Rossi, E. (2011). *Agroforestry as a tool for landscape restoration*. New York: Nova Science Publishers. Retrieved from http://search.ebscohost.com/login.aspx?direct=true&scope=site&db=nlebk&db=nlabk&A N=431069
- Mugalavai, E. M., Kipkorir, E. C., Raes, D., & Rao, M. S. (2008). Analysis of rainfall onset, cessation and length of growing season for western Kenya. *Agricultural and Forest Meteorology*, 148(6), 1123–1135.
- Muhinyuza, J. B., Shimelis, H., Melis, R., Sibiya, J., & Nzaramba, M. N. (2012). Participatory assessment of potato production constraints and trait preferences in potato cultivar

development in Rwanda. *International Journal of Development and Sustainability*, 1(2), 358–380.

- Murasa, A. (2013). The role of farmers' organisations in enhancing economic efficiency in potato production in Musanze District, Rwanda. Retrieved from http://www.dspace.mak.ac.ug/handle/10570/2894
- Musanze District. (2013). Musanze development plan (2013-2018). Kigali: Government of Rwanda.
- Musanze, R. (2013). Determinants of Farmers' Participation in Formal Credit Markets in Rural Rwanda.
- Musshoff, O., & Hirschauer, N. (2008). Adoption of organic farming in Germany and Austria: an integrative dynamic investment perspective. *Agricultural Economics*, *39*(1), 135–145.
- Mutandwa, E., & Kanyarukiga, R. (2016). Understanding the role of forests in rural household economies: experiences from the Northern and Western provinces of Rwanda. *Southern Forests: A Journal of Forest Science*, 0(0), 1–8.
- Nabahungu, N. L., & Visser, S. M. (2013). Farmers' Knowledge and Perception of Agricultural Wetland Management in Rwanda. *Land Degradation & Development*, *24*(4), 363–374.
- Nahayo, A., Ekise, I. E., & Niyigena, D. (2013). Assessment of the contribution of Non Timber Forest Products to the improvement of local people's livelihood in Kinigi sector, Musanze District, Rwanda. *Ethiopian Journal of Environmental Studies and Management*, 6(6), 698–706. https://doi.org/10.4314/ejesm.v6i6.13
- Nhemachena, C., & Hassan, R. (2007). *Micro-level analysis of farmers adaption to climate change in Southern Africa*. Intl Food Policy Res Inst.
- NISR. (2012). EICV3 DISTRICT PROFILE: Musanze. Rwanda: National Institute of Statistics of Rwanda.
- Osbahr, H., Dorward, P., Stern, R., & Cooper, S. (2011). Supporting agricultural innovation in Uganda to respond to climate risk: linking climate change and variability with farmer perceptions. *Experimental Agriculture*, 47(02), 293–316.
- Pannell, D. J., & Vanclay, F. (2011). *Changing land management adoption of new practices by rural landholders* (Vols. 1–1 online resource (x, 195 p.): ill.). Collingwood, Vic.: CSIRO
- Parry, M. L. (2007). Climate Change 2007: impacts, adaptation and vulnerability: contribution of Working Group II to the fourth assessment report of the Intergovernmental Panel on Climate Change (Vol. 4). Cambridge University Press.
- Pritchard, M. F. (2013). Land, power and peace: Tenure formalization, agricultural reform, and livelihood insecurity in rural Rwanda. *Land Use Policy*, 30(1), 186–196.
- REMA. (2011). Atlas of Rwanda's changing environment: implications for climate change resilience. Kigali, Rwanda: REMA.
- Rugazura, E., & Murugesan, R. (2015). Opportunities for rural development in Musanze District, Africa: a rural livelihood analysis. *International Journal of Business Management and Economic Research*, 6(4), 231–248.
- Sanchez, P. A. (2002). Soil fertility and hunger in Africa. *Science(Washington)*, 295(5562), 2019–2020.
- Scarborough, V., Killough, S., Johnson, D. A., Farrington, J., & Axinn, G. H. (1997). 2.
 Challenges to agricultural extension in the twenty-first century. In *Farmer-led Extension* (pp. 13–22). Practical Action Publishing.

- Schuck, E. C., Nganje, W., & Yantio, D. (2002). The role of land tenure and extension education in the adoption of slash and burn agriculture. *Ecological Economics*, *43*(1), 61–70.
- Sheeder, R. J., & Lynne, G. D. (2011). Empathy-conditioned conservation: "Walking in the shoes of others" as a conservation farmer. *Land Economics*, 87(3), 433–452.
- Stainback, G. A., Masozera, M., Mukuralinda, A., & Dwivedi, P. (2011). Smallholder Agroforestry in Rwanda: A SWOT-AHP Analysis. *Small-Scale Forestry*, 11(3), 285– 300.
- Swart, R., & Raes, F. (2007). Making integration of adaptation and mitigation work: mainstreaming into sustainable development policies? *Climate Policy*, 7(4), 288–303.
- Tambo, J. A., & Abdoulaye, T. (2013). Smallholder farmers' perceptions of and adaptations to climate change in the Nigerian savanna. *Regional Environmental Change*, 13(2), 375– 388.
- Tessema, Y. A., Aweke, C. S., & Endris, G. S. (2013). Understanding the process of adaptation to climate change by small-holder farmers: the case of east Hararghe Zone, Ethiopia. *Agricultural and Food Economics*, 1(1), 13.
- Twomlow, S., Mugabe, F. T., Mwale, M., Delve, R., Nanja, D., Carberry, P., & Howden, M. (2008). Building adaptive capacity to cope with increasing vulnerability due to climatic change in Africa–A new approach. *Physics and Chemistry of the Earth, Parts A/B/C*, 33(8), 780–787.
- Wali, U. G., Nhapi, I., Ngombwa, A., Banadda, N., Nsengimana, H., Kimwaga, R. J., &
 Nansubuga, I. (2011). Modelling of Nonpoint Source Pollution in Akagera
 Transboundary River in Rwanda. *Open Environmental Engineering Journal*, 4, 124–132.

- Wennink, B., Heemskerk, W., & KIT Royal Tropical Institute. (2006). Farmers' organizations and agricultural innovation : case studies from Benin, Rwanda and Tanzania. KIT Publishers.
- Whitmore, A. P., & Schröder, J. J. (2007). Intercropping reduces nitrate leaching from under field crops without loss of yield: a modelling study. *European Journal of Agronomy*, 27(1), 81–88.
- Yaro, J. A. (2013). The perception of and adaptation to climate variability/change in Ghana by small-scale and commercial farmers. *Regional Environmental Change*, 13(6), 1259– 1272.
- Zimmerman, L. S., Byizigiro, & Vaillant. (2012). Rapid risk and capacities assessment and livelihoods prolifering in Nyabihu, Musanze and Burera Districts affected by floods and landslides. Kigali: MIDIMAR.

Chapter 4: Ecosystem services in riparian agricultural landscapes: A study of smallholder farmers' perceptions in the northern highlands of Rwanda

Abstract

Riparian zones in many rural developing regions in Africa are severely degraded at the expense of food production. Despite a recognition of the potential of agricultural landscapes to generate or maintain ecosystem services (ESs), agricultural intensification in many developing countries in Africa, including Rwanda, is being promoted as a way to support food security and economic growth at the expense of ESs on which sustainable agriculture depends. Establishing multifunctional riparian buffers within farmlands is vital for the restoration of ESs in agricultural landscapes in response to declining water quality, stream bank erosion, soil erosion, and degradation of riparian farmlands, in addition to fuel wood demands among the majority of rural smallholder farmers. Views and experiences about the importance of ESs, and strategies for restoring and maintaining those ESs were elicited from smallholder farmers in the Upper Mukungwa Watershed in northern Rwanda. This region is prone to flooding and soil erosion due to its topography and climate variability. The 430 farmers interviewed in this study recognized benefits of riparian buffers: water quality, fluvial flood mitigation, provision of fuel wood, fodder for livestock, medicinal plants, stakes to support beans, building material, soil conservation and increased farm yield. Farmers reported their willingness to adopt riparian buffers that are economically profitable and that can meet their various short-term multiple needs. However this willingness was directly related to income level. The Theory of Planned Behavior (TPB) suggests that farmers' perceptions of ESs as informed by personal experience, local sources of knowledge, and external sources of techno-scientific information, and perceived ability to restore or/and maintain ESs are essential for effective policies geared toward food

security and environmental stewardship. Smallholder farmers' perception of the importance of securing ESs through the adoption of functional riparian buffers need to be captured within a larger policy framework, the benefits of which should extend beyond short-term goals identified by farmers to a larger, multifunctional agricultural landscape that provides not only regulating but also provisioning ESs. Once we understand perspectives of farmers, we can focus on ESs as a means to promote and support sustainable agriculture.

Key words

Agroforestry, climate change, food security, soil erosion, soil fertility loss, Rwanda.

Introduction

Synergies between ecosystem services (ESs), agriculture and food security have drawn attention among many ecologists, agricultural scientists, and policy makers (Cruz-Garcia et al., 2016). ESs are benefits that people obtain from ecosystems (Millennium Ecosystem Assessment, 2005), and could be instrumental in supporting sustainable agriculture if well managed (Cruz-Garcia et al., 2016; Reid, 2016). Ongoing discourse about the intricate synergies between agricultural landscapes and the ESs they can support suggests that rehabilitation and maintenance of functional riparian buffers, coupled with intercropping practices, including agroforestry, represent an effective strategy not only to address problems of soil loss and non-point pollution in streams due to agricultural practices (National Agroforestry Center (U.S.), 2012; Montagnini et al., 2011; Etchevers et al., 2009; Gregory et al., 2009; Sanchez, 2002; Zougmoré et al., 2000; Horwith, 1985 but also to provide additional valuable ESs including fuel wood, stakes for climbing beans, building material and medicinal plants.

Unfortunately, agricultural intensification in African countries is often being promoted as a way to support food security and economic growth (Lichtfouse, 2016; Campbell et al., 2014; Booth and Golooba-Mutebi, 2012; 2011) at the expense of ESs, including loss of organic matter, soil compaction, erosion and other impacts on soil (Lal, 2015; Cantore, 2011;Tilman et al., 2011; Wali et al., 2011; Gordon et al., 2010; Herder et al., 2010; Viala, 2008; Bossio et al., 2007; Okalebo et al., 2007; RADA, 2005; Millennium ESs, 2005; Sanchez, 2002. Understanding synergies between sustainable management of farmland that maintain ESs and improves crop yield is key to food security (Cruz-Garcia et al., 2016; Reid, 2016).

Farm management approaches, including tillage, crop diversity, crop rotation, cover cropping, agroforestry, and maintenance of riparian buffers have the potential to ensure both food security and provide ESs that support the provisioning services, including pollination, pest control, genetic diversity for future agricultural use, soil retention, regulation of soil fertility and nutrient cycling (Rasmussen et al., 2016; Schulp et al., 2014; Wunder et al., 2014; de Groot, Alkemade, Braat, Hein, & Willemen, 2010). Riparian buffers are defined as transitional semiterrestrial/semiaquatic areas regularly influenced by fresh water, usually extending from the edges of water bodies to the edges of upland communities (Naiman, 2010). Riparian buffer contribute to provision of ESs including but not limited to intercepting and retaining nutrients and sediments, protecting banks of water bodies against erosion, improving microclimate in adjacent fields, creating more connectivity in landscapes and improving pollination (Swinton et al., 2007; Mayer et al., 2007; Mander et al., 1997). Riparian buffer trees, shrubs and grass show different relative effectiveness for providing specific benefits. For example, some tree and shrub species are very effective in stabilizing stream bank erosion while grass is very effective in filtering sediments, nutrients and pesticides. Trees are particularly also very effective in flood protection (Dosskey, 1997). The minimum acceptable riparian buffer width depends on the anticipated function of the buffer. For stream bank erosion and sediment control, a riparian buffer width might go up to 30 meters (Wenger, 1999) and soluble nutrient control requires a minimum riparian buffer width of about 100 meters (Dosskey, 1997).

Most previous studies on ESs have put more emphasis on natural habitats, such as protected natural forests for example, than human-dominated agricultural landscapes (e.g. Costanza et al., 1997; Kareiva, 2011; Sweeney et al., 2004; Cruz-Garcia et al., 2016; Schulp et al., 2014). Failure

to fully incorporate the agricultural landscape in assessments of ESs such as freshwater resource quality, fuel energy, pollination and many others has potentially important consequences for management of land and resources (Rasmussen et al., 2016). In agricultural landscapes, the restoration of ESs on farmland depends for the most part on landowners goals, needs and also their knowledge or understanding (Doré et al., 2011; Bommarco et al., 2013). Therefore, it is crucial for policy makers and agricultural managers and conservation actors to understand farmers' perceptions and definition of ESs to design agricultural land management policies and tailor interventions that profitably meet farmers' goals (Greiner et al., 2009; Adger et al., 2007) while meeting national conservation and economic goals as well. However, some of the goals may be in conflict and the revision of national goals may be warranted if the other goals are to be achieved.

Studies on factors that drive farmers' adoption of particular agricultural practices have revealed a whole range of determinants which are not organized in a consolidated framework to help understand what affects farmer behaviors (e.g. (Zegeye et al., 2010; Swinton et al., 2007; Wejnert, 2002; Temesgen et al., 2009; Hassan et al., 2008; Bryan et al., 2013). The Theory of Planned Behavior (TPB), originally developped by Ajzen (1991) and supported by recent research (Kautonen et al., 2015; Case et al., 2016) presents a systematic framework for understanding the major drivers of human intentions and attitudes. TPB explains the relationship between attitudes, social norms and perceived behavior is driven by perceptions based on past experience, perceived self-efficacy in terms of ability to perform a behavior with success, and social norms, (Ajzen, 1991; Zimmerman, 2000; Armitage & Christian, 2004; Frank et al., 2011).

TBS is useful in the context of my research because it sheds a light on factors that need to be considered or discussed for a better framing of my research and particularly with regard to drivers that could affect smallholder farmers' willingness and intention to adopt climate adaptive strategies.

According to the TPB, "human behavior is guided by three kinds of considerations: beliefs about the likely consequences or other attributes of the behavior (behavioral beliefs), beliefs about the normative expectations of other people (normative beliefs), and beliefs about the presence of factors that may further or hinder performance of the behavior (control beliefs). In their respective aggregates, behavioral beliefs produce a favorable or unfavorable *attitude toward the behavior;* normative beliefs result in perceived social pressure or *subjective norm;* and control beliefs give rise to *perceived behavioral control,* the perceived ease or difficulty of performing the behavior. In combination, attitude toward the behavior, subjective norm, and perception of behavioral control lead to the formation of a behavioral *intention.* Finally, given a sufficient degree of *actual* control over the behavior, people are expected to carry out their intentions when the opportunity arises. Intention is thus assumed to be the immediate antecedent of behavior" (Ajzen, 2002).

While the application of TPB has been extensively explored in health, economics and social psychology fields (Godin & Kok, 1996; Kautonen et al., 2015; Ajzen, 2002), its application in the environmental field is less common (e.g.Holloway & Ilbery, 1996; Kaiser et al., 1996; Oreg & Katz-Gerro, 2006). Yet this theory can provide critical insights for addressing the tensions between agriculture and ESs. These include failure to gauge farmers' attitudes and motivation

for adopting certain farming practices that can impair our understanding of farmer adaptive behavioral choices (Frank et al., 2011). For example, recent research indicates that farmers' perception of their capabilities to produce given targets termed by Bandura (1977) as "selfefficacy" is another important component that can strongly influence farmers' intentions (Frank et al., 2011; Fielding et al., 2008; Armitage & Christian, 2004). Farmers who believe in their potential to implement best agricultural practices that foster ESs will have a positive position for the adoption of those practices. Investigating farmers' perceptions of ESs as informed by their personal experience, local sources of knowledge, external sources of techno-scientific information, and their perceived ability to restore or/and maintain ESs is essential for effective policies geared toward food security and environmental stewardship.

This study examines whether or not smallholder farmer attitudes for adoption of agricultural practices that maintain ESs within riparian buffers, agroforestry in this case, are independent of their perceptions of factors they believe could hinder or promote the uptake of agroforestry practices, and of other climatic and socio-economic factors as well. This research sets out to explicitly answer the following questions:

- What is the relationship between crop choice among smallholder farmers, belief that some crops are more vulnerable to soil erosion than others, and economic profitability?
- What is the relationship between smallholder farmer income level and crop choice?
- What factors influence smallholder farmers' willingness to maintain ESs, benefits they anticipate from the ecosystem, loss of benefits they have experienced due to ecosystem degradation, or socio-economic factors such as level of income and farm size?

• What is the relationship between choice of tree species for riparian buffer zones, trends in cultivation of certain crops, agriculture policies in Rwanda, and fuel wood needs?

To address these questions, 430 farm households were surveyed in the Upper Mukungwa Watershed of northern Rwanda. This region was chosen because it is an ideal location to study these questions given that it is in a region that has, over the past few years, been inflicted with frequent floods. Crop cultivation in many places extends to the edges of the streams and steep slopes.

Methods

Study area.

The study was conducted in the Upper Mukungwa Watershed (Figure 2) in Musanze District, Northern Province, Rwanda and covered about 60% of the Musanze District land surface area located in the northern highlands of Rwanda. The region lies on volcanic soil dominated by alfisols, andosols and entisols types (Nzeyimana et al., 2016) in a moderate-temperature and humid climate zone with an annual temperature and rainfall average of 16⁰ C and 1400mm respectively (Musanze District, 2014). The rainfall seasons in the region are significantly influenced by the movement of the Inter-Tropical Conversion Zone (ITCZ) which oscillates between the northern and southern tropics over the course of one year (McSweeney et al., 2010). The Rwandan climate is characterized by a bimodal rainfall pattern with the major rainy season typically between February and June with a peak in April, and a minor rain between September and December with a peak in November. The agriculture crop calendar is organized in two seasons (Table 4-1). The region is prone to soil erosion due to steep slopes and intense rainfall (Nahayo et al., 2013; Cantore, 2011; MINAGRI, 2009; RADA, 2005).

Table 4-1

Rwanda Crop Calendar for Major Crops (Source: FAO/GIEWS accessible at http://www.fao.org/giews/countrybrief/country/RWA/pdf/RWA.pdf). Black in the table indicates sowing time, lighter black is growing time, lightest black is harvest time and black lines is lean period, which is a time during which there is not enough food

Major Food crops	Jan.	Feb.	Mar.	Apr.	May	Jun.	Jul.	Sep.	Oct.	Nov.	Dec.
Beans (A season)											
Beans (B season)											
Maize and sorghum (A season)							1				
Maize and sorghum (B season)											
Lean Period											

The Musanze District is 76% agricultural land. Over 90% of its population relies on smallholder agriculture for its economy. The population density, at 820 pers/km² (Musanze District, 2014), is one of the highest in the country and in Africa. Dominant staple crops in the Musanze District include bananas, maize, sorghum, beans, potatoes, and sweet potatoes. Most families are very poor. The Musanze landscape is dominated by agriculture.

Superpixels and farmer sampling.

Sampling units consisted of seventy 170-m radius areas called superpixels (Hartter, 2009; Goldman et al., 2008) randomly distributed across the Upper Mukungwa Watershed study area (Figure 4-1) which covers about 243km². In total 430 adult smallholder farmers identified as heads of households or spouses to heads of households were individually interviewed (approximately six per superpixel). Smallholder farmers who have lived in the area for at least 10

years were considered for inclusion in this study because the assumption was that they would have experience with environmental changes on their farmland in the study area; these individuals were in addition assumed to be the key individuals involved in decision-making about agricultural adaptation practices. Working with field assistants who had been trained to collect data for this research, we traveled throughout the study area to meet the smallholder farmers in their households. After arriving in a given superpixel, sampling would start from the center of the superpixel outward. Households were selected based on their availability for the interview. Where the head of a household was not available, we would interview her (his) spouse if she (he) was available for the interview. Wherever possible, we would apply gender balance to select interview participants.

Before a smallholder farmer was interviewed, we would check whether the farmer has lived in the area for at least 10 years and had farming as the main livelihood activity. Once conditions met, then we would ask the participants whether or not she (he) accepted to be interviewed. People who were drunk or had obvious mental disabilities, or were not willing to participate were not considered for the interviews. In total two men did not want to be interviewed unless they were offered a beer or cash payment, and another was drunk and it would have been inappropriate to interview him.



Figure 4-1. Survey superpixels (in orange circles, n=70) across the upper inhabited Mukungwa watershed study area, Rwanda

Farmer survey on perception of the importance of ESs.

ESs vs. economic profitability

To test the hypothesis that the maintenance of ESs is considered less important in farmers' agricultural practices than economic profitability, I explored whether criteria driving smallholder farmers' choice of plants to grow within riparian zones (dependent variables) were independent of perception of ESs farmers anticipate from functional riparian buffers, issues of concern experienced in farm plots adjacent to streams, or level of income or farm size (independent variables).

I first asked farmers: (a) their agreement on a list of criteria, including proximity to streams, slope, soil characteristics, changes of seasons, government policy, nutritional needs and economic profitability, considered to influence their decision to choose which crop to grow in any given season; (b) to what degree they believe each type of commonly grown crop is vulnerable to erosion; and (c) their agreement on whether each of the commonly cultivated crops has increased in cultivation in recent years compared to ten years ago. Possible responses were in one of three categories: 1 = strongly agree, 2 = somewhat agree, 3 = disagree. I performed a Chi-Square analysis to examine if there was an association between variable (a) and variables (b) and (c).

Perceived role of riparian buffers and maintenance of regulating ESs.

I explored whether smallholder farmers' willingness or unwillingness to establish a riparian buffer and their choices of particular plants for riparian buffers were independent of their anticipated regulatory and provisioning ESs as well as their level of income. I used a test of independence to analyze the relationship between the perception that fluvial floods are a concern (strongly agree=1 and otherwise =0) and the perceived role of riparian buffers in abating the impact of fluvial floods (strongly agree = 1 and otherwise = 0) as well as their level of income or farm size per household. Questions I asked to the farmers in this respect included their agreement on the importance of riparian buffers in addressing the issue of floods, whether farmers were willing to adopt riparian buffers, benefits expected from riparian buffers and tree species they propose for riparian buffers.

Results

Economic profitability was ranked the highest among crop choice criteria, followed by nutritional needs and government policy. Proximity to streams, slope and soil characteristics play a relatively minor role in determining crop choice (Figure 4-2).



Figure 4-2. Proportion of total farmers who agree on crop choice criteria. The dark bars indicate the degree of strong agreement and the light grey bars indicate the degree of disagreement

Regarding the question about which crops are perceived to be the most vulnerable to erosion, most smallholder farmers strongly agreed that beans, corn and Irish potatoes are most susceptible to damage due to heavy rains that cause erosion (Figure 4-3).



Figure 4-3. Proportion of all respondents in agreement on crop susceptibility to erosion in the Upper Mukungwa Watershed, Rwanda

Cultivation of corn, beans and potatoes, according to famers in the Upper Mukungwa Watershed, has increased more than other crops in the Upper Mukungwa Watershed in recent years compared to ten years ago (Figure 4-4).



Figure 4-4. Increase in farm plots allocated to cultivation of certain crops as reported by survey participants in the Upper Mukungwa Watershed, Rwanda from 2009 – 2014. Y-axis is the percent of smallholder farmers reporting the increase of farm plots planted with specific crops

Perceived increase in the cultivation of corn, beans, and potatoes was not independent of the perception of economic profitability as criteria for crop choice ($\chi 2 = 461.87$, df = 20, p < 0.001), beans ($\chi 2 = 457.57$, df = 20, p < 0.001), and Irish potatoes ($\chi 2 = 549.75$, df = 20, p < 0.001). I also found a significant association between strong agreement on economic profitability (Figure 4-2) as the predominant criteria for crop choice and strong agreement that beans ($\chi 2 = 64.19$, df = 15, p<0.001), corn ($\chi 2 = 448.28$, df = 20, p < 0.001) and Irish potatoes (59.05, df = 15, p < 0.001) are the most vulnerable to erosion compared to other crops grown in the area (Figure 4-3). Smallholder farmer perception of an increase in cultivation of beans, corn and Irish potatoes is not independent of their belief that the same crops are the most vulnerable to erosion if compared to other food crops grown in the area.

Smallholder farmers in the study plant crops right down to the streams edges. When I asked farmers which crops they would prefer to plant in riparian zones, the vast majority of smallholder farmers preferred corn and beans. Irish potatoes ranked fourth (Figure 4-5).



Figure 4-5. Plants preferred by smallholder farmers for riparian zones in the Upper Mukungwa Watershed, Rwanda

Perception of soil erosion, floods and riparian zones.

I analyzed the relationship between the perception that fluvial floods are a concern (strongly agree=1 and otherwise =0) and the perceived role of riparian buffers in abating the impact of fluvial floods (strongly agree = 1 and otherwise = 0) to examine if farmers in the Upper Mukungwa Watershed think that riparian buffers could be a solution to the problem of fluvial floods. There was a significant interaction between perception of fluvial floods and the role of riparian buffers ($\chi 2$ (1) = 61.47, p < .01). Respondents who strongly agreed that fluvial floods are a problem were also in support of riparian buffer zones as a solution to adverse impacts of fluvial floods.

There was also a significant association between the perception that farms adjacent to streams experience soil erosion and sediment and nutrient loss into adjacent streams (strongly agree = 1 and else =0) and the role of a riparian buffers in controlling sediment and nutrient loss into streams (strongly agree = 1 and else = 0) indicating these are not independent ($\chi 2$ (1) = 26.43, p < .01). Respondents who strongly agreed that soil erosion within farms proximal to streams is a serious threat were more likely to support the idea that establishing a riparian buffer would help entrap sediments and mitigate soil erosion within farmlands adjacent to streams.

Smallholder farmers' willingness to manage riparian zones and challenges.

I examined whether smallholder farmers' willingness to maintain ESs is associated with expectation of benefits from the ecosystem, with concern about the degradation of ESs they have benefited from in the past, or socio-economic factors such as level of income and farm size. I first asked participants if they were willing to adopt agricultural practices that help control soil and nutrient loss, improve water quality in streams and maintain stream bank stability (1 = Agree, 0 = Do not agree). Results show that 98% of participants reported they were willing to maintain ecosystems services on their land. A binary regression analysis shows that willingness to maintain those ESs was associated with income level (χ^2 (1) = 7.9, p = .044). The higher the level of income, the more likely the willingness to maintain ESs.

Secondly, I examined farmer level of agreement about the importance of ESs generated from best agricultural practices within riparian zones in the area, which include, among others, provision of food, fuel wood, building materials, medicinal plants, fluvial flood buffering, wood poles to support beans, and hay for livestock (1 = Strongly agree, 2 = somewhat agree, 3 =

disagree). Fuel wood, stakes to support pole beans, hay for livestock, and reduced use of synthetic fertilizers and maintenance of stream bank stability rated very important by the majority of respondents (more than 70% scores each). Another interesting finding is that, while stream bank stability was considered to be very important by the majority of respondents (70%), water for domestic use was rated very important by a smaller number of participants (less than 50% of scores) (Figure 4-6).



Figure 4-6. Types of ESs expected from fields adjacent to streams if well maintained riparian zones are present, according to respondents farming in the Upper Mukungwa Watershed, northern Rwanda

Lastly I asked smallholder farmers about their concerns regarding farm plots that are proximal to streams, related to erosion, water logging, and crop damage caused by fluvial floods and stream bank erosion. Results show that concern about crop damage due to erosion had the highest scores

(74%), followed by water logging (72%), fluvial floods (72%) and finally stream bank erosion (71%). A logistic regression shows a significant association between a strong agreement about the maintenance of a functional riparian buffer zone and fluvial floods experienced in fields adjacent to streams ($\chi^2(2) = 12.3$, p = .040). In other words, smallholder farmers who perceive that fluvial floods are of serious concern are more likely to maintain riparian buffer zones.

Plants and ESs provision: Drivers of smallholder farmer decision making.

I examined farmers' choice of plant species they consider to be important and appropriate for a riparian buffer. Smallholder farmers identified plants they consider important for the riparian buffer (Figure 4-7). Results reveal that *Eucalyptus* spp. was chosen by a majority of smallholder farmers.





I tested if the choice of plants was associated with perception of the importance of ESs that functional riparian zones can offer, issues of concern experienced in farm plots adjacent to

streams, level of income or farm size. Results show that the choice of *Eucalyptus* spp. was significantly associated with respondents' strong agreement that riparian buffers offer food, fuelwood, building material, medicinal plants, poles to support beans, hay for livestock and income (Table 4-2). Some smallholder farmers conceded that they prefer *Eucalyptus* spp. because "it grows fast,....its branches offer strong and durable sticks to support beans...., and it burns well when cooking food,provides medicine for a cold,and good quality timber". Besides *Eucalyptus* spp., *Grevillea robusta* was associated with respondents' awareness of the importance of functional riparian zones to provide hay for livestock. The choice of *Arundinaria alpina* (bamboo), was on the other hand associated with respondents' awareness of the role of functional riparian zones in buffering floods, providing poles to support beans and protecting stream water. *Pennisetum purpureum* was associated with awareness about the potential of riparian buffers to provide fuelwood.

The choice of plant species types was not associated with household income level or farm size except for *Eucalyptus* spp. which was associated with income level. While *Eucalyptus* was mostly associated with provisional services (fluel wood and building material), *Arundinaria alpina* was more associated with perceived environmental regulation (flood control and retainment of sediments for water quality control).

Table 4-2

Chi-square Test Results Showing Relationship between Farmers' Choice of Plant Species for the Riparian Buffer and Benefits Farmers Expected from a Functional Riparian Buffer (The grey cells in the table show where plant choice is not statistically independent of expected ESs from riparian buffers)

	Eucalyptus spp			Grevilia robusta			Arundinaria alpine			Pennisetum purpureum			Calliandra calothyrus		
	χ2	df	р-	χ2	df	р-	χ2	df	р-	χ2	df	р-	χ2	df	р-
Food	11.525	1	0.001	.036	1	.850	2.262	1	.133	.180	1	.672	.921	1	.337
Fuel wood	19.147	1	<0.001	.563	1	.453	1.107	1	.293	6.618	1	.010	.128	1	.721
Building material	8.086	1	0.004	.259	1	.611	2.585	1	.108	.013	1	.908	.223	1	.637
Medicinal plants	8.325	1	0.004	1.713	1	.191	.566	1	.452	1.165	1	.280	.368	1	.544
Flood buffering	1.478	1	.224	2.734	1	.098	4.351	1	.037	.086	1	.770	.494	1	.482
Poles to support beans	12.738	1	< 0.001	.559	1	.455	5.180	1	.023	2.730	1	.098	.566	1	.452
Less fertilizer use	.064	1	0.800	.614	1	.433	2.941	1	.086	.030	1	.863	.455	1	.500
Hay for livestock	10.429	1	0.001	5.620	1	.018	.184	1	.668	2.876	1	.090	.168	1	.683
Crops washed away	4.631	1	0.031	2.120	1	.145	.624	1	.430	1.288	1	.256	.005	1	.945
Water for domestic use	1.389	1	0.239	.041	1	.840	3.897	1	.048	1.276	1	.168	3.268	1	.071
Farm size	5.783	1	0.16	.117	1	.733	2,074	1	150	.000*	1	.983	2.858	1	.091
Income level	34.270	1	< 0.001	.052	1	.819	2.035	1	.154	.033	1	.856	2.292	1	.130

Challenges to the maintenance of ESs.

Smallholder farmers reported a number of challenges that affect the adoption of activities that support ESs maintenance. Among these, the vast majority of smallholder farmers indicated the following: lack of financial assistance, lack of equipment, lack of information about needed or useful technologies, and insufficient arable land. A vast majority of smallholder farmers (73%)

reported that they get information about best agricultural practices primarily from extension services. Farmer-to-farmer extension was mentioned by about 12% of respondents, while farmers' own experimentation as source of information was mentioned by 11%.

Discussion

Preferred ESs

Farmers in the Upper Mukungwa watershed understand the importance of ESs but their understanding is virtually restricted to provisional ESs that include the provision of food primarily, then fuel wood, building material, hay/fodder for livestock, sticks to support climbing beans, and medicinal plants. This study suggests that smallholder farmers in the study area prefer to grow food crops within riparian buffer zones if they were not required by the government to plant trees and shrubs instead and they also report that the same food crops are among the most vulnerable to erosion. It seems that establishing riparian buffers that provide some regulating ESs is not one of their primary goals despite the fact that smallholder farmers in this upland region are in a region prone to heavy rains and flood events, a fact that is also recognized by smallholder farmers' reported vulnerability of the most commonly grown crops, beans and corn, to the effects of climate change. Previous studies conducted in Rwanda have also found that farmers value food over other ESs (Nabahungu & Visser, 2013b).

The Government of Rwanda's new agricultural policy involving mono-cropping systems coupled with intensive use of synthetic fertilizers and pesticides seems to align with farmers' indication that food and money from selling those food crops are most important to them; however these approaches are likely to compromise the fertility of the soil, pollination services and water

quality (Isaacs et al., 2016). Agricultural ecosystems are managed by people chiefly to meet food, fiber and fuel needs especially in places where those needs are most pressing, the case in many rural regions of Sub-Saharan Africa (Garrity et al., 2010; Leach & Mearns, 2013), including Rwanda (Clay et al., 1995; Branca et al., 2011). However, agricultural ecosystems rely on a suite of supporting ESs to provide food, fiber and fuel (Swinton et al., 2007a).

The awareness and support for these services that support agricultural activities has been called "Ecological intensification" which entails the environmentally friendly replacement of anthropogenic inputs and/or enhancement of crop yield, by including regulating and supporting ESs management in agricultural practices (Bommarco et al., 2013; Tittonell, 2014). Put in the context of the present study, farmers' understanding of ESs through the lens of food production and other functions that support food production such as the provision of fuel wood for cooking, fodder for animals and sticks for climbing beans, is an indication that meeting household nutritional needs by selling the crops they grow are of primary priority. However, concerns about issues related to soil fertility restoration and water quality, which require a long-term perspective, do not seem to be part of the farmers' vision. Nevertheless, there is an opportunity for agricultural policy to bridge farmer's goals with long-term regulatory ES restoration and maintenance by providing incentives. Growing riparian trees to the maturity stage might take about ten years, a period corresponding to about 20 crop seasons. During the tree growing period, the yield of crops farmers normally grow in those riparian zones will certainly decline, thus cutting their net yearly income from the riparian farm plots. To compensate for this deficit, the government could provide incentives to offset the gap until trees have grown to a stage in which farmers experience their benefits and may not need any more incentives.

Several studies show that farmers prioritize food as the primary provisioning ESs and understand the interconnectedness between food provisioning and several supporting services such as soil conservation, nutrient cycling, water availability, soil erosion control and flood regulation, which can affect crop yields (e.g. Logsdon et al., 2015; Lamarque et al., 2011; Smith & Sullivan, 2014; Vignola et al., 2010). However, farmers in the study area did not seem to fully understand this interconnectedness because, for example, they did not report, among other concerns, the impact of current agricultural practices or the establishment of riparian buffers on other landscapes downstream. However, research has shown that crops in individual fields are dependent on services provided by nearby ecosystems, whether native or managed, and nearby ecosystems are often influenced by their agricultural neighbors (Swinton et al., 2007a; Asbjornsen et al., 2014; Sepp, 2012). Neighboring ecosystems provide food, refugia, and reproductive habitat for pollinators and biocontrol agents; they provide wildlife habitat, and they help to attenuate some of the unwelcome effects of agricultural production, including the escape of nitrogen, phosphorus, and pesticides into non-agricultural ecosystems where they may produce undesirable impacts (Swinton et al., 2007a). Unfortunately, current agricultural systems in many parts of Africa, including Rwanda, have been, since the last century, based primarily on shortterm economic rather than ecological concerns (Salam et al., 2000) often resulting in the degradation rather than maintenance of regulatory ESs (Poppy et al., 2014; Thiaw et al., 2011; Godfray et al., 2010). As a result, degradation of such regulating ESs, which include the control of erosion, floods and nutrient cycling, for example have inflicted a decline in food crop production over the past few years (Godfray et al., 2010; Fisher et al., 2013) and this impact is exacerbated by climate change (Palmer et al., 2008; Zhao et al., 2005).

In Rwanda soil erosion causes a total soil loss of about 15 million metric tons per year (Rwandan Ministry of Agriculture and Animal Resources, 2009), equivalent to losing the capacity to feed 40,000 people annually. A high reliance on scarce farmland, as is the case for Rwanda where about 57% of farms are less than 0.5 ha (Singh, 2000) with an average farm size of 0.76 ha (Booth & Golooba-Mutebi, 2012), explains why farmers prioritize food production over other ESs that require the planting of trees for example. The majority of Rwandan farmers cannot afford to devote land to trees or pasture (Malyon, 2014).

Determinants of smallholder farmers' choice of crops: environmental concerns vs. economic profitability.

In this study, smallholder farmers of the Upper Mukungwa Watershed reported that the cultivation of corn, beans and Irish potatoes has increased significantly over the past few years, compared to other food crops. This increase may be attributed to the recent agricultural intensification policy put in place by the Rwandan Government in 2008 to meet food security objectives. Farmers however reported that beans, corn and potatoes are the most susceptible to erosion compared to other crops.

The most prominent factor that drove farmers' crop choice in this study was economic profitability, followed by "nutritional value" and "government recommendations". The reason why these crops are preferred for economic reasons despite the fact that they are vulnerable to heavy rains is answered in other studies (Jayne & Jones, 1997; Smale et al., 2013). Economic profitability has been shown to be a major driver underpinning farmers' choice of farming

systems and crop choice (Chouinard et al., 2008; Musshoff & Hirschauer, 2008; Cary & Wilkinson, 1997). Economic profitability includes maximization of gross margin, and minimization of working capital, hired labor, management difficulties, and risk (Sheeder & Lynne, 2011). However, other studies of farmers' decision making processes show that the assumption of "economic profitability" is not the sole driver to adopt agricultural practices. This explanation ignores the complexity of human behavior (Karali et al., 2011). Humans do not solely aim at profit maximization (Hansen, Marx, & Weber, 2004). Instead, humans tend to follow different decision-making pathways and make sub-optimal choices based on a complex web of factors, such as economic, demographic, social, cultural, psychological, technological, biophysical and environmental issues (Karali et al., 2011; Vignola et al., 2010; Keshavarz & Karami, 2014).

The Theory of Planned Behavior (TPB) brings together the economic aspect and other sociodemographic and political aspects to explain how farmer choices are influenced by multi-layered and interconnected factors. Based on the theory, adoption of any agricultural practice hinges on past behavior, attitudes, and perceived behavioral control; these factors can be significant predictors of intentions, and intentions significantly predict self-reported behavior (Zubair & Garforth, 2006). Group norms and intergroup perceptions can also be significant predictors of intentions providing support for the inclusion of social identity concepts in the theory of planned behavior (Ellis-Iversen et al., 2010; Greiner et al., 2009). My study reveals that, besides economic profitability, policy recommendations also play a pivotal role in influencing farmer decisions, which is included as an influencing factor of behavior. A study conducted in Australia (Greiner & Gregg, 2011) where nation-wide conservation programs attracted criticism for low

levels of effectiveness and efficiency because farmers' perceptions and characteristics were not considered.

Choice of multifunctional riparian buffer plants.

Smallholder farmers in this study preferred to plant *Eucalyptus* spp. within riparian zones mainly because it meets multiple needs which include the provision of medicine, building material, fuelwood, sticks to support climbing beans, and shade where grass can grow for livestock. Smallholder farmers also seem to support the role of *Eucalyptus* in protecting soil against erosion. However, my results show that the choice of this tree species is associated with relatively higher levels of income. Since Eucalyptus is generally planted in degraded and marginal lands in woodlots (MINAGRI, 1991), one could infer that farmers with relatively larger farm size or higher income are more likely to adopt woodlots on their farms than poor smallholder farmers (Ndayambaje et al., 2013). While the adoption of *Eucalyptus* trees for riparian zones is not a new practice, it is contingent upon access to information and financial resources to make it viable. Studies show that farmers with higher household income are more likely to prefer Eucalyptus for riparian restoration, given greater access to information and financial resources as is the case for the adoption of any other farming practices (Franzel, 1999; Knowler and Bradshaw, 2007; Croppenstedt et al., 2003; Nhemachena and Hassan, 2007; Gandure et al., 2013; Deressa et al., 2011) given risks and costs involved over which poor uninformed smallholder farmers have little control. In this study, farmers demonstrated willingness to establish riparian buffers within their farmland proximal to streams. The types of plants they proposed to plant in the riparian zones are those which they consider to fulfill a multiplicity of functions ranging from poles to support climbing beans, fodder for livestock, fuelwood, building material, medicinal plants and flood control. Besides Eucalyptus spp.,

farmers also prefer *Grevilea robusta*, reportedly for fodder for livestock, and *Arundinaria alpina* (bamboo), because it provides sticks to support climbing beans and also contributes to flood buffering.

Eucalyptus spp. was the most mentioned tree for riparian buffer zone planting presumably because it is the most commonly used material for supporting climbing beans given its durability, but also because it is an economically high-energy fuel wood, in addition to its use for medicinal purposes. About 83% of the Rwandan population relies on wood energy for cooking (Ndayambaje & Mohren, 2011); the demand for wood resources exceeds the supply in Rwanda and the situation will continue to deteriorate (Safari, 2010). In addition, Eucalyptus spp. is a fastgrowing, deeply-rooted tree (Robinson et al., 2006) efficient in holding soil. However, it removes moisture from soil to depths of at least 8-10 m and reduces water recharge, which can reduce baseflow in streams (Robinson et al., 2006; Bouillet et al., 2002). While it appears that farmers have their own preferences when it comes to trees species to plant within riparian zones, the Rwandan agricultural policy could encourage the planting of multipurpose trees that meet farmers'various needs and which help protect stream banks, control riparian zone erosion and nutrient leaching into streams at the same time. Farmers' understanding of economic opportunities and consequences of supporting clearly defined ecosystems services appears to influence their willingness and decision to customize their agricultural practices, by establish functioning riparian buffers for example, in order to maintain those ESs (Bommarco et al., 2013).

Perception of ESs and experienced environmental degradation.

I found an association between farmers' perception of the importance of riparian zones and their perceived experience with environment problems in and around streams, including soil erosion and fluvial floods. Smallholder farmers reported that erosion of stream banks and farmlands adjacent to streams significantly affect their households. Consequently, farmers support the establishment of riparian buffers that include the planting of trees in combination with other shrubs and grass. A study conducted in Ethiopia on the assessment of soil erosion and farmer perception of land conservation (Zegeve et al., 2010) revealed that the major cause of soil erosion mentioned by 98% of smallholder farmers was the lack of conservation structures. A study on risk perception of climate change among farmers showed a relationship between perception of increasing degradation of farmland soil and the importance of regulation services (Vignola et al., 2010). However, as evidenced by previous studies, the awareness of the importance of restoring and maintaining ESs can be overshadowed by economic interests. For example, recent increase in the use of synthetic fertilizers within the framework of a Rwandan government-mandated crop intensification program launched in 2007 has resulted in an increase in crop production, which in turn overshadows implementation of measures to control soil fertility loss in some cases.

The Rwandan Vision 2020 was launched in 2000 with a focus on the transformation of agriculture into a productive, high value, market oriented sector (Kaberuka et al., 2000; Pritchard, 2013). Started in September 2007, the Crop Intensification Program (CIP) focuses on six priority crops, maize, wheat, rice, Irish potato, beans and cassava. As a result of this policy, crop yield has increased. The production of maize and wheat has increased by 6-fold, and that of

Irish potato and cassava has tripled. The production of rice and beans has increased by 30% in the past four years, and the land area under cultivation of those crops was projected to increase (MINAGRI, 2011). Since the implementation of CIP, the cultivation of maize, beans and Irish potatoes has increased (MINAGRI, 2011) (Figure 4-8). Areas of cultivation have extended to some wetlands which have been cultivated under the government plan with the aim of growing rice as the main crop (Nabahungu & Visser, 2013b) which, over the last few years has also added to crop net production (Figure 4-9). In the face of such short-term alluring profits, it is hard for smallholder farmers who are equipped with limited information to foresee long term impacts of the CIP on water quality and availability as well as on soil fertility, just to mention a few.



Figure 4-8. Recent trends in consolidation of land use areas under cultivation of priority crops in Season A (from September to February) in Rwanda. Source: MINAGRI (2011)



Figure 4-9. Changes in on-farm yields of major crops in response to the use of distributed inputs (improved seeds/planting materials and/or fertilizers) in Rwanda. Source: MINAGRI (2011)

In year 2016, Rwandan agricultural smallholder farmers used inorganic fertilizers mostly on Irish potatoes (73%), NPK being the most used. In the northern highlands of Rwanda, the proportion of farmers using pesticides is the highest in the country (68%), and country wide dithane was the most highly used pesticide by farmers. It is used on Irish potatoes, climbing beans and tomatoes as well as other crops (NISR, 2016). While efforts have been deployed to shift from subsistence farming to commercial farming characterized by intensive use of inorganic fertilizers and pesticides within a monocropping system, studies have shown that increased crop yields get rapidly counterbalanced by negative environmental impacts such as soil erosion, groundwater pollution, river eutrophication, excessive water use, and the development of weeds and diseases resistant to chemical control (Lichtfouse et al., 2009; Foley et al., 2011). For example, intensive agriculture in Rwanda contributes to the pollution of the Akagera River (Wali et al., 2011) and the eutrophication of Lake Victoria (Kimwaga et al., 2012).

Studies have shown that riparian buffer strips are most effective at retaining sediments and reducing the transfer of phosphorus into streams, thus reducing non-point pollution generated from intensive agriculture (Cuttle et al., 2007). However, for smallholder farmers upstream to become aware of the negative impacts of their farming practices that include increased use of synthetic fertilizers and pesticides to human populations downstream, requires the intervention of scientist working collaboratively with farmers, local NGOs involved in agriculture and policy makers. A policy change in agriculture practices to define strategies to engage scientists to work collaboratively with smallholder farmers in monitoring the impact of their agricultural practices to human populations downstream is needed to ensure sustainable management of riparian buffers upstream.

Smallholder farmers' willingness to maintain ecosystem services (ES).

A vast majority of smallholder farmers interviewed within the study area reported a willingness to establish riparian buffers that restore and maintain ESs services within riparian zones as long as the practice provide profits equal or greater than those they normally get from the same plot if planted with food crops. Their willingness to maintain ESs was significantly associated with their preference for particular ES which include building material, medicinal plants, flood buffering and poles to support climbing beans. However, willingness to adopt riparian buffers was not significantly associated with level of income or farm size which might suggest that the need for ESs to be generated by riparian buffers is felt by all smallholder farmers regardless of their level of income. While some literature on riparian buffers focuses primarily on ecological benefits of functional riparian buffers for biodiversity conservation (Olson et al., 2007; Young-Mathews et al., 2010; Palmer & Bennett, 2006), water quality (Miserendino et al., 2011; Li et al.,

2009; Sahu & Gu, 2009; Dosskey et al., 2010) and pollination (Cole et al., 2015; Wratten et al., 2012; Gray & Lewis, 2014; Smukler et al., 2012), other studies include the role of riparian buffers in providing supporting services such as fuel, pest control and maintenance of soil fertility (Swinton et al., 2007b; Zhang et al, , 2010). Recognition of benefits of riparian buffers to local farmers could fuel their motivation to establish those buffers in their farmland proximal to streams in order to maintain ESs they generate.

The Theory of Planned Behavior (TPB) provides a framework to understand how beliefs influence decisions about riparian zone management. Strong intentions to manage riparian zones can be associated with a favorable cost-benefit analysis that farmers are aware of, greater perceptions of normative support for the practice and perceptions of the extent to which barriers would impede management of riparian zones (Fielding et al., 2005). Perceptions of costs and benefits of riparian zone management are considered the most important factors influencing farmers' intentions to manage riparian zones (Fielding et al., 2005). Farmers with low income have a reduced capacity to grow trees on their scarce farmland; wealthier farmers tend to have more land and the financial capability to augment their energy needs by growing firewood in woodlots or purchasing firewood or charcoal (Nabanoga, 2005). The integration of riparian buffers can be acceptable to poor smallholder farmers if they are assured of realizing a timely? multiplicity of benefits (Mugonola et al., 2013; Lovell et al., 2010) that are direct and which outweigh the income they normally get from the same area if grown with food crops such as corn, beans or potatoes.
Determinants of plant choice for riparian zones.

Farmers choose plant species for riparian zone protection based on the ESs they expect from such choices. Other food crop plant species such as potatoes, beans and corn are based upon financial returns. Farmers' preference for Eucalyptus spp., Grevillea robusta and Arundinaria *alpine* are driven by the high economic value these species have and the ESs they can offer, including the provision of fuel wood, poles for climbing beans, fodder and other wood products. When farmers decide to establish or maintain a riparian buffer on their farmland, they have preferences about which plant species to use based on their goals (Lynch & Brown, 2000). Tree planting choices by farmers is believed to be determined by direct use of those tree species in their everyday life, mostly to support their farming and other household needs such as hay/fodder for livestock (Scherr, 1995; Assogbadjo et al., 2012; Roothaert & Franzel, 2001), food (Assogbadjo et al., 2012), medicines, stakes to support beans, and building material (Degrande et al., 2013). Tree planting on farmlands is also generally determined by farmers' income. Previous studies have shown that high income landowners tend to grow trees on their farmland more than smallholder farmers (Salam et al., 2000; Lovell et al., 2010; Temesgen et al., 2009). Furthermore, farmers' choice of trees could be limited by not being aware of other species they could plant. Therefore, they will tend to choose species with which they are familiar.

A vast majority of farmers in this study preferred particularly *Eucalyptus* spp. for riparian buffer planting. In Rwanda, the demand for fuel wood is high (Atangana, Khasa et al., 2014); 85% of the Rwandan population uses firewood and 0.6% use charcoal to meet their energy demands (MINITERE, 2004). Despite efforts to establish trees in agricultural fields as part of agroforestry systems, the gap between the supply and demand for fuel wood in Rwanda is increasing

(Ndayambaje & Mohren, 2011). I found a positive association between the preference for *Eucalyptus* spp. for riparian buffers and the need for fuel wood as one of the most preferred ESs from riparian buffers. *Eucalyptus* spp. is known for its fast growth, high calorific value and coppicing ability, which increases the stock density and profitability per unit area (Bagchi & Mittal, 1996; Babitha et al., 2000; Turnbull et al., 2000; Little & Gardner, 2003). Besides the provision of energy for cooking and heating, *Eucalyptus* spp. provides high-value timber for construction and making furniture and stakes for climbing beans. *Eucalyptus* spp. is the most commonly planted tree species in Rwanda (Ndayambaje et al., 2012).

Findings from this study reveal that the preference for *Eucalyptus spp*. was associated with the level of income, corroborated by a recent study by Ndayambaje et al. (2012) which found that *Eucalyptus* spp. is mostly grown by farmers with relatively higher income or larger farm size. *Grevilea robusta* was also highly ranked as a tree species of choice by farmers, and was associated with farmers' reported need for hay for livestock. In Rwanda it is a fast-growing plant with minimal competitiveness with crops (Nath et al., 2011; Bucagu et al., 2013) and with proper management, the species can be grown in plantations and on farms for firewood, poles and timber (Kalinganire, 1996; Bucagu et al., 2013). The choice of bamboo (*Arundinaria alpina*), a fast growing herbaceous species that can contribute to soil erosion control and restoration of soil fertility (Andrew & Masozera, 2010; Embaye et al., 2005) was associated with the need for poles to support climbing beans, the need to buffer the effect of fluvial floods and to improve water quality in streams by entrapping nutrients generated from agriculture. In addition, *Arundinaria alpina* does not need much management effort once it is established. It grows fast and can

supplement smallholder farmers' income. However, its spikes for climbing beans are less durable than those of *Grevilea robusta* or *Eucalyptus* spp.

Theory of Planned Behavior and perception of ESs.

Failure to gauge farmers' attitudes and motivation for adopting certain farming practices can impair our understanding of adaptive behavioral choices and effective adoption of best agricultural practices (Frank et al., 2011). Recent research indicates that incorporating insights from motivation theory can enhance theorization of adaptive capacity (Frank et al., 2011). This study demonstrates that the main motivation that supports farmers' positive attitude towards the establishment of riparian zones stems both from their past experience of land loss through erosion and their need for timber and non-timber products to support food crop production and their livelihoods. The TPB has been examined with regards to the adoption of agroforestry and it has been found that socio-economic factors in association with self-efficacy, i.e. confidence in the ability to adopt agroforestry and perceived control of the adoption process, can significantly influence willingness to maintain ESs through the adoption of riparian buffers hinges on both farmer socio-economic status, including income, farm size, and self-perceived capacity, also termed self-efficacy, to perform tasks related to the establishment of the riparian buffer.

The benefits identified in this study were mostly provisional ESs. Less regard was given to regulating ESs such as the role of riparian buffers in entrapping nutrients from agriculture landscapes, hence contributing to water quality. While the literature supports potential water quality benefits of riparian buffer zone adoption (Swinton et al., 2007b; Zhang et al., 2010), results from this study show that farmers did not indicate any connection between current

agricultural practices and quality and quantity of water in the Mukungwa river. This is an externality that the Rwandan government policy makers need to consider by using for example a payment for ecosystem services framework whereby communities downstream can contribute to the payment of best landscape management upstream. A multifunctional riparian buffer zone that meets farmers' goals and address environmental needs at the same time may therefore be an option in this landscape. The success in establishing a functional riparian buffer that contributes to improving the quality and the quantity of water in streams is contingent on addressing financial and technological challenges identified by farmers.

Any policy geared toward changing smallholder farmers' behavior in terms of agricultural practices, to be effective, needs to recognize the importance of smallholder farmers' perceived overall goals. These goals include economic profitability, satisfaction of family nutritional needs and compliance with the national agrarian policy as key drivers of behavior change for the adoption of practices that promote ESs such as the establishment and management of riparian buffers. Smallholder farmers' perceptions to support the provision of ESs through the establishment of functional riparian buffer zones could lead to positive behavioral outcomes if factors such as cost of provision, economic incentives, landowner preferences (Buckley, Hynes, & Mechan, 2012; Golkowska et al., 2016) and access to information are taken into account.

Access to information can come through extension services (Franz et al., 2010) or farmer-tofarmer extension, which, coupled with government-led extension, is considered to be effective given its contagious nature and trust among neighboring farmers (Kiptot et al., 2006; Davis et al., 2012). However, government-led extension services are limited in their reach and efficacy, and

often do not reach poor, marginalized farmers including women, minorities, and people in remote areas (Davis et al., 2012; Anandajayasekeram et al., 2007). It is important to note that extension services often bring policy guidelines to farmers without discussing with them practical modalities of their implementation on the ground and without exploring realistic options (Scarborough et al., 1997; Akinnagbe et al., 2010; Deressa et al., 2011; Islam et al., 2011). This might explain why a vast majority of farmers in the Upper Mukungwa Watershed claim they get information about agricultural practices primarily from extension services, yet they also complain that they lack access to technology and technological information about establishment of economically and environmentally profitable riparian buffers. Access to credit or loans is limited among smallholder farmers in sub-Saharan Africa. Micro-credits schemes in Rwanda, for example, are widespread and are believed to contribute to the efficiency of agricultural production.

Provision of ESs within agricultural landscapes is dependent on farmers' goals, choices and awareness of environmental issues related to fields adjacent to streams, and is closely linked to experienced damage inflicted on food production by flood events. Farmers in the Upper Mukungwa Watershed of the Musanze District favor ESs that support food production or help meet priority domestic needs such as the provision of fuel wood or building material. In the face of scarce land resources, farmers appear to expect the most of trees planted on a small portion of their farmland devoted to a riparian buffer to meet their various timber and non-timber related needs. Agricultural policy should consider a farmer-centered maximization of riparian buffer performance that accommodates both farmers' goals in planting trees while helping increase soil stability and fertility. This strategy could, if well managed, also indirectly benefit other

regulatory ESs such as soil nutrients, soil conservation and stream bank stability. To achieve this however, technological information about how to establish riparian buffer zones in a more economically profitable and less agro-ecologically risky way needs to be infused into farmers' practices through a more efficient, farmer-centered, collaborative approach to extension services. Agrarian policy-making should consider engaging farmers in the riparian buffer planning process for effective adoption of jointly agreed upon plans to meet provision and regulatory requirements sustainably. Given evidence from farmers' perceptions about their strong inclination for provisional ESs, educational efforts should be deployed among farmers to promote synergies between their agricultural practices within riparian zones and regulating ESs maintenance. Further research to quantify and value regulating ESs provided by functional riparian buffer zones should be encouraged and smallholder farmers should be kept informed about outcomes through extension services.

References

- Ajzen, I. (1991). The theory of planned behaviour. Organizational Behaviour and Human Decision Processes, 50, 179-211.
- Ajzen, I. (2002). Perceived behavioral control, self-efficacy, locus of control, and the theory of planned behavior. *Journal of Applied Social Psychology*, *32*(4), 665–683.
- Akinnagbe, O. M., Ajayi, A. R., & others. (2010). Challenges of farmer-led extension approaches in Nigeria. *World Journal of Agricultural Sciences*, 6(4), 353–359.
- Ali, D. A., Deininger, K., & Duponchel, M. (2014). Credit Constraints and Agricultural Productivity: Evidence from rural Rwanda. *Journal of Development Studies*, 50(5), 649– 665.
- Anandajayasekeram, P., Davis, K. E., & Workneh, S. (2007). Farmer field schools: an alternative to existing extension systems? Experience from Eastern and Southern Africa. *Journal of International Agricultural and Extension Education*, 14(1), 81–93.
- Andrew, G., & Masozera, M. (2010). Payment for ecosystem services and poverty reduction in Rwanda. *J Sustain Develop Afr*, *12*, 122–139.
- Armitage, C. J., & Christian, J. (2004). *Planned behavior: the relationship between human thought and action*. New Brunswick, N.J.: Transaction Publishers.
- Asbjornsen, H., Hernandez-Santana, V., Liebman, M., Bayala, J., Chen, J., Helmers, M., ... Schulte, L. A. (2014). Targeting perennial vegetation in agricultural landscapes for enhancing ecosystem services. *Renewable Agriculture and Food Systems*, 29(2), 101– 125.
- Assogbadjo, A. E., Glèlè Kakaï, R., Vodouhê, F. G., Djagoun, C. A. M. S., Codjia, J. T. C., & Sinsin, B. (2012). Biodiversity and socioeconomic factors supporting farmers' choice of

wild edible trees in the agroforestry systems of Benin (West Africa). *Forest Policy and Economics*, *14*(1), 41–49.

- Atangana, A., Khasa, D., Chang, S., & Degrande, A. (2014). Major Agroforestry Systems of the Humid Tropics. In *Tropical Agroforestry* (pp. 49–93). Springer Netherlands. Retrieved from http://link.springer.com.antioch.idm.oclc.org/chapter/10.1007/978-94-007-7723-1_4
- Babitha, M., Sreenivasulu, A., Meru, E., Rao, P. S., & others. (2000). Regenerated shoot number and coppicing ability of two year old Eucalyptus tereticornis stumps of different girth classes. *Indian Forester*, 126(7), 721–726.
- Bagchi, S. K., & Mittal, M. C. (1996). Regenerated shoot number (coppicing ability) after pruning at different height levels in one year old Eucalyptus (Mysore-gum). *Indian Forester*, 122(8), 731–733.
- Bandura, A. (1977). Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review*, 84(2), 191–215.
- Bommarco, R., Kleijn, D., & Potts, S. G. (2013). Ecological intensification: harnessing ecosystem services for food security. *Trends in Ecology & Evolution*, *28*(4), 230–238.
- Booth, D., & Golooba-Mutebi, F. (2012). Policy for agriculture and horticulture in Rwanda: a different political economy? Retrieved from http://opendocs.ids.ac.uk/opendocs/handle/123456789/2250
- Bouillet, J.-P., Laclau, J.-P., Arnaud, M., M'Bou, A. T., Saint-André, L., & Jourdan, C. (2002).
 Changes with age in the spatial distribution of roots of Eucalyptus clone in Congo:
 Impact on water and nutrient uptake. *Forest Ecology and Management*, 171(1–2), 43–57.

- Branca, G., McCarthy, N., Lipper, L., & Jolejole, M. C. (2011). Climate-smart agriculture: a synthesis of empirical evidence of food security and mitigation benefits from improved cropland management. *Mitigation of Climate Change in Agriculture Series*, 3, 1–42.
- Bryan, E., Ringler, C., Okoba, B., Roncoli, C., Silvestri, S., & Herrero, M. (2013). Adapting agriculture to climate change in Kenya: Household strategies and determinants. *Journal* of Environmental Management, 114, 26–35.
- Bucagu, C., Vanlauwe, B., Wijk, M. T., & Giller, K. E. (2013). Assessing farmers' interest in agroforestry in two contrasting agro-ecological zones of Rwanda. *Agroforestry Systems*, 87(1), 141–158.
- Buckley, C., Hynes, S., & Mechan, S. (2012). Supply of an ecosystem service—Farmers' willingness to adopt riparian buffer zones in agricultural catchments. *Environmental Science & Policy*, 24, 101–109.
- Cantore, N. (2011). The crop intensification program in Rwanda: A sustainability analysis. Retrieved from http://dspace.cigilibrary.org/jspui/handle/123456789/31634
- Cary, J. W., & Wilkinson, R. L. (1997). Perceived profitability and farmers 'conservation behaviour. *Journal of Agricultural Economics*, *48*(1–3), 13–21.
- Case, P., Sparks, P., & Pavey, L. (2016). Identity appropriateness and the structure of the theory of planned behaviour. *British Journal of Social Psychology*, *55*(1), 109–125.
- Chouinard, H. H., Paterson, T., Wandschneider, P. R., & Ohler, A. M. (2008). Will Farmers Trade Profits for Stewardship?: Heterogeneous Motivations for Farm Practice Selection. *Land Economics*, 4(1), 66–82.
- Clay, D. C., Byiringiro, F. U., Kangasniemi, J., Reardon, T., Sibomana, B., Uwamariya, L., ... others. (1995). *Promoting food security in Rwanda through sustainable agricultural*

productivity: Meeting the challenges of population pressure, land degradation, and poverty. Michigan State University, Department of Agricultural, Food, and Resource Economics. Retrieved from https://ideas.repec.org/p/ags/mididp/54054.html

- Cole, L. J., Brocklehurst, S., Robertson, D., Harrison, W., & McCracken, D. I. (2015). Riparian buffer strips: their role in the conservation of insect pollinators in intensive grassland systems. *Agriculture, Ecosystems & Environment*, 211, 207–220.
- Costanza, R., d'Arge, R., De Groot, R., Farber, S., Grasso, M., Hannon, B., ... others. (1997). The value of the world's ecosystem services and natural capital. *Nature*, *387*(6630), 253–260.
- Cruz-Garcia, G. S., Sachet, E., Vanegas, M., & Piispanen, K. (2016). Are the major imperatives of food security missing in ecosystem services research? *Ecosystem Services*, *19*, 19–31.
- Cuttle, S. P., Macleod, C. J. A., Chadwick, D. R., Scholefield, D., Haygarth, P. M., Newell-Price,
 P., ... Humphrey, R. (2007). An inventory of methods to control diffuse water pollution
 from agriculture (DWPA). *Defra ES0203, London*. Retrieved from
 http://www.cost869.alterra.nl/UK_Manual.pdf
- Davis, K., Nkonya, E., Kato, E., Mekonnen, D. A., Odendo, M., Miiro, R., & Nkuba, J. (2012).
 Impact of Farmer Field Schools on Agricultural Productivity and Poverty in East Africa.
 World Development, 40(2), 402–413.
- de Groot, R. S., Alkemade, R., Braat, L., Hein, L., & Willemen, L. (2010). Challenges in integrating the concept of ecosystem services and values in landscape planning, management and decision making. *Ecological Complexity*, 7(3), 260–272.

- Degrande, A., Tadjo, P., Takoutsing, B., Asaah, E., Tsobeng, A., & Tchoundjeu, Z. (2013).
 Getting Trees Into Farmers' Fields: Success of Rural Nurseries in Distributing High
 Quality Planting Material in Cameroon. *Small-Scale Forestry*, *12*(3), 403–420.
- Den Herder, G., Van Isterdael, G., Beeckman, T., & De Smet, I. (2010). The roots of a new green revolution. *Trends in Plant Science*, *15*(11), 600–607.
- Deressa, T. T., Hassan, R. M., & Ringler, C. (2011). Perception of and adaptation to climate change by farmers in the Nile basin of Ethiopia. *The Journal of Agricultural Science*, 149(01), 23–31.
- Deressa, T. T., Hassan, R. M., Ringler, C., Alemu, T., & Yesuf, M. (2009). Determinants of farmers' choice of adaptation methods to climate change in the Nile Basin of Ethiopia. *Global Environmental Change*, 19(2), 248–255.
- Doré, T., Makowski, D., Malézieux, E., Munier-Jolain, N., Tchamitchian, M., & Tittonell, P.
 (2011). Facing up to the paradigm of ecological intensification in agronomy: revisiting methods, concepts and knowledge. *European Journal of Agronomy*, 34(4), 197–210.
- Dosskey, M. G., Vidon, P., Gurwick, N. P., Allan, C. J., Duval, T. P., & Lowrance, R. (2010). *The role of riparian vegetation in protecting and improving chemical water quality in streams1*. Wiley Online Library. Retrieved from http://onlinelibrary.wiley.com/doi/10.1111/j.1752-1688.2010.00419.x/full
- Ellis-Iversen, J., Cook, A. J., Watson, E., Nielen, M., Larkin, L., Wooldridge, M., & Hogeveen,
 H. (2010). Perceptions, circumstances and motivators that influence implementation of
 zoonotic control programs on cattle farms. *Preventive Veterinary Medicine*, 93(4), 276–285.

- Embaye, K., Weih, M., Ledin, S., & Christersson, L. (2005). Biomass and nutrient distribution in a highland bamboo forest in southwest Ethiopia: implications for management. *Forest Ecology and Management*, 204(2), 159–169.
- Etchevers, J. D., Prat, C., Balbontín, C., Bravo, M., & Martínez, M. (2009). Influence of Land Use on Carbon Sequestration and Erosion in Mexico: A Review. In E. Lichtfouse, M. Navarrete, P. Debaeke, S. Véronique, & C. Alberola (Eds.), *Sustainable Agriculture* (pp. 87–96). Springer Netherlands. Retrieved from http://link.springer.com.antioch.idm.oclc.org/chapter/10.1007/978-90-481-2666-8_8
- Fielding, K. S., McDonald, R., & Louis, W. R. (2008). Theory of planned behaviour, identity and intentions to engage in environmental activism. *Journal of Environmental Psychology*, 28(4), 318–326.
- Fielding, K. S., Terry, D. J., Masser, B. M., Bordia, P., & Hogg, M. A. (2005). Explaining landholders' decisions about riparian zone management: The role of behavioural, normative, and control beliefs. *Journal of Environmental Management*, 77(1), 12–21.
- Fisher, J. A., Patenaude, G., Meir, P., Nightingale, A. J., Rounsevell, M. D., Williams, M., & Woodhouse, I. H. (2013). Strengthening conceptual foundations: analysing frameworks for ecosystem services and poverty alleviation research. *Global Environmental Change*, 23(5), 1098–1111.
- Foley, J. A., Ramankutty, N., Brauman, K. A., Cassidy, E. S., Gerber, J. S., Johnston, M., ... others. (2011). Solutions for a cultivated planet. *Nature*, 478(7369), 337–342.
- Frank, E., Eakin, H., & López-Carr, D. (2011). Social identity, perception and motivation in adaptation to climate risk in the coffee sector of Chiapas, Mexico. *Global Environmental Change*, 21(1), 66–76.

- Franz, N. K., Piercy, F., Donaldson, J., Westbrook, J., & Richard, R. (2010). Farmer, agent, and specialist perspectives on preferences for learning among today's farmers. *Journal of Extension*, 48(3), 3RIB1.
- Gandure, S., Walker, S., & Botha, J. J. (2013). Farmers' perceptions of adaptation to climate change and water stress in a South African rural community. *Environmental Development*, 5, 39–53.
- Garrity, D. P., Akinnifesi, F. K., Ajayi, O. C., Weldesemayat, S. G., Mowo, J. G., Kalinganire,
 A., ... Bayala, J. (2010). Evergreen Agriculture: a robust approach to sustainable food security in Africa. *Food Security*, 2(3), 197–214.
- Godfray, H. C. J., Beddington, J. R., Crute, I. R., Haddad, L., Lawrence, D., Muir, J. F., ...
 Toulmin, C. (2010). Food Security: The Challenge of Feeding 9 Billion People. *Science*, 327(5967), 812–818. https://doi.org/10.1126/science.1185383
- Godin, G., & Kok, G. (1996). The theory of planned behavior: a review of its applications to health-related behaviors. *American Journal of Health Promotion*, *11*(2), 87–98.
- Goldman, A., Hartter, J., Southworth, J., & Binford, M. W. (2008). The human landscape around the island park: impacts and responses to Kibale National Park. Retrieved from http://scholars.unh.edu/geog_facpub/28/
- Golkowska, K., Rugani, B., Koster, D., & Van Oers, C. (2016). Environmental and economic assessment of biomass sourcing from extensively cultivated buffer strips along water bodies. *Environmental Science & Policy*, 57, 31–39.
- Gordon, L. J., Finlayson, C. M., & Falkenmark, M. (2010). Managing water in agriculture for food production and other ecosystem services. *Agricultural Water Management*, 97(4), 512–519.

- Gray, C. L., & Lewis, O. T. (2014). Do riparian forest fragments provide ecosystem services or disservices in surrounding oil palm plantations? *Basic and Applied Ecology*, *15*(8), 693–700.
- Gregory, P. J., Johnson, S. N., Newton, A. C., & Ingram, J. S. (2009). Integrating pests and pathogens into the climate change/food security debate. *Journal of Experimental Botany*, 60(10), 2827–2838.
- Greiner, R., & Gregg, D. (2011). Farmers' intrinsic motivations, barriers to the adoption of conservation practices and effectiveness of policy instruments: Empirical evidence from northern Australia. *Land Use Policy*, 28(1), 257–265.
- Greiner, R., Patterson, L., & Miller, O. (2009). Motivations, risk perceptions and adoption of conservation practices by farmers. *Agricultural Systems*, *99*(2–3), 86–104.
- Hansen, J. W., Marx, S. M., & Weber, E. U. (2004). The Role of Climate Perceptions,
 Expectations, and Forecasts in Farmer Decision Making: The Argentine Pampas and
 South Florida: Final Report of an IRI Seed Grant Project. Retrieved from
 https://academiccommons.columbia.edu/catalog/ac:126344
- Hartter, J. (2009). Attitudes of rural communities toward wetlands and forest fragments around Kibale National Park, Uganda. *Human Dimensions of Wildlife*, *14*(6), 433–447.
- Hassan, R., Nhemachena, C., & others. (2008). Determinants of African farmers' strategies for adapting to climate change: Multinomial choice analysis. *African Journal of Agricultural* and Resource Economics, 2(1), 83–104.
- Holloway, L. E., & Ilbery, B. W. (1996). Farmers' attitudes towards environmental change, particularly global warming, and the adjustment of crop mix and farm management. *Applied Geography*, 16(2), 159–171.

Horwith, B. (1985). A role for intercropping in modern agriculture. *BioScience*, 35(5), 286–291.

- Isaacs, K. B., Snapp, S. S., Chung, K., & Waldman, K. B. (2016). Assessing the value of diverse cropping systems under a new agricultural policy environment in Rwanda. *Food Security*, 8(3), 491–506.
- Islam, M. M., Gray, D., Reid, J., & Kemp, P. (2011). Developing sustainable farmer-led extension groups: lessons from a Bangladeshi case study. *The Journal of Agricultural Education and Extension*, 17(5), 425–443.
- Jayne, T. S., & Jones, S. (1997). Food marketing and pricing policy in Eastern and Southern Africa: A survey. *World Development*, *25*(9), 1505–1527.
- KABERUKA, D., & others. (2000). Rwanda Vision 2020. Retrieved from https://repositories.lib.utexas.edu/bitstream/handle/2152/5071/4164.pdf?sequence=1
- Kaiser, F. G., & others. (1996). Environmental attitude and ecological behavior. Retrieved from http://eric.ed.gov/?id=ED409179
- Kalinganire, A. (1996). Performance of Grevillea robusta in plantations and on farms under varying environmental conditions in Rwanda. *Forest Ecology and Management*, 80(1), 279–285.
- Karali, E., Rounsevell, M. D., & Doherty, R. (2011). Integrating the diversity of farmers' decisions into studies of rural land-use change. *Procedia Environmental Sciences*, 6, 136–145.
- Kareiva, P. (2011). *Natural capital: theory and practice of mapping ecosystem services*. Oxford University Press. Retrieved from https://books.google.com/books?hl=en&lr=&id=s-AVDAAAQBAJ&oi=fnd&pg=PP1&dq=Ecosystem+service+%2B+natural+corridors&ot s=VjqjkdyXxv&sig=lmI00UykQsqPatIIv3raCR_sacE

- Kautonen, T., Gelderen, M., & Fink, M. (2015). Robustness of the Theory of Planned Behavior in Predicting Entrepreneurial Intentions and Actions. *Entrepreneurship: Theory & Practice*, 39(3), 655–674.
- Keshavarz, M., & Karami, E. (2014). Farmers' decision-making process under drought. *Journal* of Arid Environments, 108, 43–56.
- Kimwaga, R. J., Mashauri, D. A., Bukirwa, F., Banadda, N., Wali, U. G., & Nhapi, I. (2012).
 Development of best management practices for controlling the non-point sources of pollution around Lake Victoria using SWAT Model: A Case of Simiyu catchment Tanzania. *Eng. J*, *5*, 77–83.
- Kiptot, E., Franzel, S., Hebinck, P., & Richards, P. (2006). Sharing seed and knowledge: farmer to farmer dissemination of agroforestry technologies in western Kenya. *Agroforestry Systems*, 68(3), 167–179.
- Lal, R. (2015). Sustainable Intensification for Adaptation and Mitigation of Climate Change and Advancement of Food Security in Africa. In R. Lal, B. R. Singh, D. L. Mwaseba, D. Kraybill, D. O. Hansen, & L. O. Eik (Eds.), *Sustainable Intensification to Advance Food Security and Enhance Climate Resilience in Africa* (pp. 3–17). Springer International Publishing.
- Lamarque, P., Tappeiner, U., Turner, C., Steinbacher, M., Bardgett, R. D., Szukics, U., ... Lavorel, S. (2011). Stakeholder perceptions of grassland ecosystem services in relation to knowledge on soil fertility and biodiversity. *Regional Environmental Change*, 11(4), 791–804.
- Leach, G., & Mearns, R. (2013). *Beyond the woodfuel crisis: people, land and trees in Africa*. Routledge. Retrieved from

https://books.google.com/books?hl=en&lr=&id=bbL9AQAAQBAJ&oi=fnd&pg=PP1&d q=Agriculture+%2B+fuel+wood+%2B+Africa&ots=bPa05sErP2&sig=xYhc9TUDprN-4r 1-XWNo6Sk8yA

- Li, S., Gu, S., Tan, X., & Zhang, Q. (2009). Water quality in the upper Han River basin, China: The impacts of land use/land cover in riparian buffer zone. *Journal of Hazardous Materials*, 165(1–3), 317–324.
- Lichtfouse, E. (2016). *Sustainable agriculture reviews. Volume 19* (Vols. 1–1 online resource (vi, 399 pages).). Cham: Springer. Retrieved from EBSCOhost http://search.ebscohost.com/login.aspx?direct=true&scope=site&db=nlebk&db=nlabk&A N=1175400
- Lichtfouse, E., Navarrete, M., Debaeke, P., Souchère, V., Alberola, C., & Ménassieu, J. (2009). Agronomy for sustainable agriculture: a review. In *Sustainable agriculture* (pp. 1–7). Springer.
- Little, K. M., & Gardner, R. A. (2003). Coppicing ability of 20 Eucalyptus species grown at two high-altitude sites in South Africa. *Canadian Journal of Forest Research*, 33(2), 181– 189.
- Logsdon, R. A., Kalcic, M. M., Trybula, E. M., Chaubey, I., & Frankenberger, J. R. (2015).
 Ecosystem services and Indiana agriculture: farmers' and conservationists' perceptions.
 International Journal of Biodiversity Science, Ecosystem Services & Management, 11(3), 264–282.
- Lovell, S., Mendez, V., Erickson, D., Nathan, C., & DeSantis, S. (2010). Extent, pattern, and multifunctionality of treed habitats on farms in Vermont, USA. *Agroforestry Systems*, 80(2), 153–171.

Lynch, L., & Brown, C. (2000). Landowner decision making about riparian buffers. *Journal of Agricultural and Applied Economics*, *32*(3), 585–596.

Malyon, S. (2014). Feeding Rwanda's livestock revolution. Appropriate Technology, 41(4), 48.

- McGinty, M. M., Swisher, M. E., & Alavalapati, J. (2008). Agroforestry adoption and maintenance: self-efficacy, attitudes and socio-economic factors. *Agroforestry Systems*, 73(2), 99–108.
- MINAGRI. (2011). Sustainable Crop Intensification in Rwanda. Republic of Rwanda, Ministry of Agriculture and Animal Resources. Retrieved from http://www.minagri.gov.rw/fileadmin/user_upload/documents/CIP/CIP_Strategies_2011. pdf
- MINITERE. (2004). Environmental policy. Ministry of Lands, Environment, Forests, Water and Mines, Kigali.
- Miserendino, M. L., Casaux, R., Archangelsky, M., Di Prinzio, C. Y., Brand, C., & Kutschker,
 A. M. (2011). Assessing land-use effects on water quality, in-stream habitat, riparian
 ecosystems and biodiversity in Patagonian northwest streams. *Science of the Total Environment*, 409(3), 612–624.
- Montagnini, F., Francesconi, W., & Rossi, E. (2011). *Agroforestry as a tool for landscape restoration*. New York: Nova Science Publishers. Retrieved from http://search.ebscohost.com/login.aspx?direct=true&scope=site&db=nlebk&db=nlabk&A N=431069
- Mugonola, B., Kimaro, D., Isabirye, M., Deckers, J., Poesen, J., Wanyama, J., & Mathijs, E.
 (2013). Economics of Grass Strips Used as Sediment Filters in the Riparian Zones of Lake Victoria, Uganda. *Agroecology & Sustainable Food Systems*, 37(7), 1040–1062.

Musanze, R. (2013). Determinants of Farmers' Participation in Formal Credit Markets in Rural Rwanda. Retrieved from http://www.krepublishers.com/02-Journals/JAS/JAS-04-0-000-13-Web/JAS-04-2-000-13-Abst-PDF/JAS-04-2-087-13-081-Hitayezu-P/JAS-04-2-087-13-081-Hitayezu-P-05-Tt.pdf

- Musshoff, O., & Hirschauer, N. (2008). Adoption of organic farming in Germany and Austria: an integrative dynamic investment perspective. *Agricultural Economics*, *39*(1), 135–145.
- Mutandwa, E., & Kanyarukiga, R. (2016). Understanding the role of forests in rural household economies: experiences from the Northern and Western provinces of Rwanda. *Southern Forests: A Journal of Forest Science*, 0(0), 1–8.
- Nabahungu, N. L., & Visser, S. M. (2013a). Farmers' Knowledge and Perception of Agricultural Wetland Management in Rwanda. *Land Degradation & Development*, *24*(4), 363–374.
- Nabahungu, N. L., & Visser, S. M. (2013b). FARMERS'KNOWLEDGE AND PERCEPTION OF AGRICULTURAL WETLAND MANAGEMENT IN RWANDA. *Land Degradation* & Development, 24(4), 363–374.
- Nabanoga, G. (2005). Transgressing Boundaries: Gendered spaces, species and indigenous forest management in Uganda. Retrieved from http://library.wur.nl/WebQuery/wurpubs/338400
- Nahayo, A., Ekise, I. E., & Niyigena, D. (2013). Assessment of the contribution of Non Timber
 Forest Products to the improvement of local people's livelihood in Kinigi sector,
 Musanze District, Rwanda. *Ethiopian Journal of Environmental Studies and*Management, 6(6), 698–706.

- Nath, C. D., Pélissier, R., Ramesh, B. R., & Garcia, C. (2011). Promoting native trees in shade coffee plantations of southern India: comparison of growth rates with the exotic Grevillea robusta. *Agroforestry Systems*, 83(2), 107–119.
- National Agroforestry Center (U.S.). (2012). *What is agroforestry?* Lincoln, Neb.: USDA National Agroforestry Center.
- Ndayambaje, J. D., Heijman, W. J. M., & Mohren, G. M. J. (2013). Farm woodlots in rural Rwanda: purposes and determinants. *Agroforestry Systems*, 87(4), 797–814.
- Ndayambaje, J. D., & Mohren, G. M. J. (2011). Fuelwood demand and supply in Rwanda and the role of agroforestry. *Agroforestry Systems*, *83*(3), 303–320.
- NISR. (2016). Seasonal Agricultural Survey 2016 (SAS2016) report. National Institute of Statistics of Rwanda. Retrieved from http://www.statistics.gov.rw
- Nzeyimana, I., Hartemink, A. E., & Geissen, V. (2016). Correction: GIS-based multi-criteria analysis for Arabica coffee expansion in Rwanda. *PloS One*, *11*(2), e0149239.
- Okalebo, J. R., Othieno, C. O., Woomer, P. L., Karanja, N. K., Semoka, J. R. M., Bekunda, M. A., ... Mukhwana, E. J. (2007). Available technologies to replenish soil fertility in East Africa. In A. Bationo, B. Waswa, J. Kihara, & J. Kimetu (Eds.), *Advances in Integrated Soil Fertility Management in sub-Saharan Africa: Challenges and Opportunities* (pp. 45–62). Springer Netherlands.
- Olson, D. H., Anderson, P. D., Frissell, C. A., Welsh Jr., H. H., & Bradford, D. F. (2007).
 Biodiversity management approaches for stream–riparian areas: Perspectives for Pacific Northwest headwater forests, microclimates, and amphibians. *Forest Ecology and Management*, 246(1), 81–107.

- Oreg, S., & Katz-Gerro, T. (2006). Predicting proenvironmental behavior cross-nationally: Values, the theory of planned behavior, and value-belief-norm theory. *Environment and Behavior*, 38(4), 462–483.
- Palmer, G. C., & Bennett, A. F. (2006). Riparian zones provide for distinct bird assemblages in forest mosaics of south-east Australia. *Biological Conservation*, 130(3), 447–457.
- Palmer, M. A., Reidy Liermann, C. A., Nilsson, C., Flörke, M., Alcamo, J., Lake, P. S., & Bond,
 N. (2008). Climate change and the world's river basins: anticipating management
 options. *Frontiers in Ecology and the Environment*, 6(2), 81–89.
- Poppy, G. M., Chiotha, S., Eigenbrod, F., Harvey, C. A., Honzák, M., Hudson, M. D., ... Dawson, T. P. (2014). Food security in a perfect storm: using the ecosystem services framework to increase understanding. *Philosophical Transactions of the Royal Society of London B: Biological Sciences*, 369(1639), 20120288.
- Pritchard, M. F. (2013). Land, power and peace: Tenure formalization, agricultural reform, and livelihood insecurity in rural Rwanda. *Land Use Policy*, *30*(1), 186–196.
- Rasmussen, L. V., Mertz, O., Christensen, A. E., Danielsen, F., Dawson, N., & Xaydongvanh, P.
 (2016). A combination of methods needed to assess the actual use of provisioning ecosystem services. *Ecosystem Services*, 17, 75–86.
- Reid, H. (2016). Ecosystem- and community-based adaptation: learning from community-based natural resource management. *Climate and Development*, 8(1), 4–9.
- Robinson, N., Harper, R. J., & Smettem, K. R. J. (2006). Soil water depletion by Eucalyptus spp. integrated into dryland agricultural systems. *Plant and Soil*, *286*(1–2), 141–151.
- Roothaert, R. L., & Franzel, S. (2001). Farmers' preferences and use of local fodder trees and shrubs in Kenya. *Agroforestry Systems*, *52*(3), 239–252.

- RWANDA, G. O., DU RWANDA, G., & others. (1991). Stratégie nationale de conservation des sols: Evaluation des système d'exploitation agricole pour une régionalisation des techniques de conservation et d'amélioration de fertilité des sols au Rwanda. Retrieved from https://repositories.lib.utexas.edu/bitstream/handle/2152/5212/2515.pdf?sequence=1
- Safari, B. (2010). A review of energy in Rwanda. *Renewable and Sustainable Energy Reviews*, 14(1), 524–529.
- Sahu, M., & Gu, R. R. (2009). Modeling the effects of riparian buffer zone and contour strips on stream water quality. *Ecological Engineering*, *35*(8), 1167–1177.
- Salam, M. A., Noguchi, T., & Koike, M. (2000). Understanding why farmers plant trees in the homestead agroforestry in Bangladesh. *Agroforestry Systems*, 50(1), 77–93.
- Sanchez, P. A. (2002). Soil fertility and hunger in Africa. *Science(Washington)*, 295(5562), 2019–2020.
- Scarborough, V., Killough, S., Johnson, D. A., Farrington, J., & Axinn, G. H. (1997). 2.
 Challenges to agricultural extension in the twenty-first century. In *Farmer-led Extension* (pp. 13–22). Practical Action Publishing.
- Scherr, S. J. (1995). Economic factors in farmer adoption of agroforestry: Patterns observed in Western Kenya. World Development, 23(5), 787–804.
- Schulp, C. J. E., Thuiller, W., & Verburg, P. H. (2014). Wild food in Europe: A synthesis of knowledge and data of terrestrial wild food as an ecosystem service. *Ecological Economics*, 105, 292–305.
- Sepp, K. (2012). Landscape function and Ecosystem Services. *Rural Development and Land Use*, (3), 39.

- Sheeder, R. J., & Lynne, G. D. (2011). Empathy-conditioned conservation: "Walking in the shoes of others" as a conservation farmer. *Land Economics*, *87*(3), 433–452.
- Singh, R. B. (2000). Environmental consequences of agricultural development: a case study from the Green Revolution state of Haryana, India. *Agriculture, Ecosystems & Environment*, 82(1), 97–103.
- Smale, M., Byerlee, D., & Jayne, T. (2013). Maize revolutions in sub-Saharan Africa. In an African green revolution (pp. 165–195). Springer.
- Smith, H. F., & Sullivan, C. A. (2014). Ecosystem services within agricultural landscapes— Farmers' perceptions. *Ecological Economics*, *98*, 72–80.
- Smukler, S. M., Philpott, S. M., Jackson, L. E., Klein, A.-M., DeClerck, F., Winowiecki, L., & Palm, C. A. (2012). Ecosystem services in agricultural landscapes. In *Integrating Ecology* and Poverty Reduction (pp. 17–51). Springer. Retrieved from http://link.springer.com/chapter/10.1007/978-1-4419-0633-5_3
- Sweeney, B. W., Bott, T. L., Jackson, J. K., Kaplan, L. A., Newbold, J. D., Standley, L. J., ... Horwitz, R. J. (2004). Riparian deforestation, stream narrowing, and loss of stream ecosystem services. *Proceedings of the National Academy of Sciences of the United States of America*, 101(39), 14132–14137.
- Swinton, S. M., Lupi, F., Robertson, G. P., & Hamilton, S. K. (2007a). Ecosystem services and agriculture: Cultivating agricultural ecosystems for diverse benefits. *Ecological Economics*, 64(2), 245–252.
- Swinton, S. M., Lupi, F., Robertson, G. P., & Hamilton, S. K. (2007b). Ecosystem services and agriculture: cultivating agricultural ecosystems for diverse benefits. *Ecological Economics*, 64(2), 245–252.

- Thiaw, I., Kumar, P., Yashiro, M., & Molinero, C. (2011). Food and ecological security:
 identifying synergy and trade-offs. UNEP Policy Series: Ecosystem Management, 4, 1–
 12.
- Tilman, D., Balzer, C., Hill, J., & Befort, B. L. (2011). Global food demand and the sustainable intensification of agriculture. *Proceedings of the National Academy of Sciences*, 108(50), 20260–20264.
- Tittonell, P. (2014). Ecological intensification of agriculture—sustainable by nature. *Current Opinion in Environmental Sustainability*, *8*, 53–61.

Turnbull, J. W., & others. (2000). Economic and social importance of eucalypts. CSIRO Publishing, Australia. Retrieved from https://books.google.com/books?hl=en&lr=&id=PENpGhQ1qmgC&oi=fnd&pg=PA1&d q=+Economic+and+social+importance+of+eucalypts.+In:+Keane+&ots=0hrPcGCEqW &sig=l0dxhcaqL3OJIsYW15WAF s9NaE

- Viala, E. (2008). Water for food, water for life a comprehensive assessment of water management in agriculture. *Irrigation and Drainage Systems*, *22*(1), 127–129.
- Vignola, R., Koellner, T., Scholz, R. W., & McDaniels, T. L. (2010). Decision-making by farmers regarding ecosystem services: Factors affecting soil conservation efforts in Costa Rica. *Land Use Policy*, 27(4), 1132–1142.

Wali, U. G., Nhapi, I., Ngombwa, A., Banadda, N., Nsengimana, H., Kimwaga, R. J., &
 Nansubuga, I. (2011). Modelling of Nonpoint Source Pollution in Akagera
 Transboundary River in Rwanda. *Open Environmental Engineering Journal*, 4, 124–132.

Wejnert, B. (2002). Integrating Models of Diffusion of Innovations: A Conceptual Framework. Annual Review of Sociology, 28, 297–326.

- Wratten, S. D., Gillespie, M., Decourtye, A., Mader, E., & Desneux, N. (2012). Pollinator habitat enhancement: benefits to other ecosystem services. *Agriculture, Ecosystems & Environment*, 159, 112–122.
- Wunder, S., Angelsen, A., & Belcher, B. (2014). Forests, livelihoods, and conservation: broadening the empirical base. *World Development*, 64, S1–S11.
- Young-Mathews, A., Culman, S. W., Sánchez-Moreno, S., O'Geen, A. T., Ferris, H., Hollander,
 A. D., & Jackson, L. E. (2010). Plant-soil biodiversity relationships and nutrient retention
 in agricultural riparian zones of the Sacramento Valley, California. *Agroforestry Systems*, 80(1), 41–60.
- Zegeye, A. D., Steenhuis, T. S., Blake, R. W., Kidnau, S., Collick, A. S., & Dadgari, F. (2010).
 Assessment of soil erosion processes and farmer perception of land conservation in Debre Mewi watershed near Lake Tana, Ethiopia. *Ecohydrology & Hydrobiology*, 10(2–4), 297–306.
- Zhang, W., Ricketts, T. H., Kremen, C., Carney, K., & Swinton, S. M. (2007). Ecosystem services and dis-services to agriculture. *Ecological Economics*, *64*(2), 253–260.
- Zhang, X., Liu, X., Zhang, M., Dahlgren, R. A., & Eitzel, M. (2010). A review of vegetated buffers and a meta-analysis of their mitigation efficacy in reducing nonpoint source pollution. *Journal of Environmental Quality*, 39(1), 76–84.
- Zhao, Y., Wang, C., Wang, S., & Tibig, L. V. (2005). Impacts of present and future climate variability on agriculture and forestry in the humid and sub-humid tropics. *Climatic Change*, 70(1), 73–116.
- Zimmerman, B. J. (2000). Self-Efficacy: An Essential Motive to Learn. *Contemporary Educational Psychology*, 25(1), 82–91.

- Zougmoré, R., Kambou, F. N., Ouattara, K., & Guillobez, S. (2000). Sorghum-cowpea intercropping: An effective technique against runoff and soil erosion in the Sahel (Saria, Burkina Faso). *Arid Soil Research and Rehabilitation*, *14*(4), 329–342.
- Zubair, M., & Garforth, C. (2006). Farm Level Tree Planting in Pakistan: The Role of Farmers' Perceptions and Attitudes. *Agroforestry Systems*, 66(3), 217–229.

Chapter 5: General Conclusion

Average farm size for a vast majority of households in the northern highlands of Rwanda is less than 0.75 hectare. Research on farmers' perceptions of climate change and adaptation in Africa has mainly focused on dry regions and adaptation has focused on farmers' adaptation practices, with less attention to their perceptions. This study focused on smallholder farmers in a humid region where flood hazards, soil fertility loss due to erosion are of concern and crop intensification practices meet.

My findings indicate that smallholder farmers in the Upper Mukungwa Watershed in the Musanze District believe there has been shifts in seasonal onset of rains in recent years and they reported they and they also plant their crops at variable dates; most of the farmers planting earlier in the season. Findings also reveal that smallholder farmers in the area seen an increase in areal cultivation of corn, beans and Irish potatoes, following a Rwandan government-mandated crop intensification program that started in 2007. However, those crops are reported by farmers to be the most vulnerable to climate change risks in comparison with other crops grown in the area. To sustain food crop yield within the crop intensification program setting and in the face of climate change risks, this study shows that farmers recognize the role of agroforestry in providing ecosystem services that support their food production. However, their understanding of the role of agroforestry in supporting other regulating ecosystem services such as water quality control, pollination and conservation of biodiversity is lacking, probably because smallholder farmers do not see a direct linkage of those ecosystem services and their food crop production. Nevertheless, smallholder farmers expressed their willingness to adopt agroforestry as well as to establish riparian buffers.

The study also shows that smallholder farmers who strongly agreed that soil erosion within farms proximal to streams is a serious threat were more likely to support the idea that establishing a riparian buffer would help entrap sediments and mitigate soil erosion within farmlands adjacent to streams. In the face of land scarcity, smallholder farmers support a multifunctional riparian buffer zone that meets their goals by providing fodder for their livestock, building material, poles to support climbing beans and erosion control.

Policy

The Government of Rwanda's new agricultural policy involving mono-cropping systems coupled with intensive use of synthetic fertilizers and pesticides seems to align with farmers' indication that food and money from selling those food crops are most important to them; however these approaches are likely to compromise the fertility of the soil, pollination services and water quality. This study reveals that, among other challenges, farmers lack sufficient technological capacity to implement techniques to maintain ESs, such as agroforestry, while they reported at the same time that they rely mostly on agricultural extension services for agricultural information and technology. This highlights a need for policy development to make a clear provision for effective engagement of extension services agents with smallholder farmers with regard to the uptake of agroforestry practices that contribute to food security and environmental integrity.

Finally, the chapter also reports smallholder farmers' need for financial support. Trees take up to ten years to grow to maturity and farmers need to harvest crops each agricultural season. During

the tree growing period, the yield of crops farmers normally grow in those riparian zones will certainly decline, thus cutting their net yearly income from the riparian farm plots. To balance farmers' short term benefits of crop cultivation and long-term benefits of agroforestry implementation, the agrarian policy should make provision for incentives provision to poor farmers, particularly those who own farm plots proximal to water bodies and are willing to establish riparian buffers, to offset the gap until trees have grown to a stage in which farmers experience their benefits and may not need any more incentives.

Practice

Implications of famers' perceptions of environmental change and adaptation for practice are discussed in Chapter 2, 3 and 4, including recommendations to engage smallholder farmers' into the land management reform process. The Second Economic Development and Poverty Reduction Strategy (EDPRS II) for 2013-2018 recognizes environment and climate change issues to be of worthy of attention. The strategy promotes (i) mainstreaming environmental sustainability into productive and social sectors; (ii) reducing vulnerability to climate change and (iii) preventing and controlling pollution. Understanding farmers' perceptions will facilitate the adoption of strategies geared toward environmental change vulnerability reduction through the recovery and maintenance of ecosystem services. While recommendations pertain to the context of the northern highlands of Rwanda, they could be transferred to other regions in Rwanda where smallholder farmers are facing similar environmental challenges.

Suggestions for Future Research

This dissertation represents a contribution to understanding smallholder farmers' perceptions of environmental change and adaptation and sets a stage for future research. First there is a need to gauge farmers' perceptions of environmental change across all agro-ecological zones in Rwanda to assess the variability of perceptions in the face of varying socio-economic and geophysical contexts. To date, information about farmers' preference for particular tree species for riparian buffers and agroforestry under various socio-economic and geographic context, for example, is lacking. A comparative study on farmers' perceptions of environmental change and adoption across the country will help agrarian policy makers design a policy that reflects farmers' goals and interests.

Secondly, results from this study suggest that research on agroforestry move from research stations that are inaccessible to smallholder farmers to farmers' own field plots. This should deviate from benefiting merely government institutions to the ownership of the research from research stations by farmers themselves. Such a strategy could provide an opportunity for smallholder farmers to learn the agroforestry technology and to participate in a cost/benefit analysis of the adoption of agroforestry and to engage actively in the establishment of riparian buffer zones. Once the technology is adopted by smallholder farmers participating in the study, the technology uptake among neighboring farmers could be effective.

Thirdly, research is needed that looks at various agroforestry practices to assess their economic and environmental contribution to environmental change adaptation. A few studies have examined agroforestry in Rwanda and have documented the practice in various parts of the

country but there is minimal quantitative documentation of economic and environmental value of various agroforestry systems in Rwanda and how agroforestry could be reconfigured to fulfil those functions.

As the Rwandan agrarian policy strongly promotes crop intensification programs, there is the possibility to reconcile monocroppping systems associated with the program and agroforestry. Therefore, research is needed to examine a feasibility analysis pertaining to the sustainable, environmentally and economically profitable integration of agroforestry with crop intensification. Such research could contribute to the national goal of moving away from subsistence towards market-oriented agriculture and also contribute to meeting smallholder farmer need for fuel, stakes for climbing beans, building material, fodder for animals, while also contributing to soil conservation.

Lastly, for riparian buffers to be effective, a study is needed to look at the effectiveness of various riparian buffer configurations in meeting smallholder farmers' goals while providing environmental benefits at the same time. In a bigger picture, it would be valuable to conduct a study on how agroforestry practices and maintenance of functional riparian buffers upstream can contribute to water quality downstream and how the reduction of the cost of water treatment downstream can contribute to providing incentives for best agricultural practices upstream.